RISK



Deborah Lupton





Also available as a printed book see title verso for ISBN details

RISK

Risk has become a key concept in western societies. People are encouraged to seek out information on risk and to take steps to avoid it as much as possible. Risk analysis, risk assessment and risk management are ever-expanding industries. We are now living in a 'risk society'.

In this book, Deborah Lupton examines why risk has come to such prominence at this particular point in history. The author traces how risk has been constructed over time from pre-modernity to the later modern era and provides an introduction to the main theories surrounding the subject. She covers a wide range of issues including

- Risk and culture
- Sociocultural and scientific perspectives
- · Risk and blame
- Risk and pleasure

Including examples of the ways in which risk is experienced in everyday life, *Risk* provides a lively and engaging introduction to one of today's major sociocultural concepts and will be essential reading for both students and teachers alike.

Deborah Lupton is Associate Professor in Cultural Studies and Cultural Policy and Director of the Centre for Cultural Risk Research at Charles Sturt University, Australia. Her latest books include *The New Public Health: Health and Self in the Age of Risk* (1996 with Alan Petersen) and *The Emotional Self* (1998).

KEY IDEAS

SERIES EDITOR: PETER HAMILTON, THE OPEN UNIVERSITY, MILTON KEYNES

Designed to complement the successful *Key Sociologists*, this series covers the main concepts, issues, debates, and controversies in sociology and the social sciences. The series aims to provide authoritative essays on central topics of social science, such as community, power, work, sexuality, inequality, benefits and ideology, class, family, etc. Books adopt a strong individual 'line' constituting original essays rather than literary surveys, and form lively and original treatments of their subject matter. The books will be useful to students and teachers of sociology, political science, economics, psychology, philosophy, and geography.

The Symbolic Construction of Community ANTHONY P.COHEN, Department of Social Anthropology, University of Manchester

Society
DAVID FRISBY and DEREK SAYER,
Department of Sociology,
University of Manchester

Sexuality
JEFFREY WEEKS, Social Work
Studies Department, University of
Southampton

Working GRAEME SALAMAN, Faculty of Social Sciences, The Open University, Milton Keynes

Beliefs and Ideology KENNETH THOMPSON, Faculty of Social Sciences, The Open

University, Milton Keynes

Equality

BRYAN TURNER, School of Social Sciences, The Flinders University of South Australia

Hegemony

ROBERT BOCOCK, Faculty of Social Sciences, The Open University, Milton Keynes

Racism

ROBERT MILES, Department of Sociology, University of Glasgow

Postmodernity

BARRY SMART, Associate Professor of Sociology, University of Auckland, New Zealand

Globalization

MALCOLM WATERS, University of Tasmania

Moral Panics

KENNETH THOMPSON, Faculty of Social Sciences, The Open University, Milton Keynes

RISK

Deborah Lupton



First published 1999 by Routledge 11 New Fetter Lane, London EC4P 4EE

This edition published in the Taylor & Francis e-Library, 2005.

"To purchase your own copy of this or any of Taylor & Francis or Routledge's collection of thousands of eBooks please go to www.eBookstore.tandf.co.uk."

Simultaneously published in the USA and Canada by Routledge 29 West 35th Street, New York, NY 10001

© 1999 Deborah Lupton

All rights reserved. No part of this book may be reprinted or reproduced or utilized in any form or by any electronic, mechanical, or other means, now known or hereafter invented, including photocopying and recording, or in any information storage or retrieval system, without permission in writing from the publishers.

British Library Cataloguing in Publication Data
A catalogue record for this book is available from the British
Library

Library of Congress Cataloguing in Publication Data
Lupton, Deborah
Risk/Deborah Lupton
p. cm—Key Ideas
Includes bibliographical references and index.
1. Risk-Sociological aspects. 2. Risk perceptionSocial aspects. I. Title. II. Series.
HM256.L86 1999
302'.12–dc21

ISBN 0-203-98054-9 Master e-book ISBN

ISBN 0-415-18333-2 (hbk) ISBN 0-415-18334-0 (pbk)

CONTENTS

	ACKNOWLEDGEMENTS	X
1	Introduction	1
	Changes in the meaning of risk	5
	This book	13
2	Theorizing risk	18
	The cognitive science perspective	18
	Sociocultural perspectives	25
	Social constructionist positions	29
	Concluding comments	35
3	Risk and culture	37
	The importance of culture	38
	Purity, danger and the body	40
	Risk and blame	46
	The grid-group model	51
	Concluding comments	57
4	Risk and reflexive modernization	60
	Beck and the 'risk society'	60
	Reflexive modernization	67
	Individualization	71

	Giddens' perspectives on risk	74
	Trust	80
	Concluding comments	83
5	Risk and governmentality	86
	Governmentality	87
	From dangerousness to risk	93
	Contemporary risk strategies	97
	Concluding comments	104
6	Risk and subjectivity	106
	Risk knowledges and reflexivity	107
	Social structures and power relations	115
	Aesthetic and habitual dimensions	120
	Concluding comments	124
7	Risk and Otherness	126
	Embodiment	127
	Hybridity and liminality	132
	The psychodynamics of Otherness	140
	Spatiality and Otherness	145
	Concluding comments	150
8	Risk and pleasure	152
	Escape attempts and edgework	153
	Risk-taking as gendered performances	161
	Desire and transgression	167
	Concluding comments	175

BIBLIOGRAPHY	177
INDEX	186

ACKNOWLEDGEMENTS

I am grateful to the two anonymous readers of the draft manuscript for their helpful comments and to Mari Shullaw of Routledge for her advice. Thanks are due, as ever, to Gamini Colless for his personal support. This book is dedicated to our daughter Phoebe, born September 1998.

1 INTRODUCTION

In his history of life in medieval France, Robert Muchembled (1985) portrays a world in which there were many threats and dangers to human health and life for both peasantry and aristocracy alike. During this era, death was 'on display everywhere, to the point of banality' (Muchembled 1985:31). The most extreme of these threats and dangers were hunger, cold, epidemic disease and war. Food supply at the end of the Middle Ages in Europe was very tenuous. Grain provided the basis of the diet, and production was vulnerable to the vicissitudes of the seasons. Infant mortality was very high and lifespans short (reaching the age of 40 being considered a fair lifespan). Epidemics of such diseases as smallpox, whooping cough, typhoid, syphilis, dysentery and the plague regularly struck villages and towns, decimating their populations. People living in rural areas faced other dangers, such as wolves, who were known to attack children and sometimes adults in the fields and near their cottages. Wild dogs presented the threat of rabies should they bite and wild pigs were known to attack and eat small children. Bands of brigands roamed the forests and high roads, regularly attacking and robbing peasants. Feuds and battles between family clans were frequent and violent, often resulting in murder.

In this world, insecurity was rife and permanent, and 'fears, real and imaginary, abounded' (Muchembled 1985: 22). Dead bodies were very much in the public eye. Bodies of hanged men dangled from gallows, slowly rotting, for

several days or weeks in public places, executions took place in public and the dead victims of brigands and soldiers were to be found on roads, as were the bodies of beggars and vagabonds who had perished from starvation or disease. Villagers and townspeople closed themselves in their dwellings at night, not daring to go out once darkness had fallen, because night was considered the domain of all dangers: the kingdom of the Devil, of demons, of witches and werewolves and monstrous beasts. They were also afraid of natural events that were seen to disturb the order of things, such as comets, extraordinary cold, earthquakes, floods (ibid.: 25–6).

In medieval France, therefore, magic, combined with a dash of Christianity, served as the belief system by which threats and dangers were dealt with conceptually and behaviourally, allowing people to feel as if they had some sense of control over their world. The presence of the supernatural was taken for granted, incorporating notions of a vengeful God with that of an evil Satan (ibid.: 27-9). A network of superstition existed to deal with evil. including beliefs in portents, pilgrimages to shrines, amulets, offerings to the gods, avoidance of tabooed places such as crossroads and of people such as lepers and gypsies (ibid.: 29-30). Everyday life was full of customs and beliefs that involved behaving in certain ways or avoiding actions so as to ward off danger or disease. Thus, for example, in some areas of France it was believed that kittens born in the month of May must be drowned, or danger would threaten (ibid.: 81).

This description of everyday life and beliefs in medieval France provides a fascinating account of how people in that particular historical and sociocultural context dealt with danger, hazard and fear. As Muchembled shows, societies develop a system of strategies and beliefs in the attempt to deal with, contain and prevent danger. To lack such systems is to throw oneself upon the mercy of fate, to relinquish any sense of control. In contemporary western societies, where control over one's life has become

increasingly viewed as important, the concept of 'risk' is now widely used to explain deviations from the norm, misfortune and frightening events. This concept assumes human responsibility and that 'something can be done' to prevent misfortune. Feelings of insecurity are common, just as they were in pre-modern times, but we now harbour somewhat different fears, different targets and causes for our anxiety. While we no longer regularly view dead bodies lying about, while the plague has all but vanished as a cause of death, while infant mortality is extremely low and most of us expect to live well into old age, we fear being the victim of a crime, falling prey to cancer, being in a car accident, losing our jobs, having our marriage break down or our children fail at school. As in pre-modern times, the symbolic basis of our uncertainties is anxiety created by disorder, the loss of control over our bodies, our relationships with others, our livelihoods and the extent to which we can exert autonomy in our everyday lives.

As in pre-modern times, we may acknowledge that threats exist, but we like to think that something can be done to deal with them. We may not perform such acts as drowning unfortunate kittens born in the wrong month as a strategy of risk prevention, but we have a range of other strategies that in emotional terms serve the same purpose. Agitating for anti-pollution legislation, watching one's diet, having tests to diagnose disease in its early stages, installing a burglar alarm in one's home, reading self-help books or taking out life insurance are all ways in which people seek to contain and manage the anxiety and fear aroused by what they perceive to be a danger or threat. Rational thinking, bureaucratic systems of prevention, ways of identifying threats before they take effect, are regularly put forward as means of managing danger and threats.

The response to the Port Arthur killings in Australia is one example. On 28 April 1996, a young man, Martin Bryant, entered the Port Arthur historic tourist site in the Australian island state of Tasmania and proceeded to shoot randomly at people visiting and working there using two semi-automatic weapons. He then left Port Arthur and travelled to a local guest house, shooting at others and taking a male hostage en route. Bryant was not arrested until the next morning, when he ran out of the guest house after having set it alight. He had left 35 people dead and a further 17 badly wounded.

A central theme in the media coverage of the Port Arthur killings, which received a high level of publicity both in Australia and abroad, questioned why such an event could have happened. In trying to give a meaning to Bryant's acts, numerous suggestions were put forward. One suggestion was that Bryant had been exposed to too many violent films or 'video nasties' which had influenced him to commit murder. Some media reports suggested that he had been bullied as a child because he was 'simple' or too effeminate looking, and that the murders may have been an act of revenge upon the world for this cruelty. Still others commented that Bryant's father was violent and had beaten him when he was a child, and that this had led him into violence himself. Several calls for action were suggested. The major one was gun control, particularly in relation to semi-automatic weapons. New gun control regulations were seized upon as a strategy of risk containment. A secondary call for action was that violent films and videos be banned or censored. Further suggestion was made that individuals with a history of psychiatric illness be prevented from acquiring gun licences. All of these strategies were directed at regaining control over what was seen to be an unanticipated expression of evil, helping people deal with the horror, anger and frustration, the loss of control they felt in response to Martin Bryant's actions. These strategies are the products of late modern ways of thinking about, and reacting to, risk.

CHANGES IN THE MEANING OF RISK

Over the centuries, the word 'risk' has changed its meaning and its use has become far more common and applied to a plethora of situations. Luhmann (1993:9) claims that the word 'risk' appeared in German in references in the mid sixteenth century and English in the second half of the seventeenth century. He notes, however, that the renaissance Latin term riscum had been in use long before in countries such as Germany. Most commentators link the emergence of the word and concept of risk with early maritime ventures in the pre-modern period. Ewald argues that the notion of risk first appeared in the Middle Ages, related to maritime insurance and used to designate the perils that could compromise a voyage: 'At that time, risk designated the possibility of an objective danger, an act of God, a force majeure, a tempest or other peril of the sea that could not be imputed to wrongful conduct' (ibid.: 226). This concept of risk, therefore, excluded the idea of human fault and responsibility. Risk was perceived to be a natural event such as a storm, flood or epidemic rather than a humanmade one. As such, humans could do little but attempt to estimate roughly the likelihood of such events happening and take steps to reduce their impact.

Changes in the meanings and use of risk are associated with the emergence of modernity, beginning in the seventeenth century and gathering force in the eighteenth century. Modernity has been defined as 'the institutions and modes of behaviour established first of all in postfeudal Europe, but which in the twentieth century increasingly have become world-historical in their impact' (Giddens 1991:14-15). Modernity is equivalent to the 'industrialized world', incorporating capitalism, the institutions of surveillance and nuclear weaponry as well as the process of industrialism. Modernity depends upon the notion. emerging seventeenth-century in the

Enlightenment, that the key to human progress and social order is objective knowledge of the world through scientific exploration and rational thinking. It assumes that the social and natural worlds follow laws that may be measured, calculated and therefore predicted.

During the eighteenth and nineteenth centuries, the early modern European states sought to harness their populations productively and deal with the social changes and upheavals wrought by mass urbanization and industrialization as part of the Industrial Revolution. The science of probability and statistics was developed as a means of calculating the norm and identifying deviations from the norm, thus embodying the belief that rationalized counting and ordering would bring disorder under control (Hacking 1990). These fields were to become important to the modernist technical notion of risk. During the eighteenth century, the concept of risk had begun to be scientized, drawing upon new ideas in mathematics relating to probability. The development of statistical calculations of risk and the expansion of the insurance industry in the early modern era meant that: 'Consequences that at first affect only the individual become "risks", systematically caused, statistically describable and in that sense "predictable" types of events, which can therefore also be subjected to supraindividual and political rules of recognition, compensation and avoidance' (Beck 1992a:99). By the nineteenth century, the notion of risk was extended. It was no longer located exclusively in nature, but was 'also in human beings, in their conduct, in their liberty, in the relations between them, in the fact of their association, in society' (Ewald 1993:226).

The modernist concept of risk represented a new way of viewing the world and its chaotic manifestations, its contingencies and uncertainties. It assumed that unanticipated outcomes may be the consequence of human action rather than 'expressing the hidden meanings of nature or ineffable intentions of the Deity', largely

replacing earlier concepts of fate or *fortuna* (Giddens 1990: 30). As Reddy argues: 'Moderns had eliminated

1990: 30). As Reddy argues: 'Moderns had eliminated genuine indeterminacy, or "uncertainty", by inventing "risk". They had learnt to transform a radically indeterminate cosmos into a manageable one, through the myth of calculability' (Reddy 1996:237). Castel goes even further, arguing that the obsession with the prevention of risk in modernity is built upon

a grandiose technocratic rationalizing dream of absolute control of the accidental, understood as the irruption of the unpredictable...a vast hygienist utopia plays on the alternate registers of fear and security, inducing a delirium of rationality, an absolute reign of calculative reason and a no less absolute prerogative of its agents, planners and technocrats, administrators of happiness for a life to which nothing happens.

(Castel 1991:289)

In modernity, risk, in its purely technical meaning, came to rely upon conditions in which the probability estimates of an event are able to be known or knowable. Uncertainty, in contrast, was used as an alternative term when these probabilities are inestimable or unknown. For example, in one form of economic theory developed in the mid nineteenth century risk was differentiated from uncertainty in this way. This distinction presupposed that there was a form of indeterminacy that was not subject to rational calculation of the likelihood of various alternative possibilities (Reddy 1996:227). John Maynard Keynes, the influential English economic theorist, built on this distinction to argue that investors' behaviour should be classified as subject to uncertainty rather than the laws of risk because they were 'driven by "animal spirits"' which 'by their very nature were not subject to probabilistic or "risk" analysis' (Reddy ibid.: 229).

Modernist notions of risk also included the idea that risk could be both 'good' and 'bad'. The notion of risk as it developed in insurance is associated with notions of chance or probability on one hand and loss and damage on the other. These two sets of notions come together in the concept of the accident, against which one insures oneself: 'Insurance's general model is the game of chance: a risk, an accident comes up like a roulette number, a card pulled out of a pack. With insurance, gaming becomes a symbol of the world' (Ewald 1991:199). From this perspective, 'risk' is a neutral concept, denoting the probability of something happening, combined with the magnitude of associated losses or gains. In other words, there once was such a thing as a 'good' risk as well as a 'bad' risk (Douglas 1992:23). This meaning of risk dominated until the beginning of the nineteenth century (Ewald 1991).

At the end of the twentieth century, these fine distinctions between risk and uncertainty and 'good risk' and 'bad risk' tend to be somewhat lost. The use of the risk, as Douglas contends, now 'has not got much to do with probability calculations. The original connection is only indicated by arm-waving in the direction of possible science: the word *risk* now means danger; *high risk* means a lot of danger (Douglas 1992:24, original emphases). Risk is now generally used to relate only to negative or undesirable outcomes, not positive outcomes. This is even often the case in more technical assessments of risk: 'While risk- and cost-benefit analyses focus on both positive and negative potential outcomes, benefits tend to receive short shrift in these analyses, as do positive aspects of risks' (Short 1984:711). In the esoteric parlance of economic speculation, however, there remain such things as 'good risks' in relation to making a profit. It is recognized that risks must be taken to make money in speculative enterprises, and that often the greater the risk of losing one's money, the greater the return should things go well (Luhmann 1993:71).

In everyday lay people's language, risk tends to be used to refer almost exclusively to a threat, hazard, danger or harm: we 'risk our life savings' by investing on the stock exchange, or 'put our marriage at risk' by having an affair. The term is also used more weakly to refer to a somewhat negative rather than disastrous outcome, as in the phrase 'If you go outside in this rain, you'll risk catching a cold.' In this usage, risk means somewhat less than a possible danger or a threat, more an unfortunate or annoying event. Risk is therefore a very loose term in everyday parlance. Issues of calculable probability are not necessarily important to the colloquial use of risk. Risk and uncertainty tend to be treated as conceptually the same thing: for example, the term 'risk' is often used to denote a phenomenon that has the potential to deliver substantial harm, whether or not the probability of this harm eventuating is estimable.

In contemporary western societies, the noun 'risk' and the adjective 'risky' have become very commonly used in both popular and expert discourses. An apparatus of expert research, knowledge and advice has developed around the concept of risk: risk analysis, risk assessment, risk communication and risk management are all major fields of research and practice, used to measure and control risk in areas as far-ranging as medicine and public health, finance, the law, and business and industry. Indeed, the use of the term risk appears to have been increasing over the last few decades. One study of the occurrence of risk in academic journal article titles found an exponential growth between 1966 and 1982, particularly after the early 1970s (Inhaber and Norman (1982), cited in Short 1984:712). A more specific search of some medical and epidemiological journals published in the United States, Britain and Scandinavia also found an increasing frequency of the use of the term over the years spanning 1967 and 1991, in which the late 1970s again marked the beginning of a period of rapid growth in the use of the

term which accelerated in the late 1980s (Skolbekken 1995).

I performed my own quick survey of the use of 'risk' in a more popular medium—a daily Australian newspaper, the Sydney Morning Herald. I used a computerized database compiled by that broadsheet to search for the use of 'risk' in the main text and headlines of all articles published over the six years spanning 1992 to 1997 (the database only went back as far as 1992). Both a high level of use of the term and an increase in its appearance over this relatively short time was found. In 1992, 'risk' appeared 2,356 times in the main text of news stories and 89 times in headlines. The number of mentions steadily increased each year. By 1997, 'risk' appeared 3,488 times in articles (almost half as many again as in 1992) and in 118 headlines. Such findings suggest that risk has definitely become more of a key word in the news media, used in place of such words as 'danger', 'threat' and 'hazard'.

Various reasons have been suggested proliferation of the concept and language of risk in expert discourses over the past few decades. These include developments in probability statistics and computer technologies, allowing the statistical manipulation of large data sets in ways that were not previously possible, and the establishment of institutions and regulatory agencies to deal with such phenomena conceptualized as highly risky, such as nuclear energy. So too, changes in scientific thinking that have moved from paradigms monocausual determinism to those incorporating multiple causes and effects and an increasing value placed on scientific rationality as a basis for certainty have been suggested as focusing more attention on issues of risk (Short 1984; Douglas 1985; Skolbekken 1995). Some commentators have also pointed to changes in the nature of risks themselves as increasing expert and public concern. In the last part of the twentieth century, they argue, risks have become more globalized, less identifiable and more

serious in their effects and therefore less easily manageable and anxiety-provoking (Beck 1992b).

At a deeper level of meaning, it may be argued that the contemporary obsession with the concept of risk has its roots in the changes inherent in the transformation of societies from pre-modern to modern and then to late modern (or postmodern, as some theorists prefer to describe the contemporary era). Late or postmodernity generally refers to broader socioeconomic and political changes that have taken place in western societies since World War Two, producing the sense for many people that we are 'living in new times' (Smart 1993:15). These changes include the end of the Cold War, the breakdown of the socialist and communist states, the spread of communications technology and changes in familial relationships and the workplace wrought by the feminist movement, economic decline and growing secularism. Most theorists agree that post or late modernity is characterized by a growing sense of the failed promises of early or 'simple' modernity and a tendency to challenge the key assumptions of this period, particularly those that unproblematically view science and medicine as the vanguards of progress. Thus, postmodernity has been defined as 'a more modest modernity, a sign of modernity having come to terms with its own limits and limitations' (Smart 1993:12).

Postmodernity, therefore, to a greater or lesser degree, is about a questioning of established thought, expression and practice, a deconstruction of tradition (ibid.: 24). For many commentators, the late or postmodern era is characterized by uncertainty and ambivalence related to constant change and flux, cultural fragmentation and the breakdown of norms and traditions (see, for example, Giddens 1990; Massumi 1993; Lash and Urry 1994; Featherstone 1995). Time and space have become compressed, and there is an increasing rapid circulation of subjects and objects and an emptying-out of the established meaning of things and social relationships. Contemporary western societies have been described as 'post-traditional' (Giddens 1994:56) in that old traditions have been called into question, leaving greater uncertainty and insecurities behind.

All of these changes are seen as contributing to a particular way of understanding the self and the world that differs dramatically from earlier eras. For the individual, it is argued, these changes are associated with an intensifying sense of uncertainty, complexity, ambivalence and disorder, a growing distrust of social institutions and traditional authorities and an increasing awareness of the threats inherent in everyday life. Luhmann (1993: ix-xii) asserts that risk awareness is characterized by a fascination about extremely improbable circumstances with grave outcomes. This is not simply because the technological conditions exist for such circumstances to eventuate, because natural disasters have always occurred. The explanation for this fascination, he claims, is that today the decisions of individuals or organizations can be identified as the root cause of disasters, and therefore it can be demanded that their decisions be opposed so as to obviate danger. The concept of risk has gained importance in recent times because the dependence of society's future on decision making has increased, now dominating ideas about the future.

For several writers on the late or postmodern era, the notion of risk is used as a keyword. It has come to stand as one of the focal points of feelings of fear, anxiety and uncertainty. Concern about risk has become intensified because of a general *fin de siècle*—and indeed a *fin de millenium*—mood of malaise and disorientation, a sense that we are living in a time of endings and major disruptive social change. The term 'panic' has been used by some writers to describe the existential state of living in a postmodern society (Kroker and Kroker 1988). Massumi argues that individuals in late modernity experience a constant low-level fear, which is vague, not as sharp as panic or as localized as hysteria, but rather 'A

kind of background radiation saturating existence' (1993: 24). He sees the assassination of American President John F. Kennedy in November 1963 as marking a divide in American culture: between the optimism of Enlightenment humanism and progress and the pessimism uncertainty of the late modern era:

Cracks began to open all around. There was no longer any safe ground. The shot could come from any direction, at any time, in any form.... Even pleasure no longer felt the same. Smoking was the insidious onset of a fatal ailment. Food becomes a foretaste of heart disease. The body itself was subversive of the 'self'; in the 'youth culture', the very existence of the flesh was the onset of decline.... Industrialization, once the harbinger of progress, threatened the world with environmental collapse.... Everywhere, imminent disaster.

(Massumi 1993:10)

Juxtaposed against this world of change are the meanings and strategies constructed around risk, which both spring from the uncertainties, anxieties and lack of predictability characteristic of late modernity and also attempt to pose solutions to them. Risk meanings and strategies are attempts to tame uncertainty, but often have the paradoxical effect of increasing anxiety about risk through the intensity of their focus and concern.

THIS BOOK

This book analyses the ways in which people in western societies give meaning to and deal with risk. A major premise of the book is that the identification of 'risks' takes place in the specific sociocultural and historical contexts in which we are located. To call something a 'risk' is to recognize its importance to our subjectivity and wellbeing. In some societies at some times, certain phenomena are selected as the focus for anxieties. In other societies and eras, other phenomena become prominent as 'risky'.

For example, at least six major categories of 'risk' that currently appear to predominate in the concerns of individuals and institutions in western societies can be identified. They are 'environmental risks', or those posed by pollution, radiation, chemicals, floods, fires, dangerous road conditions and so on; 'lifestyle risks', those believed to be related to the consumption of such commodities as food and drugs, engagement in sexual activities, driving practices, stress, leisure and so on; 'medical risks', those related to experiencing medical care or treatment (for example, drug therapy, surgery, childbirth, reproductive technologies, diagnostic tests); 'interpersonal risks', related intimate relationships, social interactions, love, sexuality, gender roles, friendship, marriage and parenting; 'economic risks' implicated in unemployment or under-employment, borrowing money, investment, bankruptcy, destruction of property, failure of a business and so on; and 'criminal risks', those emerging from being a participant in or potential victim of illegal activities.

That these categories of risk identified are prominent at this point in the history of western societies is indicative of the nature of the broader sociocultural, political and economic context in which they acquire meaning. Our awareness and knowledge of these risks, and others, contribute to various aspects of subjectivity and social life, including how we live our everyday lives, how we distinguish our selves and the social groups of which we are members from other individuals and groups, how we perceive and experience our bodies, how we spend our money and where we choose to live and work.

Those phenomena that we single out and identify as 'risks', therefore, have an important ontological status in our understandings of selfhood and the social and material worlds. Societies—and within them, social institutions, social groups and individuals—need this

selection process as part of their continued operation. Risk selection, and the activities associated with management of risk, are central to ordering, function and individual and cultural identity. The point of this book is not to engage in discussion about the relative claims to truth of competing arguments about what phenomena should be considered 'real risks' or not, or how serious these risks are. Rather, it is to look at the ways in which the concept of risk operates in western societies at the end of the twentieth century, and its implications for how we think about our selves, others, organizations, institutions and governments and the non-human world.

The term 'reflexivity' is often used in the sociological literature to denote the response of people contemporary western societies to risk, and this term appears throughout the book. Reflexivity means a response to conditions that arouse fear or anxiety that is active rather than passive. Reflexivity is a defining characteristic of all human action, involving the continual monitoring of action and its contexts (Giddens 1990:36-7). It involves the weighing up and critical assessment of institutions and claim-makers, including those who speak with 'expert' voices about risk. As will be elaborated upon later in this book, some commentators argue that reflexivity is an inherent aspect of the contemporary meanings of risk: they are two sides of the same coin. The heightened sensitivity to risk evident in the late modern era is developed through a highly reflexive approach to the world.

'Discourse' is another term that often appears in the book, in my attempts to identify and speculate upon the sociocultural meanings and implications of risk. A discourse may be understood as a bounded body of knowledge and associated practices, a particular identifiable way of giving meaning to reality via words or imagery. Through discourses we perceive and understand the social, cultural and material worlds in which we move. Discourses both delimit and make possible what can be said and done about phenomena such as risk. Thus, for example, it may be said that there are a series of discourses on risk that serve to organize the ways in which we perceive and deal with risk. Discourses are constantly in a state of flux: some come to prominence at certain times but then make way for others, and this has implications for our understanding of and response to phenomena. For example, the discourse on risk that represented it as both 'good' and 'bad' in early modernity has been superseded in late modernity by that which portrays it almost exclusively as 'bad'. Discourse analyses of risk reveal the shifting meanings around risk phenomena and the struggles over these meanings.

The prevalence of uncertainties and anxieties about what are seen to be risks, the nature of discourses on risk and how these influence the ways in which we conduct our selves and our social relationships and how societies are governed, are precisely the issues to which a number of major social and cultural risk theorists have recently directed their attention. The book draws on these theorists to examine and analyse the notion of risk as it is understood in contemporary western societies. Chapter 2 discusses the major ways in which risk perception has been theorized in the social sciences, comparing the technicoscientific perspective that is employed in approaches such as cognitive science with sociocultural critiques, and particularly the social constructionist position on risk. Three major and distinctive theoretical perspectives in contemporary sociocultural theory—the 'cultural/ symbolic', 'risk society' and 'governmentality' perspectives —are introduced in this chapter.

The next three chapters go on to discuss in detail these theoretical perspectives and the views of their major exponents on risk. Chapter 3 overviews the 'cultural' symbolic' perspective presented by anthropologist Mary Douglas and her followers, Chapter 4 discusses the writings of 'risk society' sociologists Ulrich Beck and Anthony Giddens, while Chapter 5 introduces the

Foucauldian-influenced work of the 'governmentality' school. The final group of chapters organize discussions of sociocultural theory around three important topics. Chapter 6 examines the notion of risk subjectivities, discussing the different ways in which people construct their understandings of risk and respond to risk. Chapter 7 looks at the more symbolic concept of Otherness, and its intertwined emotions of fear, disgust, fascination and desire, as they underlie notions of risk. The final chapter, Chapter 8, addresses the relatively neglected area of risk and pleasure, examining the ways in which people may deliberately engage in the transgressions of risk-taking in the attempt to escape the banalities of everyday life.

2 THEORIZING RISK

There are a number of ways in which the phenomenon of risk is addressed in the social scientific literature on risk perception. The most common is the realist perspective, which has developed and is expressed principally in technical and scientific approaches. One major approach adopting this perspective is that of cognitive science, based psychology. An alternative perspective is constructionism. advocated bv those predominantly interested in the social and cultural aspects of risk. This chapter reviews these contrasting perspectives, discussing the epistemologies (knowledges) on which they are based and the different ways in which they represent risk, risk perception and the risk actor.

THE COGNITIVE SCIENCE PERSPECTIVE

Technico-scientific approaches to risk, emerging from such fields as engineering, statistics, actuarialism, psychology, epidemiology and economics, bring together the notion of danger or hazard with calculations of probability. They define risk as 'the product of the probability and consequences (magnitude and severity) of an adverse event [ie. a hazard]' (Bradbury 1989:382). Debates over risk in these technico-scientific fields tend to revolve around issues of how well a risk has been identified or calculated, the level of seriousness of a risk in terms of its possible

effects, how accurate is the 'science' that has been used to measure and calculate risk and how inclusive are the causal or predictive models that have been constructed to understand why risks occur and why people respond to them in certain ways.

Much of the technico-scientific literature addresses what is seen to be the problem of conflict between scientific, industrial and government organizations and the public in relation to the health and environmental risks associated with science, technology and industry. The public is traditionally described as having become progressively concerned about such risks over the past half century or so and as directing a more critical and challenging eye upon the activities of industry and government. The literature on risk addressing this problem tends to attempt to identify the social and psychological factors influencing greater public cynicism and distrust of institutions and lay people's assessment of risk. The objective it seeks is to understanding between the public institutions, and as Brown puts it, 'to provide a route out of the ever-growing bitterness of clashes between affected publics and the managing institutions' (1989:2).

One question that tends *not* to be asked in this research is 'How are risks constructed as social facts?', for the nature of risk is taken for granted. While most practitioners working in probabilistic risk assessment would acknowledge that 'subjectiveness' is an inevitable element of human judgement, and that therefore technical risk assessment is not value-free, the calculations they produce tend to be treated as if they were 'objective facts', or 'absolute truths' (Bradbury 1989:382). Risks, according to this model, are pre-existing in nature and in principle are able to be identified through scientific measurement and calculation and controlled using this knowledge.

In the technico-scientific literature on risk there is sometimes evident an ill-masked contempt for lay people's lack of what is deemed to be 'appropriate' or 'correct' knowledge about risk. Lay people are often portrayed as responding 'unscientifically' to risk, using inferior and unsophisticated sources of knowledge such as 'intuition'. This distinction is found, for example, in the Royal Society's (1992) report on risk, which drew a distinction between 'objective' and 'subjective' risk. It was argued in this report that a range of 'objective' risks exist in any situation, to which individuals and social groups respond in more or less 'subjective' ways.

Exponents of the cognitive science approach, for example, the dominant approach in the social sciences, are primarily interested in using various psychological models of human behaviour to identify the ways in which people respond cognitively and behaviourally to risk. In this literature, 'the hazard is taken as the independent variable and people's response to it as dependent' (Douglas 1985: 25). The 'objective facts' of risk as they are calculated by 'experts' are contrasted with the subjective understandings of lay people, which are seen as being more or less accurate compared with these 'facts'. Such risk calculations tend not to acknowledge the role played by the 'ways of seeing' on the part of the experts themselves that produce such calculations. Their understandings of risk are represented as neutral and unbiased.

A number of cognitive science researchers have sought to identify patterns in the ways in which lay people assess and respond to various risks. Some take a psychometric perspective, seeking to measure the relative influence of different cognitive factors in shaping lay responses. They attempt to identify the 'mental strategies' or 'heuristics' used in making judgements about risk and which are viewed as often leading to 'large and persistent biases' on the part of lay people (Slovic 1987:281). Several psychometric researchers have developed a 'taxonomy' by which hazards are categorized and dealt with cognitively. They have argued that lay people over-estimate and underestimate some categories of risk, and find it difficult to assess risk using probabilities.

For instance, it has been found by psychometric researchers that lay people are more likely to calculate that risk is likely to occur if information related to it is available and easily recalled, and tend to overestimate risk related to circumstances where it can be easily imagined as happening to oneself. It has also been found that people are more likely to be concerned about risks that they see as being close to them. Risks that are seen to be rare but memorable tend to be overestimated while those that are considered to be common and less serious are underestimated. Risks that are perceived as familiar or voluntary are considered more acceptable and less likely to happen than those that are perceived to be new or imposed. People tend to be risk averse when faced with gains and risk seeking when faced with losses. Psychometric researchers have also reported that disasters that receive a high level of media attention arouse more concern than those that do not, even if they are relatively rare occurrences; dangers which are seen to occur in a cluster are considered more serious than an equivalent number of events that happen over a longer period of time; and the consequences of catastrophes that occur immediately arouse more concern than those that are delayed. (See Douglas (1985), Slovic (1987), Heimer (1988), Hansson (1989) and Brown (1989) for reviews of psychometric research).

In one example of a psychometric study of risk perception (Marris and Langford 1996), the researchers asked a sample of English people to assess the seriousness of common risks associated with the following: sunbathing, food colouring, genetic engineering, nuclear power, mugging, home accidents, ozone depletion, car driving, microwave ovens, AIDS, war, terrorism and alcoholic drinks. They chose these phenomena because it was possible to categorize the potential harms according to two variables: to what extent possible harmful effects are delayed and have catastrophic potential, and whether the harm is seen as imposed and unavoidable or

voluntary. The findings supported previous psychometric models of risk perception. Familiar and voluntary potential hazards from microwave ovens, food colouring and alcohol were all seen as not particularly risky, while 'catastrophic' dangers such as war, genetic engineering, ozone loss and nuclear power were rated as highly risky. The researchers also found that people most feared risks that they perceived as unacceptable to society. They often used the term 'dread' to describe risks they considered involuntary, unfairly imposed upon individuals or highly likely to kill or permanently injure people.

Psychometric risk analyses are founded on a theory of rational behaviour and the notion of the ideal rational investigator and the rational, risk-perceiving agent: 'Both are driven to seek order in the world; both recognize inconsistency; both assess probability' (Douglas 1985:28). They represent the individual as an information-processing unit, taking in information about risk and dealing with it in certain ways that are typically represented as biased or limited. The 'health belief model', which dominates ideas about risk perception in the fields of health promotion and health education, is another example of this approach. Like other such models of behaviour, it represents human action as volitional and rational, invariably categorizing risk avoidance as rational and risk-taking as irrational (Bloor 1995: Chapter 5). According to the health belief model, a number of perceptions have to be in place before an individual will take steps to protect herself or himself from a health threat. Individuals must see themselves as vulnerable to the threat, they must perceive the threat as having serious consequences, they must believe that taking preventive action will be effective and they must believe that the benefits of that action will outweigh the costs. This model relies upon an understanding of the human actor in which there is a linear relationship between knowledge of a risk, developing the attitude that one is at risk and adopting a practice to prevent the risk happening to oneself.

The cognitive science approach, therefore, constructs individuals as calculating and emotion-free actors. assuming that they all share the responses and preferences of the actor in utilitarian philosophy:

Warm-blooded, passionate, inherently social beings though we think we are, humans are presented in this context as hedonic calculators calmly seeking to pursue private interests. We are said to be riskaversive, but, alas, so inefficient in handling information that we are unintentional risk-takers; basically we are fools.

(Douglas 1992:13)

Douglas further asserts that cognitive science presents too narrow a view of rational action, so that anything outside this becomes viewed as irrational: 'So instead of a sociological, cultural, and ethical theory of human judgement, there is an unintended emphasis on perceptual pathology' (1985:3).

An epistemological uncertainty is evident in relation to the concept of risk as it is used in psychometric and psychological research endeavours, even though this does not tend to be openly acknowledged by its exponents. As Bradbury points out: 'In examining the individual's response to risk, this research provides a subjectivist interpretation within a realist paradigm' (1989:384). Some psychometric researchers argue that 'actual risks' exist but they can only ever be interpreted as perceptions. Others continue to argue that some definitions of risk (particularly those made by 'experts') are 'real' and 'correct'. Some, confusingly, include both concepts of risk in their work interchangeably (ibid.: 384).

Another difficulty with such approaches is that they tend to reduce the meanings and behaviours associated with risk perception and assessment to the individualistic level. Cognitive science does not generally take into account the symbolic meanings, created through the social

world, that humans give to things and events. Perception is limited to how humans see and understand the world through their senses and brain-functioning, without acknowledging the ways in which cultural conceptual categories mediate judgement. People tend to be positioned outside the cultural and political frameworks, relationships and institutions within which they construct their beliefs and engage in behaviours. In such research, individuals are therefore represented as atomized and selfinterested, ideally behaving in response to their carefully considered calculations of risk as it affects them individually. They are portrayed as free actors who are constrained only by their ignorance about the threat to which they may be exposed or their lack of self-efficacy in feeling able to do something about a risk. So too, risks and behaviours that are associated with such risks are themselves singled out for attention in this literature as often separate from other risks and behaviours. This is convenient for psychometric statistical testing and modelling, but reduces and over-simplifies such phenomena.

Nonetheless, it would appear that at least some researchers using a psychometric approach have recently attempted to consider issues of social and cultural group membership in people's responses to risk. Research has shown, for example, that members of social groups that are less powerful tend to be more concerned about risks than members of powerful social groups. Women compared with men, and non-whites compared with whites tend in quantitative surveys to rate their concern about specific risks more highly. For instance, in one quantitative study in which a sample of Americans to rate the public health risks associated with each of 25 hazards, Flynn et al. (1994) found that white men produced risk perception ratings that were consistently much lower than white women, non-white men and non-white women. In a similar study, Graham and Clemente (1996) found that again, white men were more sceptical than other groups

about the threat to human health posed by the risks to which they were asked to give ratings. Both studies showed that of white men, those who had higher levels of education, higher household incomes and were politically more conservative were most likely to rate risks as less serious. Both studies concluded that 'further research was needed' on sociopolitical factors and risk perception, particularly the ways in which power, social status and political orientation influence people's views on risk.

It is evident, too, that since the late 1980s some sociocultural perspectives on risk—particularly those by Mary Douglas—have been taken enthusiastically by some researchers in cognitive science and other techno-scientific fields interested in risk perception and risk management. This usage, however, has been somewhat selective and has, at times, somewhat distorted Douglas' views. (See the discussion in Chapter 3 following a detailed review of Douglas' writings.)

SOCIOCULTURAL PERSPECTIVES

Sociocultural perspectives on risk emphasize the very aspects that cognitive science and other technico-scientific approaches have been criticized for neglecting: the social and cultural contexts in which risk is understood and negotiated. They have emerged from such disciplines as cultural anthropology, philosophy, sociology, social history, cultural geography and science and technology studies.

As I noted in Chapter 1, theorists on the sociocultural dimensions on risk could loosely be categorized into at least three major groups based on the perspective adopted. The perspective advanced by anthropologist Mary Douglas and her colleagues and followers constitutes the 'cultural/symbolic' perspective. A second group is that of the 'risk society' theorists, dominated by the sociologists Ulrich Beck and Anthony Giddens. A third group may be

described as the 'governmentality' theorists, who draw on the writings of the French philosopher Michel Foucault.

scholars adopting the 'cultural/symbolic' perspective direct their attention to the ways in which notions of risk are used to establish and maintain conceptual boundaries between self and Other, with a particular interest in how the human body is used symbolically and metaphorically in discourses and practices around risk. The sociologists adopting the 'risk society' perspective are predominantly interested in the macro-social processes they see as characteristic of late modern societies and their relation to concepts of risk. These processes include reflexive modernization, or the move towards criticism of the outcomes of modernity, and individualization, or the breaking down of traditional norms and values. 'Governmentality' scholars have taken up in particular Foucault's insights on governmentality and on ethical self-formation to explore risk in the context of surveillance, discipline and regulation of populations, and how concepts of risk construct particular norms of behaviour which are used to encourage individuals to engage voluntarily in self-regulation in response to these norms.

While, as will be explained in detail in later chapters, there are some major differences between these perspectives, their exponents all tend to argue that there are a number of important new features in notions of risk in contemporary western societies. They see risk as having become a central cultural and political concept by which individuals, social groups and institutions are organized, monitored and regulated. The following insights are central to all three major perspectives: risk has become an increasingly pervasive concept of human existence in western societies; risk is a central aspect of human subjectivity; risk is seen as something that can be managed through human intervention; and risk is associated with notions of choice, responsibility and blame.

There is another loose typology that can be used to characterize the more discrete epistemological and methodological positions taken up within sociocultural investigations into risk. This typology distinguishes structuralists. poststructuralists. phenomenologists and psychoanalytic theorists.

Structuralists approach their analyses of risk primarily by seeking to identify the ways in which underlying cultural structures, hierarchies and categories serve to define risk knowledges and practices. They may adopt a functional structuralist approach, interested in how social and cultural structures and systems serve to maintain social order and the status quo and deal with 'deviance', or divergence from accepted norms and social rules concerning behaviour. Mary Douglas and some of her followers may be categorized as functional structuralists (see Chapter 3). Alternatively, a critical structuralist approach may be taken, which builds on the Marxist critical legacy to focus more on social conflict, inequities and dissent and the need for social change in relation to risk. Critical structuralists tend to be interested in critiquing the ways in which social institutions (such as government, the economic system and the legal system) wield power over individuals, reducing their capacity for agency and autonomy. The work of Ulrich Beck and Anthony Giddens (discussed in Chapter 4) largely adopts a critical structuralist approach.

The poststructuralist perspective, as it has been adopted to explore issues of risk, builds largely upon Foucauldian theory (Chapter 5 discusses Foucauldian perspectives on risk in greater detail). Poststructuralism emphasizes the importance of identifying the discourses that participate in the construction of notions of realities, meanings and understandings. Exponents of post-structuralism tend to focus less on what they see to be the overly rigid definitions of structures identified in structuralism. They are more interested in change and flux in social structures and meanings. There is also a different view on power relations in poststructuralist writings than is generally put forward in structuralist accounts. A central preoccupation of poststructuralists is with the relationship between power and knowledge. They point out that power relations are always implicated with knowledge and that no knowledge, therefore, can be said to be 'neutral' or 'disinterested' (including poststructuralists' own truth claims). Individuals are seen not to be fixed in social or cultural identities. but constantly shifting, the products of various combinations of power-knowledge formations. Power is seen as operating through manifold sites, rather predominantly through monolithic institutions. Power is seen as not simply coercive or oppressive, as critical structuralism tends to have it, but also as productive and inevitably present in any social relation.

Phenomenological or hermeneutic accounts of risk are interested in the situated meanings that are given to risk. Exponents of this perspective, therefore, are less interested in the macro-structures organizing and constraining the meanings of risk than are the structuralists. Rather, they turn their attention towards 'lived experience', or how individuals experience their world as an interpretive reality with the use of shared commonsense meanings and knowledges. Qualitative research methods such as indepth interviews with respondents about their understandings and experiences of risk is the basis of phenomenological inquiries. Phenomenologists argue that the meanings of risk differ from locale to locale: that is, in the microcontext of risk meanings. Phenomenological accounts examine how specific actors (or sub-groups) within a certain sociocultural setting construct their risk understandings as part of their interactions with others, albeit within the broader frame of social structures. Meaning is not simply drawn from the social environment, therefore: it also works the other way, with social actors influencing their environments. Chapters 6 and 8 draw upon phenomenological investigations in addressing issues of risk and subjectivity and the pleasurable aspects of risk-taking.

The psychoanalytic theoretical perspective is interested in exploring the unconscious psychodynamic processes which mediate people's responses to other people and to objects and events. In relation to ideas about risk it is perhaps best represented in the work of Julia Kristeva and Elizabeth Grosz, both feminist psychoanalytic theorists who have built upon and extended Mary Douglas' While Kristeva and Grosz theories. explicitly discuss the topic of risk, many of observations on the notion of the 'abject' have strong relevance for a understanding of the ways in which notions of risk are linked to those of the body and Otherness. The abject is that which both disgusts or horrifies and fascinates us, which we seek to expel from our bodies and our selves as part of maintaining our sense of autonomy and individuation. As such, the abject is fraught with symbolic riskiness, in the sense of challenging our sense of subjectivity and individuality, our ability to demarcate ourselves from others, our feelings of purity and containment. (See Chapter 7 for further elaboration of this concept in relation to risk.)

SOCIAL CONSTRUCTIONIST **POSITIONS**

In the sociocultural literature there are a range of positions on risk that are taken up, with some falling more towards a relativist and others more towards a realist approach, and yet others somewhere in the middle. Sometimes risk is talked about as if it were based on objective facts about dangers and hazards, amenable to rationalistic calculation, which are then mediated, perceived and responded to in particular ways via social, cultural and political processes. This may be described as the 'weak' social constructionist thesis, which overlaps to some extent with those psychometric and psychological studies (some of which are discussed above) which go beyond a focus on individuals to directing attention at social group membership and power relations. Exponents of the more relativist perspective, or the 'strong' social constructionist position, contend, as Ewald has put it, that: 'Nothing is a risk in itself; there is no risk in reality. But on the other hand, anything *can* be a risk; it all depends on how one analyzes the danger, considers the event' (1991:199, original emphasis). As I will go on to demonstrate in later chapters, the 'risk society' approach tends to waver uncertainly between a realist and weak constructionist approach, while the 'cultural/symbolic' approach is somewhat more towards the strong end of constructionism and exponents of the 'governmentality' approach generally adopt an even stronger relativist position.

Those who have adopted social constructionism, regardless of the strength of their position, tend to argue that a risk is never fully objective or knowable outside of belief systems and moral positions: what we measure, identify and manage as risks are always constituted via pre-existing knowledges and discourses. This approach to risk is indebted to writings in the sociology of knowledge, the sociology of science and technology and theorizing from poststructuralist perspectives. Social constructionists argue that humans and their social world exist in a dialectical relationship, in which each creates the other. Although the material and social worlds are experienced by most individuals as objective, pre-existing realities, these realities involve the reproduction of meaning and knowledges through social interaction and socialization and rely upon shared definitions. Because of the continually constructed nature of reality, its meanings are precarious and subject to change.

From the constructionist perspective, all knowledge about risk is bound to the sociocultural contexts in which this knowledge is generated, whether in relation to scientists' and other experts' knowledges or lay people's knowledges. Scientific knowledge, or any other knowledge,

is never value-free but rather is always the product of a way of seeing. A risk, therefore, is not a static, objective phenomenon, but is constantly constructed and negotiated as part of the network of social interaction and the formation of meaning. 'Expert' judgements of risk, rather than being the 'objective' and 'neutral' and therefore 'unbiased' assessments which they tend to be portrayed in the technico-scientific literature, are regarded as being equally as constructed through implicit social and cultural processes as are lay people's judgements. For social constructionists, it is not a matter of doing more research to obtain a clearer view of exactly to which risks people are exposed. Instead, the primary focus is on examining how concepts of risk are part of world views. There is a cultural pattern in the ways in which certain phenomena are identified and dealt with as 'risks' and this pattern is subject to change over time and space. Rather than seeing 'risks' as realities lying outside of society and culture, therefore, they can be viewed as assembledges of meanings, logics and beliefs cohering around material phenomena, giving these phenomena form and substance. We can only ever know and experience risks through our specific location in a particular sociocultural context. This approach to risk highlights the importance of understanding the embeddedness of understandings and perceptions of risk, and emphasizes that understandings and perceptions often differ between actors who are located in different contexts and thus bring competing logics to bear upon risk.

Debates about risk always involve questions of cultural representation and meaning and political positions. The weak social constructionist position sees risks as cultural mediations of 'real' dangers and hazards. For the strong social constructionist position, by contrast, a 'hazard' or 'danger' itself is also seen as socially constructed, coming into social existence when human actors recognize and label it as such (Fox 1999). Judgements about risk, therefore, are not simply cultural interpretations of

objective dangers or hazards. What is deemed a 'danger' or 'hazard' in one historical or cultural context may not be so identified in another, and this has implications for how knowledges and understandings about developed. Hilgartner (1992) argues that constructionist accounts have tended to neglect the social construction of what he calls 'risk objects' (things, activities or situations to which harmful consequences are conceptually attached) or to examine systematically the construction of networks of causal attribution that links chains of risk objects to harm or danger. To become risk objects, he claims, objects must first become constructed as 'objects', and then as 'risky', or identified as the cause of harm or danger. The process of defining harm or danger is a third construction in this linkage.

The extent to which objects may be linked with each other and with harm in a causal attribution model is potentially infinite: anything may be defined as a 'risk'. What is of importance for a sociocultural analysis of risk is the ways in which certain linkages are defined. The task of constructing a risk object is essentially a rhetorical process, performed in specialized texts or in public arenas and usually involves building networks of heterogeneous risk objects. It often involves intense struggles over meaning, particularly in relation to those actors who are deemed to be responsible for the risk object. These struggles are complemented by struggles with a variety of human and non-human actors to identify and control risk objects (Hilgartner 1992).

For example, in a scare over the quality of the water supply in Sydney in mid 1998, two parasitic microorganisms, cryptosporidium and giardia, were identified in routine testing of the water in levels that were deemed to potentially pose a health risk to consumers. A substance that had previously been treated as risk-free—tap water—turned into a risk object. All Sydney residents were warned to boil tap water before drinking it or using it for other purposes such as brushing their teeth to avoid infection

that could lead to gastric upsets. Sales of bottled water increased dramatically during this time. advertisements for water filters promising to remove the micro-organisms from water and for law firms seeking clients who wished to sue the water authority because of illness or financial loss appeared in newspapers.

After a week, the micro-organism content of the water was eventually deemed to have dropped to acceptable levels and consumers were encouraged to go back to using the water straight from the tap. Much blame was pointed at the authority responsible for ensuring water quality for posing such a risk to and alarming their clients, particularly in news media accounts, some of which that Sydney was experiencing World' conditions. Yet some scientists made the point that these micro-organisms could well have been in the water supply at such levels before, and indeed could never be totally eradicated from the water. It was only because the water authority had begun testing for the micro-organisms that they had now been identified and concern over the possible risk to consumers was generated. There was also some debate among experts as to what constituted a 'safe' and a 'risky' level of the micro-organisms in the water supply. In the construction of this new risk in the water supply, therefore, a number of human actors—including water authority officials, members of the news media, scientific experts and members of the legal profession interacted in shaping the meanings of risk, together with non-human actors—the water, the micro-organisms and the technologies for testing the water.

As this example suggests, it is rarely lay people who play a major role in the construction of risk objects at the level of public debates. Rather, 'expert' knowledgesparticularly those emerging from science, medicine, the 'psy' disciplines (psychology, psychiatry, counselling), social work, the law and economics-embedded within organizational contexts and often mediated through the mass media, are central to the construction and publicizing

of risk. Debates among scientists and other experts often occur around such uncertainties as what is considered adequate 'proof that a phenomenon is hazardous, how acceptable the level of hazard is and what the consequences of attempting to control that hazard might be. While disputes over the validity of technical data may contribute to these debates, at a more fundamental level different systems of values and ways of seeing shape experts' judgements of these data. Experts, in seeking validity for their knowledge claims, do not tend to acknowledge the situated and localized nature of their risk calculations and prognoses, however, preferring to represent them as objective universal truths. Neither do they acknowledge that their knowledges are culturally shaped, evincing particular assumptions about the 'natural', the 'cultural' and what it means to be human (Wynne 1996).

If a 'risk' is understood as a product of perception and cultural understanding, then to draw a distinction between 'real' risks (as measured and identified by 'experts') and 'false' risks (as perceived by members of the public) is irrelevant. Both perspectives are describing forms of risk, and both lead to certain actions. It is the ways in which these understandings are constructed and acted upon that is considered important, not the extent to which one perspective may be considered to be more 'accurate' or less 'biased' than the other, for this distinction is also considered to be irrelevant. The questions that might be asked about risk from the constructionist perspective, therefore, are very different from those asked from the technico-scientific perspective. They include the following (adapted from Hall 1997:45-6): What statements are used to construct certain kinds of knowledges about risk at a particular historical moment and sociocultural setting?; What rules prescribe certain ways of talking about risk and exclude other ways?; What types of subject are constructed through risk discourses?; How does knowledge about risk acquire authority, a sense of

embodying the 'truth' about it?; What practices are used in institutions and by individuals for dealing with the subjects of risk discourses?; and How do new discourses on risk emerge, sup-planting other discourses, and what are the effects of this for risk knowledges and subjects of risk?

CONCLUDING COMMENTS

This chapter has provided a review of various epistemological and methodological approaches analysing the role of risk in subjectivity and social relations. Table 2.1 provides a model of how these different approaches may be mapped against each other, based on the notion of a continuum that moves from the realist position at one pole 'strong' social to constructionist, or relativist, positions on risk, at the other pole. Any such schema will inevitably be somewhat reductive—some approaches to risk, for example, may combine aspects of more than one perspective rather than being able to be neatly slotted into a specific category. Nonetheless, the model is useful as a device in demonstrating the links between the epistemological position and associated perspectives and theories that have been discussed in this chapter, as well as outlining the key questions about risk that are asked from the various approaches. Later chapters build upon this foundational discussion in providing greater detail about various approaches to risk in sociocultural theorizing and research

Table 2.1 The continuum of epistemological approaches to risk in the social sciences

Epistemological position	Associated perspectives and theories	Key questions
Realist. Risk is an objective hazard, threat or danger that exists and can be measured independently of social and cultural processes, but may be distorted or biased through social and cultural frameworks of interpretation.	Technico-scientific per- spectives/most cogni- tive science theories.	What risks exist? How should we manage them? How do people respond cognitively to risks?
Weak constructionist: Risk is an objective hazard, threat or danger that is inevitably mediated through social and cultural processes and can never be known in isolation from these processes.	'Risk society' perspectives/critical structuralism/some psychological approaches.	What is the relation- ship of risk to the structures and processes of late modernity? How is risk understood in different sociocultural contexts?
	'Cultural/symbolic' perspectives/functional structuralism, psychoanalysis, phenomenology.	Why are some dangers selected as risks and others not? How does risk operate as a symbolic boundary measure? What are the psychodynamics of our risk responses? What is the situated context of risk?
Strong constructionist: Nothing is a risk in itself – what we under- stand to be a 'risk' (or a hazard, threat or dan- ger) is a product of his- torically, socially and politically contingent 'ways of seeing'.	'Governmentality' perspectives/post- structuralism.	How do the discourses and practices around risk operate in the construction of subjectivity and social life?

3 RISK AND CULTURE

The cultural anthropologist Mary Douglas has been a pivotal figure in sociocultural analyses of risk, and she is the primary exponent of and influence in 'cultural' symbolic' perspectives on risk. Douglas' approach to risk is best understood as part of a trajectory of theorizing on the body, selfhood and the regulation of contamination and danger that she began three decades ago, in which the symbolic aspects of judgements about danger, pollution and Otherness were identified. Risk, for Douglas, is a contemporary western strategy for dealing with danger and Otherness. Much of her writings on risk seek to explain why it is that some dangers are identified as 'risks' and others are not. Her main explanations revolve around the importance for social groups, organizations or societies to maintain boundaries between self and Other, deal with social deviance and achieve social order. As such, her approach and those of her followers may be described as taking a functional structuralist approach to risk.

This chapter reviews the most integral aspects of Douglas' work on risk. It begins with a discussion of the importance of culture in constructions of risk, moves onto issues concerning purity, danger and the body, then looks at the relationship between risk and blame. The chapter concludes with discussion of the concept of grid-group as it structures cultural responses to risk.

THE IMPORTANCE OF CULTURE

In her extensive writings on risk, Douglas is trenchant in her critique of cognitive scientific and other technicoscientific approaches to understanding risk. She particularly critical of the individualistic approach taken by the psychological researchers dominating risk perception research in their focus on processes of cognition and choice. Douglas contends that: 'The professional discussion of cognition and choice has no sustained theorizing about the social influences which select particular risks for attention. Yet it is hard to maintain seriously that perception of risk is private' (1985: 3). The difference that is commonly observed between 'expert' and 'lay' judgements of risk is founded not in the fact that lav people cannot think in terms of probabilities. as some psychometric risk analysts have contended (see Chapter 2), but rather that other concerns are brought to bear in the ways they judge risk. These concerns are essentially cultural rather than individual: 'individuals do not try to make independent choices, especially about big political issues. When faced with estimating probability and credibility, they come already primed with culturally learned assumptions and weightings' (Douglas 1992:58).

So too, in discussing why people may prefer to engage in activities they know to be labelled as 'risky', Douglas argues that: 'A refusal to take sound hygienic advice is not to be attributed to weakness of understanding. It is a preference. To account for preferences there is only cultural theory' (ibid.: 103). Lay responses to risk should not be considered as erroneous or biased if they differ from expert assessments. Rather, their use and value within a particular cultural context needs to be acknowledged. Douglas (1985:80) further argues that the heuristics or mental models that people use to make judgements about risks should not be considered merely as 'cognitive aids for the individual decisionmaker', as the psychometric perspective would have it. Instead, they should be

regarded as shared conventions, expectations and cultural categories that are founded on clear social functions and responsibilities. She (ibid.: 81) describes culture as a 'mnemonic system' which helps people calculate risks and their consequences. Not only does culture help people understand risk, it also contributes to a communal rather than an individualistic notion of risk, taking into account mutual obligations and expectations: 'A community uses its shared, accumulated experience to determine which foreseeable losses are most probable, which probable losses will be most harmful, and which harms may be preventible. A community also sets up the actors' model of the world and its scale of values by which different consequences are reckoned grave or trivial' (ibid.: 69).

Douglas emphasizes the cultural relativity of judgements about risks, including the differences between groups within the same culture in terms of what is considered a risk and how acceptable it is thought to be. Traditional risk research ignores the conceptual, ethical and moral difficulties around the definition of equality and justice —'each type of society has its custom-built ethical system' (ibid.: 15)—and thus fails to acknowledge or address the related problem of how risk is to be judged acceptable or not. It is pointless, therefore, to concentrate on providing 'better' communication or more education about risk to the lay public as a means of settling risk disputes, for the issue is not one of misguided perception but rather is the result of clashes in political, moral and aesthetic judgements on risk.

As is evident from the title of her most recent collection of essays on risk, Risk and Blame: Essays in Cultural Theory (1992), Douglas' approach to risk strongly emphasizes the political use of the concept of risk in attributing blame for danger threatening a particular social group. She argues that risk is intimately related to of politics, particularly in accountability, responsibility and blame. She is also interested in how risk is a selective process: why some risks are ignored or downplayed while others are responded to with high anxiety, fear or anger.

Despite her emphasis on the importance of culture, Douglas demonstrates a 'weak' rather than a 'strong' constructionist (Chapter 2) approach to risk. She argues that a range of risks or dangers exist in the real world, emphasizing that 'the reality of dangers is not at issue. The dangers are only too horribly real, in both cases, modern and pre-modern. This argument is not about the reality of dangers, but about how they are politicized. This point cannot be emphasized too much' (1992:29). Certain dangers are selected out from others for attention by a society and entitled 'risks' for certain reasons that make sense to a particular culture, based on its shared values and concerns. In other words, Douglas sees risk as a socially constructed interpretation and response to a 'real' danger that objectively exists, even if knowledge about it can only ever be mediated through sociocultural processes.

PURITY, DANGER AND THE BODY

Douglas' writings on risk build on ideas put forward in her earlier writings, particularly her influential work Purity and Danger (first published in 1966). In this book, an anthropological analysis of ideas and rituals concerning pollution and cleanliness in a range of societies is a Adopting structuralist theoretical framework, Douglas set out to explain how taboos act in cultures to protect them from behaviours that threaten to destabilize them. As will be argued later in the chapter, Douglas' theorizing about purity, pollution and danger underpin her understanding of the cultural role and importance of risk in contemporary western societies, particularly the use of risk as a concept for blaming and marginalizing an Other who is positioned as posing a threat (and thus a risk) to the integrity of self.

Central to Douglas' ideas about the symbolic nature of purity and pollution strategies is her insight that the

human, fleshly body is a conceptual microcosm for the body politic (or the community of which it is a part). This is particularly the case in relation to how the flow of phenomena in and out of both bodies' openings is symbolically conceptualized and controlled and how boundaries between 'inside' and 'outside' are constructed and policed. Just as the human body is conceptualized as having certain boundaries between inside and outside, so too the notion of society sees it as having form, external boundaries, margins, internal structure: The body is a model which can stand for any bounded system. Its boundaries can represent any boundaries which are threatened or precarious. The body is a complex structure. The functions of its different parts and their relation afford a source of symbols for other complex structures' (Douglas 1966/1969:115). Notions of the body and its openings and boundaries address a major preoccupation of human societies: how to deal with the threats to order and stability posed by disorder, contamination and pollution.

For Douglas, bodily control is an expression of social control. If social controls over boundaries are relaxed, so too are controls over the openings of the individual's body. One of the central problematics of *Purity and Danger* was the identification of fixed systems by which notions of hygiene are understood and upon which they are acted. Thus, for example, ideas about what substances should be incorporated in the fleshly body—what is pure, and therefore safe, to ingest-mirror notions about the body politic and how the boundaries of a society are maintained, regulating the entry of certain types of people 'in' and keeping others 'outside' the body politic. Ideas about order and disorder fundamentally underlie beliefs about what is 'dirty' and what is 'clean'.

Classification systems in response to purity and contamination operate in all cultures. In some cultures, menstrual blood is considered especially defiling, while in others death pollution is a constant preoccupation (Douglas 1966/1969:121). In 'primitive' cultures, however, these classification systems have more force and totality, whereas in modern urbanized cultures, they are more fragmented and dispersed and thus hold less power (1966/1969: 40). Pollution beliefs are therefore arbitrary, but within their own cultural context reflect deeper anxieties and fears.

Contemporary European notions of dirt and cleanliness, for example, are not so much the effect of concerns about microorganisms than symbolic concerns, for dirt

is essentially disorder. There is no such thing as absolute dirt: it exists in the eye of the beholder. If we shun dirt, it is not because of craven fear, still less dread or holy terror. Nor do our ideas about disease account for the range of our behaviour in cleaning or avoiding dirt. Dirt offends against order. Eliminating it is not a negative movement, but a positive effort to organise the environment.

(Douglas 1966/1969:2)

Thus shoes are not dirty in themselves, but become dirty if placed on a dining table, because they are 'out of place'; food becomes dirt if bespattered on clothing or left on plates after eating is completed. Dirt is found to be offensive and disturbing because it threatens the 'proper' separateness of the individual from other things and people, it bespeaks intermingling, the breaking down of boundaries. It is particularly at the margins of the body/ society that concerns and anxieties about purity and danger are directed. Because margins mark and straddle boundaries, they are liminal and therefore dangerous, requiring high levels of policing and control:

All margins are dangerous. If they are pulled this way or that the shape of fundamental experience is altered. Any structure of ideas is vulnerable at its margins. We should expect the orifices of the body to

symbolise its specially vulnerable points. Matter issuing from them is marginal stuff of the most obvious kind. Spittle, blood, milk, urine, faeces or tears by simply issuing forth have traversed the boundary of the body. So also have bodily parings, skin, nail, hair clippings and sweat. The mistake is to treat bodily margins in isolation from all other margins.

(Douglas 1966/1969:121)

Rituals of purity and impurity serve to contain disorder, to support and bolster social ties, to 'create unity in experience' in particular cultural settings (ibid.: 2). Douglas (ibid.: 123-4) gives the example of the Coorgs, a Hindu caste living in a mountainous region in India, who take great care in policing their body boundaries, treating the body 'as if it were a beleaguered town, every ingress and exit guarded for spies and traitors' (ibid.: 123). The Coorgs do not allow anything that issues from the body to re-enter it, for this is seen to be the most dangerous pollution. Their culture itself is isolated, having only occasional contact with the world around it. The Coorgs' obsession with policing the boundaries of their own bodies is mirrored in their anxiety about the contact that their community has with the outside, and the necessity of protecting the integrity, unity and survival of their minority group.

Ideas about pollution therefore operate at two levels of meaning. At the first level of meaning, which is largely instrumental, pollution ideas reinforce social pressures and rules, uphold moral values and support political power. Pollution ideas may be used as threats to maintain social order:

At this level the laws of nature are dragged in to sanction the moral code: this kind of disease is caused by adultery, that by incest; this meteorological disaster is the effect of political disloyalty, that the effect of impiety. The whole universe is harnessed to men's [sic] attempts to force one another into good citizenship.

(Douglas 1966/1969:3)

On a more symbolic level, pollution beliefs act as analogies for broader concerns about the social system, mirroring ideas about hierarchy or symmetry in social relations: 'Reflection on dirt involves reflection on the relation of order to disorder, being to non-being, form to formlessness, life to death' (ibid:5). According to Douglas' schema, there are four kinds of social pollution within a social system or community. The first kind relates to a danger that threatens the external boundaries of the community, while the second comes from transgressing the internal boundaries of the community. The third kind of social pollution is engendered along the margins of boundaries, while the fourth is created by internal contradictions within the community 'when some of the basic postulates are denied by other basic postulates, so that at certain points the system seems to be at war with itself (ibid.: 122). Pollution rules are often associated with moral codes. Indeed, pollution beliefs often serve to punish wrongdoing and thus support the moral system.

When moral principles lag, pollution beliefs can be used to bolster them by providing another rationale for disapproval. This may operate in four ways:

- (i) When a situation is morally ill-defined, a pollution belief can provide a rule for determining *post hoc* whether infraction has taken place, or not.
- (ii) When moral principles come into conflict, a pollution rule can reduce confusion by giving a simple focus for concern.
- (iii) When action that is held to be morally wrong does not provoke moral indignation, belief in the harmful consequences of a pollution can have the effect of

- aggravating the seriousness of the offence, and so of marshalling public opinion on the side of the right.
- (iv) When moral indignation is not reinforced by practical sanctions, pollution beliefs can provide a deterrent to wrong-doers.

(Douglas 1966/1969:133)

points out the special stigmatized and contaminating position of someone who is seen to transgress cultural boundaries: 'A polluting person is always in the wrong. He [sic] has developed some wrong condition or simply crossed some line which should not have been crossed and this displacement unleashes danger for someone' (ibid.: 113). Polluting people, therefore, are seen as wicked both because they have transgressed cultural norms or taboos and because they place others in danger by their actions.

All cultural classification systems have anomalies, things that do not fit, and ambiguities, things that may fit in more than one category. Anomalies and ambiguities are considered with anxiety, and are therefore treated as 'risky', or threatening. Douglas (ibid.: 38) gives the example of treacle in western societies, which as a viscous substance breaches the border between the liquid and the solid, and in its liminal status, its stickiness, its attacking the boundaries between oneself and it, is greeted with feelings of unease and repulsion. All cultures have ways of dealing with these anomalies and ambiguities. One way to deal with ambiguity is to classify a phenomenon into one category only and maintain it within the category, thus reducing the potential for uncertainty. Another method of dealing with anomaly is to physically control it, removing it. A third way is to avoid anomalous things by strengthening and affirming the classification system that renders them anomalous. Alternatively, anomalous events or things may be labelled dangerous. Lastly, they may be dealt with by being used in ritual, as in poetry and

mythology, 'to enrich meaning or call attention to other levels of existence' (ibid.: 40).

To summarize, the notion of 'dirt' and the related notions of 'contamination', 'pollution' and 'defilement' are inherently fraught with ideas about danger and risk. These notions spring from cultural concepts concerning boundaries, classifications and categories, the violation of which confounds cultural values and expectations. In this approach, 'risk' may be understood as the cultural response to transgression: the outcome of breaking a taboo, crossing a boundary, committing a sin. At the heart of these 'risks' are the emotional dimensions of transgression: anger, anxiety, frustration, hatred, rage, fear. As will be discussed in further detail in Chapter 7, transgression also evokes conflicting emotions and feelings, such as fascination, excitement and desire.

RISK AND BLAME

In her book *Risk Acceptability According to the Social Sciences* (1985), Douglas develops some of her insights from *Purity and Danger* in focusing directly on the sociocultural nature of risk. She therefore extends her earlier argument that 'humans pay attention to a particular pattern of disasters, treating them as omens or punishments' (ibid.: 2). In *Risk and Culture* (1982), written by Douglas with Aaron Wildavsky, the focus was primarily on western societies' preoccupation with technological and environmental hazards.

Douglas (1985:60) argues that in a situation where potential dangers dog one's every action and choice, the risks that receive most attention in a particular culture are those that are connected with legitimating moral principles. Like the distinctions drawn between dirty and pure objects or actions, danger is explained *qua* risk using cultural frames that are inevitably moral and political, and which rely on identifying responsibility for risk. In contemporary western cultures, every death, every

IKE 4/

accident and every misfortune is 'chargeable to someone's account'—someone must be found to be blamed:

Whose fault? is the first question. Then, what action? Which means, what damages? what compensation? what restitution? and the preventive action is to improve the coding of risk in the domain which has turned out to be inadequately covered. Under the banner of risk reduction, a new blaming system has replaced the former combination of moralistic condemning the victim and opportunistic condemning the victim's incompetence.

(Douglas 1992:16)

The central tenet underlying cultural understandings of risk, therefore, is that 'in all places at all times the universe is moralized and politicized. Disasters that befoul the air and soil and poison the water are generally turned to political account: someone already unpopular is going to be blamed for it' (ibid.: 5). This tenet underpins Douglas' (ibid.: 5-6) discussions on the 'forensic theory of danger', or how people, as members of certain social groups or communities, explain misfortune by looking back to what might have caused it. Here her earlier analyses of social pollution clearly influence her thinking. One type of explanation for misfortune, she argues, is moralistic and relates to a sin committed—someone died because they broke a taboo, and purification rituals are called for. Another explanation is that which attributes a misfortune to the work of individual adversaries. A third explanation blames the misfortune on an outside enemy, who must be punished.

In contemporary secularized societies, The concept of risk emerges as a key idea...because of its uses as a forensic resource' (Douglas 1992:24). Risk has largely replaced older ideas about the cause of misfortune. Concepts such as sin, which were once used to provide

explanation for misfortune, are now discredited. In their place is the 'modern, sanitized discourse of risk', which

is all we have for making a bridge between the known facts of existence and the construction of a moral community.... Indeed, risk provides secular terms for rewriting scripture: not the sins of the fathers, but the risks unleashed by the fathers are visited on the heads of their children, even to the *nth* generation.

(Douglas 1992:26)

The difference between danger in the context of taboo as it is used in pre-modern societies and risk as a central concept of danger in modern societies is that taboo is tied into a rhetoric of retribution and accusation against a specific individual. As such, it is a means of binding a community together by ensuring that norms boundaries are maintained. In contrast, the concept of risk, as part of a more individualistic community, is invoked to protect individuals against others: 'The dialogue about risk plays the role equivalent to taboo or sin, but the slope is tilted in the reverse direction, away from protecting the community and in favour of protecting the individual' (ibid.: 28). Being nominated as 'at risk', she argues, 'is not the equivalent but the reciprocal of being 'in sin' or 'under taboo'. To be "at risk" is equivalent to being sinned against, being vulnerable to the events caused by others, whereas being "in sin" means being the cause of harm' (ibid.: 28). Being 'at risk', in other words, entails being placed in the role of victim, threatened by risks imposed upon oneself by other agents, rather than being seen as bringing risk upon oneself through one's own actions.

Douglas (ibid.: 14–15) asks why it is that the word 'risk' has come to prominence at this point of western history, replacing concepts such as danger? Her answer is that it is not simply a matter of risk having claims to be 'scientific', although this is an important aspect of the

process. Rather, risk concerns have emerged as part of a complex of new ideas, incorporating a heightened sensitivity to issues of danger. Douglas sees the current concern with risk as a product of globalization, which has resulted in a new level of inter-community discourse and a sense of vulnerability in being part of a world system. As a consequence of globalization, the nation has to provide new kinds of protection. In this context:

The idea of risk could have been custom-made. Its universalizing terminology, its abstractness, power of condensation, its scientificity, connection with objective analysis, make it perfect. Above all, its forensic uses fit the tool to the task of building a culture that supports a modern industrial society.

(Douglas 1992:15)

According to Douglas, this 'new concern with risk' is part of 'a public backlash against the great corporations' (ibid.: 15). The political pressure that is brought to bear in relation to risk disputes is largely against exposing others to risk. This pressure is therefore centred less on individuals as being to blame and more on large organizations. In remarking on the ways in which social organizations such as the green movement respond to risk, Douglas points to the political and moralistic nature of risk discourses:

Though the public seems to be thinking politically in terms of comparative risks, the number-crunching does not matter; the idea of risk is transcribed simply as unacceptable danger. So 'risk' does not signify an all-round assessment of probable outcomes but becomes a stick for beating authority, often a slogan for mustering xenophobia.

(Douglas 1992:39)

Douglas and Wildavsky (1982) explain the influence of environmental activists in debates over the risks associated with ecological degradation and pollution in terms of the environmental movement having a secular sect-like nature. To achieve internal cohesiveness, they claim, the environmental movement positions an Other—in this case, particular industries and government departments—as the enemy that is demonized and blamed for risk.

Imputations of risk, therefore, like those of impurity, may be used as a means of social coercion and maintaining the moral and social order, a way of dealing with 'polluting people' who are culturally positioned as on the margins of society. This positioning occurs not just with organizations but also with social groups and individuals. Certain classes of people are singled out as likely victims of hazards, as being 'at risk' and therefore requiring control to bring them back to conforming to moral values. If the blame is removed from the victims and placed on the shoulders of their close kin, the tendency towards moral conformity is even more effective, as when parents are blamed for a child's handicap (Douglas 1985:57).

Here, despite Douglas' insistence, discussed above, that being 'at risk' is different from being 'in sin', it can be seen that people may sometimes be blamed for being 'at risk' just as they were once blamed for being 'in sin'. Blame may also be located externally, upon enemies, as a means of diverting attention away from oneself. Each type of strategy tends to be used in different contexts, albeit for the same purpose: to maintain social cohesion:

Clearly, blaming the victim is a strategy that works in one kind of context, and blaming the outside enemy, a strategy that works in another. Victim blaming facilitates social control; outsider blaming enhances loyalty. Both ploys would serve as an intention to prevent the community from being riven by dissention.

(Douglas 1985:59)

THE GRID-GROUP MODEL

Another important feature of Douglas' structuralist cultural theory is her attempt to distinguish between certain defined modes of organization and their related responses to risk. Douglas (ibid.: 61-3) defines two different ideal-types of social organization in terms of their dominant approach to risk. The first is that in which the members have a strong commitment to strengthening and maintaining internal bonds. To this end, they institute sharply defined separations within a hierarchy and close the community against the outside. This type of community prefers to allocate responsibility for disaster to victims and their kin, along with strategies for them to repent and expiate their sins. It is therefore both morally punitive and conciliatory. The second type of social organization has a strong commitment to individual enterprise and fair competition, and thus is founded on internal conflict rather than solidarity. This heroic, adversarial community has a more neutral approach to disasters, based on notions of chance and fate rather than culpability and punishment.

In other writings, this dichotomy is expanded to present a four-cell model of risk rationalities or world views. With Wildavsky, Douglas developed the 'grid-group' model of behaviour in understanding different logics of risk as they are expressed in social groups or organizations (Douglas and Wildavsky 1982; Douglas 1992). The group index contrasts two ideal types of groups: those who have a high group and those with a low group ethos. This index refers to 'the outside boundary that people have erected between themselves and the outside world' (Douglas and Wildavsky 1982:138). The high group ethos emphasizes cohesion amongst group members and makes strong distinctions between 'us' (members of the group) and 'them' (the world outside the group), while the low group ethos emphasizes individuality and weaker ties with others. The grid index relates to 'all the other social

distinctions and delegations of authority that [people] use to limit how people behave to one another' (ibid.: 138). These include the extent of social constraints on individual behaviour, imposed either by group membership or by other structural factors such as gender, ethnicity/race and social class. Individuals in groups classified as ideal type 'high grid' are subject to a large number of cultural constraints, while those who are 'low grid' have few constraints shaping their actions.

Putting these two indexes together, four ideal types and consonant four approaches to risk can be identified. They include: hierarchists (high group and high grid) who respect authority, conform closely to group norms and expectations relating to risk and trust established organizations; egalitarians (high group and low grid) who strongly identify with their group and blame outsiders for risk, tending to be distrustful of externally imposed norms and supportive of social equality issues and a participatory approach to risk; individualists (low group and low grid), who are individualistic and entrepreneurial, support self individuals rather regulation of risk. trust organizations, believe in market forces, see risk-taking as bringing benefits as well as dangers with it and resent external constraints; and fatalists (low group and high grid) who lack strong cohesion to a group but are otherwise highly constrained in their behaviours, and tend to trust to luck and fate in relation to risk, seeing themselves as having little personal control over it (Douglas and Wildavsky 1982).

This model may be subject to criticism for its apparent rigidness, static nature and inability to account for the ways in which most individuals constantly move between the four world-views rather than belonging to one or the other. It tends to treat the concept of 'risk' as a given in suggesting that it is the world-view rather than the nature of the risk itself that is the source of differential responses to risk. Yet it may be argued that world-views themselves shape what phenomena are singled out as 'risks' and how

serious they are perceived as being. They are therefore complicit in the production of risks and not simply responses to given risks. Nonetheless, if it is accepted that the model presents ideal types rather than assuming that people can be accurately identified as belonging and adhering to only one of the four world views, it provides a basis for examining the cultural locations within which risk is conceptualized and dealt with in a particular sociocultural setting.

Douglas (1992), for example, has herself used the gridgroup to explore the ways in which a particular city or town may include four types of groups, all of which have different responses to a threat such as HIV/AIDS. Building on her previous assertions about the relationship between notions of the human body and those of the body politic, she relates the group response to HIV/AIDS risk to specific conceptualizations of the individual body. She describes four abstract conceptualizations of the body in relation to the threat of illness and disease found among residents in Brittany, a region of France. The first conceptualizes the body as porous, completely open to the invasion of infection and therefore not able to protect itself against illness. The second is a vision of the body as very strong, with an immune system that is resistant to infection and thus requiring little in the way of protection apart from engagement in hygienic routines that allow it to function well. The third is a conceptualization of the body as having two protective layers, its own physical skin and the community, each of which acts to police entries and exits and thus the risk of contamination to the individual body. The fourth is a concept of the body as a machine that has its own protective envelope, which if pierced through an act of carelessness (for example, allowing penetration by an infected person's bodily fluids) renders the body susceptible to infection.

In each conceptualization of the body, HIV/AIDS risk is viewed differently. For the first conceptualization, risk is ever-present and difficult to guard against, but for this reason, those who become infected are not seen as being to blame. For the second, risk is a minor problem, because the body is regarded as inherently so resistant to infection. The third conceptualization sees risk prevention as directed at guarding the boundaries of the two protective layers, particularly monitoring the boundaries of the community. For the fourth, risk is engendered by lack of proper care of the body and therefore those who succumb to infection are viewed as lacking appropriate self regulation. Douglas notes that:

The most baffling thing about the pattern is that a large number of the community at risk are impervious to information; either they know unshakeably that they themselves are immune, or recognizing that death is normal they draw the conclusion that to live trying to avoid it is abhorrent.

(Douglas 1992:111)

Education about the transmission routes of HIV infection often founders in the face of these cultural models of the body.

Douglas then explores four typologies of city groups that reflect the conceptualizations of the body described above. The first is the city core or the central community, a hierarchical structure which dominates the city and defends its centrality against outsiders. In the face of an epidemic such as HIV/AIDS, the city core tightens its defences, adopting hygienic measures proposed by experts. It adopts the model of the community as having two skins, a double protective envelope of bodily defence and community defence against invasion. The core group therefore ranges all its forces against the external threat, which is seen to emerge from among those who deviate from the moral order, such as gay men, prostitutes, injecting drug users. For those in the core group, therefore:

The simple and quite correct idea that a new infection has external origins is transmuted in the course of the cultural project into a complex weapon of control. If townsfolk could really believe it is a problem caused by transients: good, then it does not concern them, so long as they can cordon off the town. The middle class believe it is a problem of the 'inner city', urban poverty: good, it does not concern them, so long as there is a cordon sanitaire. Stable citizens believe it is a problem of itinerancy, beggars, strangers, and travellers. So long as the class at risk can be kept in the margins, the public concern to pay for the research and welfare of the victims will be the weaker.

(Douglas 1992:117)

The second group is that of the dissenting minorities, who are not organized hierarchically but through solidarity in their protest against the dominant city core group, who they position as oppressive of their agency. This group rejects the expert knowledge of the core group, often engaging in self-help or alternative strategies. In the case of HIV/AIDS, such groups may include gay community groups and AIDS activist groups. They also adopt the model of the community as having a double protective envelope, seeing the body as needing to protect itself from infection and the community as having to guard itself from outsiders who may contaminate it. This minority community, however, unlike the core community, is inherently defined as unhealthy and diseased, for it is characterized by having many members who are ill with HIV/AIDS (hence their marginalization from the core). Members of the dissenting enclave may therefore sometimes have a fatalistic approach to HIV risk.

The third group is individualist, comprising of individuals adopting the culture of entrepreneurialism who operate as autonomous units and resist being defined as part of a community. Those in this group want to be left

free to pursue their own interests. They tend to be idiosyncratic in their response to risk advice. They see themselves as being in control of the routes of infection to the extent that they want to be, and often scorn the importance of engaging in precautionary strategies such as safer sex activities. The fourth group is the isolate or fatalist group, a residual category expelled to the margins by the centre community. They rather unwillingly find themselves restricted by the structures imposed by others. They tend not to respond as active agents to HIV/AIDS risk, but instead accept whatever fate might bring them. Unfortunately Douglas is not explicit about which of the models of the body these latter two groups might favour in terms of HIV/AIDS risk, but it is clearly not that of the double envelope of protection, as these groups would not think in terms of community membership. Rather, risktaking would be conceptualized at the level of one's own deportment of the body and interaction with others' bodies. It seems likely that these groups may adopt any of the other three models of the body described by Douglas, as each of these models is individualistic, relying on individuals to take their own actions to protect themselves from risk.

This analysis has something to offer, particularly in bringing together models of the human body in relation to HIV/AIDS with those of the body politic. The way in which dominant social groups have reacted against those they have positioned as 'deviant' and 'risky' by attempting to marginalize and exclude them, drawing up a *cordon sanitaire* of hygienic strategies to demarcate boundaries, is a clear trend in the history of the HIV/AIDS epidemic (and many other infectious diseases) in most societies. (See Chapter 7 for further elaboration on strategies that define self against Other in response to risk.) Again, however, this model may be criticized for being rather too static and general in its categorizations, not acknowledging the heterogeneity of social groups. Gay men, for example, even those who are members of AIDS activist groups, do

not all necessarily reject the expert advice issuing forth from the core community and turn towards self-help remedies. Indeed, many activist groups have called for more rather than less advice and therapeutic treatment from mainstream medicine.

CONCLUDING COMMENTS

Mary Douglas' theoretical approach to risk has been influential because she has provided a trenchant and persuasive critique of the realist approaches that have dominated the field. Her 'cultural/symbolic' approach emphasizes that risk judgements are political, moral and aesthetic, constructed through cultural frameworks of understanding and implicated with notions of the body and the importance of establishing and maintaining conceptual boundaries. This provides a perspective on risk that sets up an important counterpoint to the individualist focus that pre-dominates in the realist perspective. For those interested in questions of risk that go beyond the individualistic to a fundamentally shared, cultural and symbolic approach to risk, Douglas' writings provide a firm basis. Her approach does tend to be somewhat static, however, as is typical of functional structuralist analyses of sociocultural phenomena. There is little explanation provided for how things might change in Douglas' accounts of risk, purity and danger.

There is also somewhat of a problem in the way that Douglas' theorizing on risk has been used by others. As pointed out in Chapter 1, a number of psychometric and other psychological researchers have begun to incorporate some of Douglas' insights into their models of risk perception, often referring to her work as the 'cultural theory' approach to risk. In this literature, however, Douglas' writings are frequently interpreted as implying that lay perceptions of risk involve inaccuracies and errors of judgement because of the 'contaminating' influence of cultural and social processes. In an article on people's

perceptions of the risk of crime, for example, Kemshall uses Douglas and other cultural theorists to support her claim that 'risk perception can be prone to bias and error leading to exaggeration and overestimation of risks. Risk perception is a subjective process, with what we identify and respond to as a risk often a matter of value judgement rather than fact' (1997:247–8). In such writings, as in much of the other psychometric and psychological literature, the risk judgement of 'experts' continue to be privileged as 'objective' and 'factual' over those of lay people, against which perceptions are compared and found wanting.

Other commentators who take a laissez-faire political approach have used Douglas' work to claim that the state should not attempt to impose its views on risk on the public in risk reduction efforts, given the 'subjective' nature of risk assessment, the uncertainty of knowledge about risk and the political dimensions of risk debates (see, for example, Adams 1995). Douglas' own position on risk sometimes seems to support a politically conservative approach, particularly in her writings environmentalist movement. Douglas and her collaborator Wildavsky have been criticized for siding with business and industry inappropriately by positioning them as the victims of the environmental movement. One reviewer of Risk and Culture, for example, has seen their ideas as seeming 'to proceed more from an animus against environmentalism and middle-class "liberalism" than from Douglas's work on pollution' (Kaprow 1985:345). In Douglas and Wildavsky's work, environmentalists tend to be portrayed as behaving politically and ideologically, in constructing certain beliefs about industrial pollution and risk. In contrast, the risk positions of industry and big business tend to be represented as politically neutral. In this respect, therefore, and in representing industry and big business as singled out as scapegoats and inappropriately blamed for risk, Douglas may herself be criticized for failing to recognize the cultural underpinnings of these

institutions' risk positions. This seems to clash with her overall position on risk, which while taking a 'weak' rather than 'strong' social constructionist approach, is adamant about the inherently cultural nature of any group or community's perceptions and judgements about risk.

4 RISK AND REFLEXIVE MODERNIZATION

As I observed in Chapter 2, the exponents of the 'risk society' perspective are primarily interested in the ways in which the concept of risk is related to the conditions of late modernity. This perspective offers an approach that considers the politics and macro-level of the current meanings and strategies of risk. 'Risk society' exponents focus on such processes as individualization, reflexivity and globalization as converging in the 'risk society' of western nations. This chapter examines the insights offered by the two major exponents of the 'risk society' thesis: Ulrich Beck and Anthony Giddens. Although they initially developed their diagnoses of risk and late modernity largely separately of each other, the writings of Beck and Giddens have much in common: hence the decision to consider them together in this chapter. As I note at the end of the chapter, however, there are also some in their writings which require major differences acknowledgment.

BECK AND THE 'RISK SOCIETY'

The German sociologist Ulrich Beck has become a prominent figure in the sociological literature on risk. His book *Risk Society: Towards a New Modernity?* was first published in English translation in 1992 (the original German version appeared six years pre-viously) and sparked much debate about the nature of risk in

contemporary western societies. Beck has followed up this book with others on aspects on risk, several of which are available in English. These works include *Reflexive Modernization* (1994, written with Anthony Giddens and Scott Lash), *The Normal Chaos of Love* (1995, written with Elisabeth Beck-Gernsheim) and *Ecological Politics in the Age of Risk* (1995), as well as a number of journal articles and book chapters on the subject (see, for example, Beck 1992a, 1996a, 1996b).

In these publications, Beck argues that individuals in contemporary western societies are living in a transitional period, in which industrial society is becoming 'risk society'. In this transitional period, the production of wealth is accompanied by that of risks, which have proliferated as an outcome of modernization. The central problem of western societies, therefore, is not the production and distribution of 'goods' such as wealth and employment in conditions of scarcity (as it was in early modernity and remains the case in developing countries) but the prevention or minimization of 'bads'; that is, risks. Debates and conflicts over risks have begun to dominate public, political and private arenas. Individuals living in these societies have therefore moved towards a greater awareness of risk and are forced to deal with risks on an everyday basis: 'Everyone is caught up in defensive battles of various types, anticipating the hostile substances in one's manner of living and eating' (Beck 1994:45).

In much of Beck's writings he demonstrates a realist approach to risk. A risk, for him, is another word for a hazard or danger, and he claims that the 'risks of modernization' are 'irreversible threats to the life of plants, animals, and human beings' (Beck 1992b:13). At many points he demonstrates anger at the ever-hazardous nature of life in late modernity, presenting an apocalyptic vision of how hazards and dangers may destroy humankind and other living creatures. This realist approach, however, is not consistently maintained throughout his work. In some parts of Beck's writings, the

social and cultural processes by which understandings and perceptions of risk are mediated are highlighted, and he 'weak' version demonstrates а constructionism. In Risk Society, for example, at one point he notes that there is a difference between 'a risk itself and 'public perception of it', commenting further that: 'It is not clear whether it is the risks that have intensified, or our view of them' (1992b:55, original emphasis). Risks are 'risks in knowledge' and therefore perceptions of risk and risks themselves 'are not different things, but one and the same' (1992b:55). In a later article, Beck similarly argues that risks are 'social constructs which are strategically defined, covered up or dramatized in the public sphere with the help of scientific material supplied for the purpose' (1996a:4).

In Ecological Politics, Beck (1995) contrasts what he sees as the two major approaches to interpreting risk: 'natural-scientific objectivism about hazards' (Beck's term for the realist approach) and 'cultural relativism about hazards' (his term for the 'strong' social constructionist approach). He sees both approaches as having their strengths and weaknesses. 'Natural-scientific objectivism' is useful, he argues, because it identifies risks using technical powers of observation, measurement and calculation: 'hazards require natural-scientific categories and measuring instruments in order to be "perceivable" at all' (1995:162). The weakness of objectivism, however, is that in its quest for neutral objectivity it fails to recognize the ways in which 'scientific facts', like other views on risk, are situated and interpreted in cultural and political contexts: 'Neither experiments nor mathematical models can "prove" what human beings are to accept, nor can risk calculations any-way be formulated solely in technological-bureaucratic terms, for they presuppose the cultural acceptance they are supposed to manufacture' (1995:91).

The 'cultural relativism' approach, in Beck's view, rightly emphasizes the contextual aspect of risk responses,

pointing out that what concerns one social group in one historical era may not worry another. Risk calculations, from this perspective, 'are now no longer thought of as arbitrators but as protagonists in the confrontation, which is enacted in terms of percentages, experimental results, projections, etc' (Beck 1995:92). Yet he sees this approach as often becoming too relativist, regarding anything as potentially classifiable as dangerous, according to the viewpoint that is taken, and failing to recognize the special nature of 'real' contemporary hazards. Beck seeks to integrate both approaches into what he calls 'a sociological perspective' (1995:76). He maintains a 'natural-scientific objectivist' approach by subscribing to the idea that 'real' risks exist, but brings in 'cultural relativism' (in other words, a 'weak' social constructionist position) by arguing that the nature and causes of risks are conceptualized and dealt with differently in contemporary western societies compared with previous eras.

Beck is interested in the 'cultural disposition' demonstrated by individuals and social groups to single out certain risks as important while others are ignored. He asks, for example, why it is that the problem of the 'dying forests' is given such attention (particularly in his country, Germany) when other hazards, such as mass death on the roads, are largely ignored? The reason is that destruction and protest are

symbolically mediated. Symbols that touch a cultural nerve and cause alarm, shattering and making comprehensible the unreality and hyperreality of hazards in everyday life, gain a key significance precisely in the abstractness, imperceptibility and impalpability of the process of devastation kept alive by the advanced industrialism of hazards.

(Beck 1995:47)

Thus, images of dying seals or forests come to stand for the vastness of the risks that surround us, rendering comprehensible the incomprehensible and giving us a target on which to fix an illusory sense of control. A usual response to grave dangers is to deny their existence as a kind of psychological self-protective mechanism, an attempt to maintain a sense of normality. Symbolic targets of contamination help us do this by focusing our attentions towards them and ignoring other risks (Beck 1995:48–9).

Beck acknowledges that all societies in all epochs in human history have been subject to threats to health and life, and therefore could be described as 'risk societies'. However, he reserves the term 'risk society' to describe exclusively the trends of the contemporary era. Beck singles out for examination the risks in three eras: premodern societies, early modern and contemporary, or late modern societies (also referred to in his work as preindustrialism, early industrialism and late industrialism). He identifies several distinct features of risk in late compared with early modernity. A major difference is that dangers and hazards in contemporary societies principally environmental problems such as air and water pollution, ionizing radiation and toxic chemicals in foodstuffs—differ significantly from previous eras. Since the middle of the twentieth century, Beck claims, industrial society has been confronted with threats to human life which are on an unprecedented scale. Such threats cannot be delimited spatially, temporally or socially, unlike the 'personal' risks produced by early industrialization. The magnitude and global nature of risks is such that risks are becoming more and more difficult to quantify, prevent and avoid. Contemporary hazards now are often open-ended events, rather than events that have a foreseeable end. Beck talks about 'higher and higher levels of hazard' becoming 'the norm' (Beck 1995:13). In the contemporary era, he argues, hazards are far more apocalyptic than in previous eras, threatening the destruction of all life on earth.

Beck also contrasts the calculability of contemporary hazards with those of previous eras. He compares ways of thinking about and dealing with threats and hazards in pre-modern societies with those of early modernity. In premodern society, common threats (plague, famine, natural catastrophes, wars, but also magic, gods, demons) were deemed incalculable because they were attributed to external, supernatural causes. Via the processes of modernity taking place in early industrialism, however, these threats were transformed into calculable risks in the course of the development of instrumental rational control (Beck 1995:30). 'Risks' as they were understood in early modernity, were 'determinable, calculable uncertainties', seen as 'products of social choice, which must be weighed against opportunities and acknowledged, dealt with, or simply foisted on individuals' (1995:77).

According to Beck, in the late modern era the foundations of risk logic as it was developed in early modernity are being subverted or suspended. The processes of modernist risk calculation fail in 'risk society'. The risks of late modern society are not easily calculable because of their non-localized nature and potential long-term effects: 'to express it by reference to a single example: the injured of Chernobyl are today, years after the catastrophe, not even born yet' (Beck 1996a:31). In the event of the worst possible disaster or accident (such as from nuclear, biological or chemical weapons of mass destruction), the effects of which are long-lasting, irreparable and incalculable, there is no institution which could prevent it or compensate for its effects. In such a scenario, it is difficult to identify a single cause upon which blame can be cast, or to award financial compensation for the damage done, given the magnitude of the threat (1996a:15). The established early modernist rules of attribution and causality, therefore, break down in the face of the globalization of risk, as do the safety systems that once dealt with risk, such as insurance and compensation arrangements. Contemporary hazards can

only be minimized by technological means—they can never be removed entirely (Beck 1995:76–7).

In 'risk society' the assessment of risk is subject to a high degree of ambivalence, due to the complexity of society and technical knowledge. In the early days of industrialization, risks and hazards were evident to the senses—they could be smelt, touched, tasted or observed with the naked eye. In contrast, many of the major risks today largely escape perception, for they are 'localized in the sphere of physical and chemical formulas (e.g. toxins in foodstuffs or the nuclear threat)' (Beck 1992b:21). These risks exist in scientific knowledge rather than in everyday experience. Expert knowledges tend contradict each other, resulting in debates standpoints, calculation procedures and results. This also tends to paralyse action. Science itself fails in response to the large-scale, indeterminate nature of contemporary hazards. Hypotheses about their safety cannot be tested empirically and science has little power to intervene in a context in which the world has become a laboratory for testing how hazards affect populations. Scientists have therefore lost their authority in relation to risk assessments: scientific calculations are challenged more and more by political groups and activists (1995:125-6).

The nature of such hazards, therefore, return the concept of risk to the pre-modern notion of 'incalculable insecurities'. In common with such hazards, they 'undercut the social logic of risk calculation and provision' (Beck 1995:77). Ways of viewing and dealing with contemporary hazards are not the same as pre-modern viewpoints, however. The notion of 'nature', the demons or acts of God that frightened individuals in pre-modern societies, is a supernatural concept. 'Nature', and its dangers, were then seen as imposed by external forces and thus beyond the control of humans. By contrast, dangers and hazards in early and late modern societies are seen as humanly generated rather than supernatural.

They are thus considered the responsibility of humans to control and in principle can be avoided or altered.

This raises another important distinction between 'older dangers' and contemporary hazards—the linking of responsibility with the latter. contemporary hazards are seen to be the outcome of action—principally the related events modernization, industrialization, urbanization globalization—the drawbacks of such events are continually confronted and challenged. While uncertainties may always have been part of life, the difference in late modernity is that many of them arise from (rather than being assuaged by) the very growth in human knowledge. Indeed, even older risks such as plagues, floods and famines, are now rarely viewed as acts of gods or demons or the result of 'nature gone wrong'. Instead, human intervention is seen to have played a major role, resulting in nature 'striking back' for having been so cruelly treated or inappropriately managed.

Thus, for example, epidemics of bacterial infections are blamed on medicines resulting in antibiotic resistant bacteria, the cause of floods, landslides and famines are traced back to over-clearing of land and weather changes due to global warming, which in turn are seen to be the industrialization. side-effects of Risks. in contemporary meaning, are fundamentally based on decisions, principally made by organizations and political groups, that consider techno-economic advantages and considerations of utility. As a result, 'it is not the number of dead and wounded, but rather a social feature, their industrial self-generation, which makes the hazards of mega-technology a political issue' (Beck 1992a:98).

REFLEXIVE MODERNIZATION

The political aspects of risk and the self-critique that is inspired by risk are pivotal to Beck's concept of risk society. He defines risk as 'a systematic way of dealing

with hazards and insecurities induced and introduced by modernization itself. Risks, as opposed to older dangers, are consequences which relate to the threatening force of modernization and to its globalization of doubt. They are politically reflexive' (1992b:21). The concept of risk is linked to reflexivity because anxieties about risks serve to pose questions about current practices. In risk society, society becomes reflexive in three ways, stemming from the newly global nature of risk. First, society becomes an issue and a problem for itself at the global level. Second, the awareness of the global nature of risk triggers new impulses towards the development of co-operative international institutions. Third, the boundaries of the political come to be removed, leading to world-wide alliances. By these processes, risk society becomes 'world risk society', in which the public sphere of political debate and action is globalized. It produces a different kind of citizenship, 'global citizenship', in which traditional means of defining identity, based on local contexts, are exchanged for a focus on the world-wide perspective. This results in the generation of new alliances, of ad hoc activist groups, a new and different form of politics beyond traditional hierarchies (Beck 1996a:2).

Beck (1994, 1996b) contends that the use of the term 'reflexive' in his concept of reflexive modernization does not denote mere 'reflection' but rather 'self-confrontation'. The move towards reflexivity is an unintended side-effect of modernity, or rather the hazards produced by modernity as part of its project (1996b:28). It is the process of modernity coming to examine and critique itself. It is thus not a repudiation of modernity per se but rather an application of the principles of modernity to itself. Reflexive modernization contains two phases. The first (the reflex stage) is part of the automatic transition from industrial to risk society, where risks are produced as part of the processes of modernization but are not yet the subject of sustained public or personal debate or political conflict. The second (the reflection stage) involves

industrial society coming to see itself as a risk society, with the growing realization of the dangers involved in modernity which then calls into question the structures of society. Reflexive modernization, therefore, is:

the combination of reflex and reflections which, as long as the catastrophe itself fails to materialise, can set industrial modernity on the path to self-criticism and self-transformation. Reflexive modernisation contains both elements: the reflex-like threat to industrial society's own foundations through a successful further modernisation which is blind to dangers, *and* the growth of awareness, the reflection on this situation.

(Beck 1996b:34, original emphasis)

This critical reflection upon the dangers of modernity is the difference between industrial society and risk society.

For Beck, reflexive modernization leads to 'possibility of a creative (self)-destruction for an entire epoch: that of industrial society' (1994:2). Progress has turned into self-destruction, but not through class struggle or revolution, as Marx predicted, but as an unintended consequence, through the inexorable and incremental processes of modernization itself. Judgements on risk represent implicit moral judgements (albeit masked in the discourse of objective, quantitative 'facts'), on the ways in which human societies have developed (Beck 1992b:176). The naive certainties of the Enlightenment—the optimism human progress wrought through science and rationalized action—have disintegrated, resulting in individuals' need to seek and invent new certainties for themselves (Beck 1994:14). Lay people have become sceptical about science, because they are aware that science has produced many of the risks about which they are concerned and that scientific knowledge about risk is incomplete and often contradictory, failing to solve the problems it has created. People must deal, therefore, with

constant insecurity and uncertainty: conventional social order seems to be breaking down in the face of the undermining of old certainties. There is a continual definitional struggle over risk, particularly between those who produce risk definitions (principally experts) and those who consume them (the lay public). In risk society, as a result, risk has become a highly political concept.

Beck (1992b:59) is critical of experts' positioning of lay people as ignorant, merely requiring more information about risk to respond appropriately. He argues that lay people's apparent 'irrationality' in relation to risk is a highly rational response to the failure of technico-scientific rationality in the face of the growing risks of late modernity. The understandable response of individuals is to become critical of these dangers generated by early modernity and its drive towards industrial production. Beck claims that this is already happening in diverse forms in organizations such as the green movement but also among members of the general public.

Beck somewhat confusingly sees risk as simultaneously reinforcing positions of inequity and as democratizing, creating a global citizenship. He acknowledges that some social groups are more affected than others by the distribution and growth of risks, and these differences may be structured through inequalities of class and social position. The disadvantaged have fewer opportunities to avoid risks because of their lack of resources compared with the advantaged: 'Poverty attracts an unfortunate abundance of risks. By contrast, the wealth (in income, power or education), can purchase safety and freedom from risk' (Beck 1992b:35). Beck also points to growing inequalities in risk distribution, where class positions and risk positions overlap. He acknowledges that what he calls 'the former colonies' of the western world are on their way to becoming the waste-dumps of the world for toxic and nuclear wastes produced by more privileged countries.

However, Beck also argues that many of the risks of risk society affect the wealthy and the poor in similar ways: 'poverty is hierarchic, smog is democratic' (1992b:36). Such risks as radiation, nuclear warfare, toxic chemicals in food and air and water pollution, because they are so widespread and 'invisible' in their mani festations or effects, or both, cannot be avoided even by those who are socioeconomically advantaged. The risks of modernization also affect those who have produced or profited from them, breaking down boundaries between classes and countries. One feature of risk society is that those individuals who are most concerned and aware about risks are from the highly educated, well-off classes, because they make great efforts to inform themselves about risks. Despite their best attempts, however, these people's knowledge about the extent of risk is incomplete, for they do not have access to scientific knowledge (that owned by 'external knowledge producers') (Beck 1992b:53). Risk society is thus characterized by the contradiction that the privileged have greater access to knowledge, but not enough, so that they become anxious without being able to reconcile or act upon their anxiety.

INDIVIDUALIZATION

The concept of individualization is also central to Beck's view of risk society and reflexive modernization. Individualization refers not to alienation or loneliness. Instead it means the requirement in late modernity that individuals must produce their own biographies themselves, in the absence of fixed, obligatory and traditional norms and certainties and the emergence of new ways of life that are continually subject to change (Beck 1994:13). Individualization is the other, private side of globalization in reflexive modernization. The core of reflexive modernity is a transformation in and freeing of accepted social roles such as gender and social class (Beck 1992b:87ff.). Beck sees individualization as the outcome

of modernization processes involving a reduction in the influence of the traditional structuring institutions of society in the formation of personal identity. Such factors as mass education, improvements in living standards, the second-wave feminist movement and changes in the labour market have contributed to the trend towards individualization.

Individualization means 'the disintegration of the certainties of industrial society as well as the compulsion to find and invent new certainties for oneself and others without them' (Beck 1994: 14). It is dependent on decision-making as it assumes agency, the ability to shape destiny through self-determination identification. There has been 'a social surge of individualisation' (Beck 1992b:87) in which people have become compelled to make themselves the centre of the conduct of life, taking on multiple and mutable subjectivities. Individualization is a social transformation that is complex and ambiguous: 'Seen from one angle it means freedom to choose, and from another pressure to conform to internalized demands, on the one hand being responsible for yourself and on the other being dependent on conditions which completely elude your grasp' (Beck and Beck-Gernsheim 1995:7). While, as a long-term outcome of social processes, individualization does not affect all social groups similarly, it has had an increasing effect on the conduct of everyday life for large numbers of people.

Where once, in pre-modern society, it was expected that one's destiny was pre-structured through the chance of one's station in life at birth, the life-course is now conceptualized as far more flexible and open, albeit through individual's endeavours rather than the vagaries of fortune. Beck (1992b:135) refers to this as 'reflexive biography', or biography that is self- rather than socially-produced. Traditional forms of coping with anxiety and insecurity—families, marriage and male-female roles—are failing, and individuals must turn to themselves, creating

new demands for social institutions such as education, therapy and politics (1992b:153). counselling. Individualization, therefore, involves a proliferation of new demands upon people at the same time as choices have become more and more complex and difficult, particularly in relation to such matters as sexual identity, work and family relationships. Such planning requires a high and continuing exertion of reflexivity upon the nature and future of one's life-course.

In such arenas as education and work, for example, people are expected to make their destinies, to compete with others for credentials and employment and make their individualized careers, no longer relying on traditional expectations or social structures. Stable employment can no longer be taken as a given, and it is considered up to people to make their own opportunities. Workers must deal with a pluralized, decentralized labour market which makes demands of them to be flexible and entrepreneurial, or else suffer the consequences of underemployment or unemployment. In the arena of personal relationships, individualization results in greater conflict between individuals in intimate partnerships, as each attempts to pursue her or his right for autonomy and selfimprovement as well as maintain a relationship. Gender roles no longer rigidly structure the life course. As a result, there are greater possibilities for choice in the conduct of relationships, but this has required intense and continual negotiation and decision-making for couples. People must juggle their need for autonomy and self-expression with their need for dependence and emotional stability in relationships. Contradictions have opened up between the demands of the family and those of the workplace, which assumes an autonomous individual who is unburdened with family responsibilities. In this context, new personal risks have been generated, particularly for women, involving insecurities and around employment and economic issues as well as around the stability of

relationships in marriage (Beck and Beck-Gernsheim 1995).

Individualization is therefore fraught with risk (ibid.). With the breakdown of many of the traditional certainties structured through age, gender and social class, a plurality of new risks are generated, including unemployment or underemployment, marital instability and breakdown, accompanied by high levels of anxiety and insecurity. Life becomes less certain even while it is placed more under one's control. This move towards individualization does not mean that social inequalities or structuring of opportunities through such attributes as class, gender or ethnicity have disappeared. Rather, in the face of individualization the influence of these structures have become less obvious and acknowledged as affecting life chances. Inequalities have become primarily viewed as individualized, perceived as 'psychological dispositions: as personal inadequacies, guilt feelings, anxieties, conflicts, and neuroses' (Beck 1992b:100). To choose the wrong kind of university degree, occupation or marriage partner, to face unemployment or marital breakdown, tends to be considered the result of an individual's faulty planning or decision-making rather than the outcome of broader social processes.

GIDDENS' PERSPECTIVES ON RISK

Anthony Giddens, in his books *The Consequences of Modernity* (1990) and *Modernity and Self-Identity* (1991), and with Lash and Beck in *Reflexive Modernization* (1994), has also written at length on risk and the uncertainty with which individuals approach life in contemporary western societies. Like Beck, Giddens sees this uncertainty as springing from the realization that the claims of modernity for human progress have been shown to be not as utopian as once was thought. Giddens' writings on risk build on his previous arguments on modernization and globalization and their relationship to

the nature of everyday life. He argues that: 'One of the distinctive features of modernity, in fact, is an increasing the two "extremes" interconnection between extensionality and intentionality: globalising influences on the one hand and personal dispositions on the other' (1991:1). He is particularly interested in identifying and analysing these interconnections and reflecting on their implications for social responses to risk.

Giddens agrees with Beck in seeing late industrialism or late modernity as being characterized by transformations in traditional habits and customs, having a radical effect on the conduct and meaning of everyday life. He identifies modern institutions as being central to the nature of modernity. These institutions affect everyday life and selfhood, but in turn are themselves shaped by individuals' actions. For Giddens, the key features of modernity are institutional and individual reflexivity combined with the reorganization of time and space and the expansion of disembedding mechanisms, or mechanisms which take social relations out of their specific time/space contexts apply them in wider locales. One of these disembedding mechanisms is systems of expert knowledge, because they deploy 'modes of technical knowledge which have validity independent of the practitioners and clients who make use of them' (ibid.: 18). Other disembedding mechanisms include global standardized time zones and symbolic tokens, or objects which are media of exchange that have standard value, such as money as it is used in the global economy.

According to Giddens, in pre-modern times space and place largely coincided, dominated by localized activities. The conditions of modernity increasingly removed space from place by fostering relationships between 'absent' others who were distant from each other. Unlike in the pre-modern era, when experiences and traditions were very much confined to particular localities and were therefore highly contextual and fragmented, modernity brings together human experiences and

knowledges. Modern institutions are unifying, establishing a 'single world', a sense of a 'we' as humanity facing problems and opportunities together which never existed in pre-modernity (ibid.: 27).

Giddens refers to the 'double-edged character of modernity' (1990:10), or the risks associated with the drive for progress. Because of disembedding mechanisms and globalization, events now may have potentially disastrous effects that are far more wide-reaching than previously. The fall of the global capitalist economy affects the life chances of millions, a malfunction in a nuclear power plant equally may kill or harm millions. Giddens describes late modernity, therefore, as 'a risk culture' (1991:3). He emphasizes that he does not mean by this term that people in contemporary western societies are more exposed to danger or are more anxious about threats to their welfare than they were earlier. In previous eras, there were fears of terrible catastrophes affecting the world and destroying humankind. In the present day, however, such fears are commonly linked to the perception that humans (rather than the gods or fate) have brought these catastrophes upon themselves (ibid.: 121-2). The threats are therefore of a different kind, part of the 'dark side of modernity' (ibid.: 122). Hazards and dangers are now conceptualized as 'risks' rather than as givens, things over which humans are seen as potentially exerting control: 'it is a society increasingly preoccupied with the future (and also with safety) which generates the notion of risk' (Giddens 1998:27).

Giddens therefore displays a similar position to Beck on the nature of risk, in relating it to hazards or dangers that exist objectively, differ substantively from previous eras and are now linked to human responsibility. He describes the notion of risk as moving through two stages which roughly correspond to Beck's observations about changes in conceptualizing and dealing with risk in early modern and late modern eras. In the first stage, risk is seen as 'an essential calculus', a way of promoting certainty and order

in the precision of risk calculations, of 'bringing the future under control', where the various components of risk are 'given' and thus able to be calculated (Giddens 1994:58-9). It is upon this notion of risk that the welfare state developed as a means of protecting populations from the risks of actuarially calculable threats such as illness and unemployment with the use of social insurance schemes (Giddens 1998:27-8). The second stage is where we are not able to precisely calculate risk, but rather develop 'scenarios' of risk with various degrees of plausibility. One example is global warming, subject to expert dispute over whether or not it is happening and how serious its ramifications are. As an outcome of modernity, there are now far greater uncertainties than ever previously existed.

Like Beck, Giddens describes modern reflexivity as being different from the reflexive monitoring which has always been part of human activity. Modern reflexivity, for both individuals and institutions, involves awareness of the contingent nature of expert knowledges and social activity, their susceptibility to revision and change (Giddens 1991:20). This reflexivity is both the product of the Enlightenment and confounds its expectations of inexorable human progress through knowledge. The conditions of modernity—the progressive separation of space, place and time and the increasing role played by disembedding mechanisms—all depend upon trust, vested not in individuals but in 'abstract capacities' (Giddens 1990:26). People now cannot simply rely on local knowledges, tradition, religious precepts, habit observation of others' practices to conduct their everyday lives, as they did in pre-modern and early modern times. Rather, they must look principally to experts they do not personally know and are unlikely ever to meet to supply them with guidelines.

Modernity is also characterized by doubt about the validity of knowledges, acknowledging that all knowledge is open to revision. Greater knowledge had led in turn to greater uncertainty: 'The fact that experts frequently

disagree becomes familiar terrain for almost everyone' (Giddens 1994:186). The reflexive organization of knowledge environments requires the constant prediction of the nature of outcomes in the future, or risk assessment. This assessment, by its very nature, is always imprecise, for these calculations rely upon abstract knowledge systems which are subject to contestation and change. As a result, people have become increasingly cynical about the claims to progress offered by traditional modernity. Hence the continual reflexivity of individuals and institutions. In pre-modern eras, reflexivity was largely structured by the traditions established within the time-space organization of individual communities. In modernity, reflexivity took on a different character—tradition became less important as a justification for action. Instead, 'social practices are constantly examined and reformed in the light of incoming information about those very practices, thus constitutively altering their character' (Giddens 1990:38).

Giddens argues that the concept of risk is antithetical to that of fate, because fate assumes a predestined path:

A world mainly structured by humanly created risks has little place for divine influences, or indeed for the magical propitiation of cosmic forces or spirits. It is central to modernity that risks can in principle be assessed in terms of generalisable knowledge about potential dangers—an outlook in which notions of *fortuna* mostly survive as marginal forms of superstition.

(Giddens 1990:111)

While notions of fate and destiny may still exist in late modernity, that of risk is also dominant. Risk assumes that there are no aspects of human action that follow a preordained course, but rather they are open to contingencies: 'Living in a "risk society" means living with a calculative attitude to the open possibilities of action, positive and negative, with which, as individuals and

globally, we are confronted in a continuous way in our contemporary social existence' (Giddens 1991:28). Risk assessment involves weighing up and choosing among various possible courses of action according to their predicted outcomes. Modernity, therefore, is oriented towards the future, while pre-modernity was oriented to the past (ibid.: 29).

Because the self is seen as a reflexive project in late modernity, as a problematic rather than a given, there is far more emphasis on the malleability of the self and the responsibility that one takes for one's life trajectory. Individuals have greater recourse to expert knowledges in constructing the project of the self (ibid.: 32-3). As knowledge is being constantly revised in late modernity, the processes of reflexivity are more complicated and uncertain. There are more choices to be made: 'the self, like the broader institutional contexts in which it exists, has to be reflexively made. Yet this task has to be accomplished amid a puzzling diversity of options and possibilities' (ibid.: 3). Because traditions have lost their power and there is more 'openness' about how one can live, the concept of lifestyle has become ever more important to selfhood, forcing people to negotiate among a range of options. The reflexive project of self-identity requires 'consideration of risks as filtered through contact with expert knowledge' (ibid.: 5). So too, Giddens argues, the body is seen less and less as a 'given' but rather as subject to individual manipulation and will: it is, therefore, 'reflexively mobilized' (ibid.: 7). Developments in the fields of science and medicine contribute to the notion that the body is amenable to human action, such reproductive technologies, genetic engineering and surgical procedures. Here Giddens' writings again closely parallel those of Beck, this time in relation to the processes that Beck calls 'individualization'.

TRUST

One difference between Beck and Giddens is that in the latter's writings on risk and late modernity, the notion of trust is dominant. Giddens asserts that with the late modern reliance on generalized expert systems over local knowledges, and upon symbolic tokens such as money, trust remains a necessary part of life: The disembedded characteristics of abstract systems mean constant interaction with "absent others"—people one never sees or meets but whose actions directly affect features of one's own life' (Giddens 1994:89). However, if expert knowledges fail, the repercussions extend far beyond the local context. Reliance upon global expert systems. therefore, is characterized by uncertainty. People are required to be more challenging of expert knowledges, requiring of them that they win their trust. They are also turning back towards face-to-face relationships in the attempt to 'reembed' their trust in those whom they know personally. This involves different sorts of trust relations, based on intimate others, and different sorts of risks. In the case of both experts and intimate others trust must be won and continually negotiated.

Trust presupposes awareness of risk, offering reliability in the face of contingent outcomes and thereby serving to minimize concern about possible risk: 'What is seen as "acceptable" risk—the minimising of danger—varies in different contexts, but is usually central in sustaining trust' (Giddens 1990:35). Trust is vital to establish ontological security from infancy onwards. Giddens defines ontological security as an emotional phenomenon, incorporating 'the confidence that most human beings have in the continuity of their self-identity and in the constancy of the surrounding social and material environments of action' (ibid.: 92). It incorporates trust in the reliability of persons and things. Trust, therefore, may be regarded as a means of dealing psychologically with risks that would otherwise paralyse action or lead to

feelings of engulfment, dread and anxiety. Without trust, people could not engage in the 'leap of faith' that is required of them in dealing with these expert knowledge systems of which they themselves have little understanding or technical knowledge because they have not been trained in them.

Trust may result from reflexive calculation or else simply from choosing to invest faith in an individual or organization. It allows individuals to develop a cocoon of invulnerability which enables them to get on with life, to fend off their knowledge of the risks that await them at every turn. This protective cocoon is sometimes pierced by experiences that highlight the existence of these risks and call into question established routines. Giddens calls these experiences 'fateful moments', which must be dealt with by reference to moral or existential criteria. Sometimes people may deliberately seek out these risks (see discussion of this in Chapter 8). But it is often the case that the cocoon is re-established, allowing the sense of relative invulnerability to return and thus fending off dread and anxiety (Giddens 1991:40). Everyday rou tines are also vital to the establishment and maintenance of ontological security, allowing people to habitually deal with dangers and associated fears (ibid.: 44).

Notions of fate tend to re-enter in the face of lowprobability but high-consequence risks over which individuals have no personal control. Sometimes the differences between expert knowledges will be so great that the individual is forced to rely on fate, to take a chance. Deciding simply to trust in abstract systems or to decide that fate will take its own course regardless—what Giddens terms 'pragmatic acceptance'—ostensibly relieves the burden of anxiety (1990:133). However, he claims that as a result, anxieties are simply pushed deeper into the unconscious. Fear is therefore not dispelled, but rather displaced, for the possibility of disaster is still there. Other reactions to risk are 'sustained optimism', or holding onto Enlightenment faith in providential reason, 'cynical

pessimism', or dampening the emotional impact of anxieties by approaching them with a humorous or world-weary perspective, and 'radical engagement', or an attitude of practical contestation towards perceived sources of danger, such as is found in the new social movements such as the green movement (1990:135–7).

Like Beck, Giddens has also recently published on love and sexual relationships in his book *The Transformation of Intimacy* (1992), where he discusses the risks associated with 'opening' oneself to another and seeking the 'pure' relationship. Giddens argues that people have become dependent on a very small number of personal relationships, particularly those involving romantic love and marriage, to find a sense of security and to construct subjectivity. He claims that there is a high level of anxiety and lack of certainty around love relationships in a social world of changing gender and family roles, characterized by a tension between retaining a sense of autonomy as an individual and a need for close relationships with others.

Giddens introduced his notion of the 'pure relationship' in Modernity and Self-Identity, arguing that it is 'one in which external criteria have become dissolved: the relationship exists solely for whatever rewards that relationship as such can deliver' (1991: 6). Trust plays a role in the pure relationship, mobilized through joint disclosure of personal thoughts and feelings rather than being anchored in criteria outside the relationship itself, such as social duty or traditional obligation. The 'pure relationship', therefore, is also highly reflexive, requiring work maintain constant to it. It presupposes 'commitment', but this commitment must be actively won and negotiated, based on intimacy. The sphere of personal relationships is seen to be a place where intimacy may be established, bolstering one against the uncertainties of the external world. Yet personal relationships and intimacy are themselves dogged by risks, because of the greater 'openness' and breakdown of traditional norms. There is far more scope for variation in intimate relationships such

as marriages and other close love relationships, such as those between same-sex partners and between parents and children (ibid.: 12).

Trust at the personal level is a project to be worked at, demanding that each party 'open out' to the other, in the absence of fixed codes (Giddens 1990:121). Giddens attributes our obsession with the quality of relationships as resulting from this need to work on trust: erotic involvement particularly calls for self-disclosure. To know the other, one must know oneself: 'Erotic relations involve a progressive path of mutual discovery, in which a process of self-realisation on the part of the lover is as much a part of the experience as increasing intimacy with the loved one' (ibid.: 122). The trust that one invests in an intimate other can be severed at any time: through a broken love affair, the intimate other becomes a stranger. The 'opening up' of oneself to another, therefore, is characterized by ambivalence, anxiety and risk. As a result, intimate relationships are the site of profound insecurity at the same time as they hold out the promise of ontological security.

CONCLUDING COMMENTS

There are major convergences between Beck and Giddens' theorizing on risk in the context of late modernity. Both see the concept of risk as a central concern in the contemporary era, emerging from the processes of modernization. Risks are seen to have changed in their modern society, having greater character in late ramifications in their impact across space and time. Both Beck and Giddens are interested in the political aspects of risk, singling out reflexivity as a primary response to uncertainty and insecurity in late modernity. They identify a greater awareness on the part of lay people that the claims of 'experts' about risk are often uncertain or clash with each other, and a willingness on lay people's part to challenge experts, governments and industry in relation to

risk concerns. Beck and Giddens extend their discussion of reflexivity into the realm of private life and intimate relationships, remarking upon the ways in which risk and its associated reflexivity have permeated this realm. Both Beck and Giddens take a weak social constructionist approach to risk and share an approach which is founded in critical structuralism, in focusing their attention on how risk is generated and dealt with at the macro-structural level of society, the political implications of this and the social conflicts that arise.

There are also a number of important differences, however, between the two theorists' takes on risk. One is the relationship that is implied between risk and reflexivity to risk. Beck implies that a heightened degree of risk reflexivity is the outcome of a greater number of risks being produced in the late modern era. Giddens sees the relationship as being the other way round. Risks are not greater in number in late modernity. It is simply that they are thought to be greater, because the nature of subjectivity in general has changed to an approach to life that is far more sensitive to the possibility of risk than in previous eras. Further, Beck and Giddens represent expert knowledge systems in different ways. For Giddens, reflexivity takes place through expert systems and is reliant upon lay people's trust in expertise. For Beck, reflexivity is a critique of expertise, based not in trust but distrust of expert systems, particularly in relation to environmental hazards. Giddens also focuses more on selfreflexivity, reflexivity directed towards the body and the self, than does Beck, who places greater emphasis on individuals' reflexive critique of the social and thus is more challenging of current social arrangements than is Giddens (Lash 1994:116; Lash and Urry 1994:38).

Some major criticisms of the reflexive modernization thesis have been advanced. One criticism is that Beck and Giddens argue that the reflexive critique of science and other expert knowledge systems, as well as social movements, are features solely of late modernity, and were not found in earlier modernity. Some commentators have argued that this is not the case, and that Beck and Giddens' representations of modernity are simplistic, not acknowledging the complexity of responses to expert knowledges. Lash (1993:5) asserts, for example, that modernity is by definition reflexive, involving continual monitoring of itself, even if through convention rather than through individualization. Other criticisms put forward are that Beck and Giddens are too speculative, making broad and loose speculations about structural and organizational processes, without grounding these specifically enough in the actual processes and experiences of institutional and everyday life. Further, it may be argued that Beck and Giddens do not sufficiently acknowledge the communal, aesthetic and shared symbolic aspects of risk in their focus on individualization (Lash 1993, 1994; Alexander 1996; see also discussion in Chapter 6).

Despite these criticisms, Beck's and Giddens' speculations on the nature of risk in contemporary societies have been enormously influential in Anglophone sociology, more so than the other perspectives discussed in this book. This is not surprising, for their insights into the structural and political features of risk, the changes in the meanings of risk over the eras of pre-modernity, early modernity and late modernity and the implications of current ideas about risk for subjectivity and social relations are valuable and suggestive.

5 RISK AND GOVERNMENTALITY

In Chapter 4, the views on risk as a product of late modernity put forward by Beck and Giddens were discussed. Those who have taken up a perspective on risk drawing on the writings of Michel Foucault are also interested in the ways in which risk operates in late modernity, particularly in relation to the political ethos of neo-liberalism, which currently dominates in Anglophone countries. They similarly see the intensification of discussions of risk and risk practices as an outcome of the social changes occurring in the wake of modernization.

One major difference between the two perspectives is that while the 'risk society' approach tends to take a weak social constructionist approach to risk in concert with a critical structuralist perspective, advocates of Foucauldian approaches mostly adopt a 'strong' version of social constructionism and a poststructuralist approach to power relations. The concept of discourse, as I noted in Chapter 2, is integral to Foucauldian theorizing. An important insight offered by Foucauldian perspectives on risk is the ways in which the discourses, strategies, practices and institutions around a phenomenon such as risk serve to bring it into being, to construct it as a phenomenon. It is argued that it is only through these discourses, strategies, practices and institutions that we come to know 'risk'. They produce 'truths' on risk that are then the basis for action. For Foucauldian writers, therefore, the nature of risk itself is not the important question for analysis. Risk is seen as a 'calculative rationality' rather than as a thing in itself (Dean 1999).

This chapter looks at the ways in which the discourses, knowledges, strategies, practices and institutions that have developed around risk both reflect and construct a distinct approach to selfhood, society and the government of populations. It begins with a discussion of the relationship between governmentality and risk, and goes on to explore the movement from the concept of 'dangerousness' to that of 'risk' in medical, legal and social welfare discourses. Then follows a discussion of three types of contemporary risk rationalities: insurantial risk, epidemiological risk and case-management or clinical risk.

GOVERNMENTALITY

Michel Foucault himself did not dwell specifically on the topic of risk in his writings at any great length. However, much of what he had to say on governmentality and modernity has been considered relevant by a number of scholars who have applied some of his ideas to the analysis of risk as a sociocultural phenomenon. Governmentality is the approach to social regulation and control that according to Foucault (1991) began to emerge in the sixteenth century in Europe, associated with such social changes as the breakdown in the feudal system and the development of administrative states in its place, based on the principles of legitimate rule. By the eighteenth century, the early modern European states began to think of their citizens in terms of populations, or 'society', a social body requiring intervention, management and protection so as to maximize wealth, welfare and productivity. Such features of populations as demographic estimates, marriage and fertility statistics, life expectation tables and mortality rates became central to the project of a technology of population. The body of both the individual and that of populations became the bearer of new variables. These variables included not only those

between the healthy and the sick, the strong and the weak, the rich and the poor, and so on, but between the more or less utilizable, more or less amenable to profitable investment, those with greater or lesser prospects of illness or death and with more or less capacity for being usefully trained (Foucault 1984:278–9).

Governmentality as a strategy and rationale, Foucault claims, has dominated political power in western countries since the eighteenth century. In its contemporary form it is characterized by an approach to political rule, neoliberalism, which champions individual freedom and rights against the excessive intervention of the state. The domain of government is extensive, focusing on the complex of human's interactions with a diversity of phenomena:

The things with which in this sense government is to be concerned are in fact men [sic], but men in their relations, their links, their imbrication with those other things which are wealth, resources, means of subsistence, the territory with its specific qualities, climate, irrigation, fertility, etc; men in their relation to that other kind of things, customs, habits, ways of acting and thinking, etc; lastly, men in their relation to that other kind of things, accidents and misfortunes such as famine, epidemics, death, etc.

(Foucault 1991:93)

Foucault, like Beck and Giddens, emphasizes the role of expert knowledges in the constitution of late modern subjectivity. Expert knowledges, he argues, are integral to the reflexive techniques and practices of subjectification, or the formation of certain types of subject. For Foucault, however, expert knowledges are not transparently a means to engage in reflexivity. Rather, they are seen as pivotal to governmentality, providing the guidelines and advice by which populations are surveyed, compared against norms, trained to conform with these norms and rendered productive. Central to these technologies is normalization,

or the method by which norms of behaviour or health status are identified in populations and sub-groups of populations. Through normalization, the late modern individual is fabricated within a network of instruments and techniques of power. The technologies of mass surveillance, monitoring, observation and measurement are central to this disciplinary power, helping to construct understandings of bodies in space and time and to use these understandings to regulate them.

From this perspective, risk may be understood as a governmental strategy of regulatory power by which populations and individuals are monitored and managed through the goals of neo-liberalism. Risk is governed via a heterogeneous network of interactive actors, institutions, knowledges and practices. Information about diverse risks collected and analysed by medical researchers. statisticians, sociologists, demographers, environmental scientists, legal practitioners, statisticians, bankers and accountants, to name but a few. Through these neverceasing efforts, risk is problematized, rendered calculable and governable. So too, through these efforts, particular social groups or populations are identified as 'at risk' or 'high risk', requiring particular forms of knowledges and interventions. Risk, from the Foucauldian perspective, is 'a moral technology. To calculate a risk is to master time, to discipline the future' (Ewald 1991:207).

The strategies of governmentality, expressed in the neoliberal states that emerged in the west in late modernity, include both direct, coercive strategies to regulate populations, but also, and most importantly, less direct strategies that rely on individuals' voluntary compliance with the interests and needs of the state. These strategies are diverse and multi-centred, emerging not only from the state but also other agencies and institutions, such as the mass media. A crucial aspect of governmentality as it is expressed in neo-liberal states is that the regulation and disciplining of citizens is directed at the autonomous, selfregulated individual. Citizens are positioned in governmental discourses, therefore, as active rather than passive subjects of governance. Rather than mainly being externally policed by agents of the state, individuals police themselves, they exercize power upon themselves as normalized subjects who are in pursuit of their own best interests and freedom, who are interested in self-improvement, seeking happiness and healthiness (Gordon 1991).

As will be explained in greater detail below, the concept of risk, as it is developed through normalization, initially deflects attention away from individuals and their behaviours towards aggregates or populations. The information gathered about risk from population data, however, is then often employed in advice to individuals about how they should conduct their lives. Discourses on risk are directed at the regulation of the body: how it moves in space, how it interacts with other bodies and things. These discourses also contribute to the constitution of selfhood, or subjectivity, and thus are part of the panoply of 'practices' or 'technologies of the self' (Foucault 1988). Through the technologies of the self, the individual becomes the 'entrepreneur of himself or herself in terms of attempting to maximize her or his 'human capital' (Gordon 1991:44). People attempt to 'transform themselves in order to attain a certain state of happiness, purity, wisdom, perfection, or immortality' (Foucault 1988:18). In doing so, they seek out and adopt advice from institutional governmental agencies, from experts who have problematized areas of life as pervaded by risk. As expert knowledge about risk has proliferated in late modernity, the various strategies which individuals are required to practise upon themselves to avoid risk have equally proliferated.

Consider the example of the pregnant woman in the contem porary era. More so than in previous eras because of the growth of risk-related knowledges and technologies surrounding pregnancy, this woman is surrounded by, and constructed through, a plethora of expert and lay advice.

This advice is directed at how she should regulate her body with its precious cargo, the foetus, which is portrayed as being highly fragile and susceptible to risk at every stage of its development. Even from the time a woman may be considering the idea of child-bearing, she is exhorted to engage in certain practices to ensure that fertilization will take place successfully and that her body is at its peak state of health. She is encouraged to read as much as possible about pregnancy and childbirth, so that she knows what to expect and what risks to avoid. She is told to avoid smoking, drinking alcohol and coffee and taking other drugs, to eat a nutritious diet and engage in regular exercise.

Once pregnant, these strategies and more must be adhered to by the woman. The pregnant woman must be highly careful of any food she eats, avoiding a range of foods that may contain listeria bacteria or toxoplasmosis protozoa, both of which may cause miscarriage or birth defects. She is advised to closely monitor her body for signs of an ectopic pregnancy or miscarriage, such as abdominal cramps or vaginal bleeding. She should not take any medicines, unless she has first checked their safety for the foetus with her doctor. She is encouraged to regularly see a doctor for antenatal checks, including a series of blood tests and such tests as ultrasound and, in some cases, amniocentesis (testing of the genetic makeup of the foetus) to monitor the health and normality of the foetus. She is encouraged to attend antenatal classes to prepare for childbirth.

Many of the discourses of risk that surround the pregnant woman suggest that it is her responsibility to ensure the health of her foetus, and that if she were to ignore expert advice, she is culpable should her baby miscarry or be born with a defect. The pregnant woman, positioned therefore. in surveillance, monitoring, measurement and expert advice that requires constant work on her part: seeking out knowledge about risks to her foetus, acting according to

that knowledge. Yet the discourses of risk that surround her are generally embraced willingly, because the woman herself wants to maximize the health of her foetus, to achieve the 'perfect child'. Although some women in some countries such as the United States have been prosecuted and imprisoned for 'foetal endangerment' by refusing to take medical advice or give up using certain drugs (Handwerker 1994), most women are not coerced through overt disciplinary means to accede to expert advice. Noone forces them to buy books on pregnancy, to watch their diet, give up alcohol, attend antenatal checks and classes. Because pregnant women have been discursively positioned within a context in which the general consensus is that foetuses are fragile and that it is up to the pregnant woman to protect them, and that infants are very important individuals who deserve the best start in life, most women voluntarily engage in such riskavoidance strategies, accepting the responsibility implied.

To resist these strategies is difficult, for it is tantamount to declaring that the woman does not care about her own health and welfare, and more importantly, that of the foetus she is carrying and is expected to protect and nourish in the proper maternal manner. Indeed, she may actively demand greater access to medical surveillance, such as numerous ultrasounds, in the attempt to alleviate her anxieties and concerns about her foetus and the risks to which it is exposed. The technologies of selfhood and embodiment in relation to pregnancy demonstrate the intersection and alignment of institutional and experts' objectives in advising and regulating the pregnant woman principally through directives intended for her to police herself, and the woman's own concerns to take such advice.

In late modern societies, not to engage in risk avoiding behaviour is considered 'a failure of the self to take care of itself—a form of irrationality, or simply a lack of skilfulness' (Greco 1993: 361). Risk-avoiding behaviour, therefore, becomes viewed as a moral enterprise relating to

issues of self-control, self-knowledge and improvement. It is a form of self-government, involving the acceptance and internalization of the objectives of institutional government. Because the project of selfhood is never complete, but rather is continuing throughout the lifespan, so too the project of risk avoidance as a technology of the self is never-ending, requiring eternal vigilance.

FROM DANGEROUSNESS TO RISK

Several commentators adopting the governmentality observed that the concept approach have 'dangerousness' has been replaced by that of 'risk' in institutions' dealings with marginalized social groups and nineteenth-century individuals. governmental In discourses, as part of the emergent new ways of thinking about the citizen, the concept of 'dangerousness' tended to be used in relation to the problems of health and crime. 'Dangerous classes' and the 'dangerous individual' were identified as possessing the inherent qualities to present a danger to themselves or to others, and therefore as prime targets for governmental intervention and treatment. The notion of 'dangerousness' was derived from expert judgements on such features as the state of living conditions and moral climate in which social groups dwelt. Members of the working class and the poor were typically constituted as dangerous. Against these dangerous classes were juxtaposed their antithesis: those (generally more socially and economically privileged) classes who were seen to be 'at risk' from the depravations or contamination of members of the dangerous classes (Kendall and Wickham 1992:11-12).

Castel (1991) links the concept of risk with that of governmentality by focusing on contemporary preventive strategies of social administration in the United States and France. He argues that these strategies are innovative in that they 'dissolve the notion of a *subject* or a concrete individual, and put in its place a combination of *factors*, the factors of risk' (ibid.: 281, original emphases). As a result, he argues, there have been changes in the ways in which intervention is carried out. No longer is the essential component of intervention 'the direct face-to-face relationship between the carer and the cared, the helper and the helped, the professional and the client. It comes instead to reside in the establishing of flows of population based on the collation of a range of abstract factors deemed liable to produce risk in general' (ibid.: 281). As a result, specialist professionals are cast in a more subordinate role, while managerial policy formations take over.

In psychiatric medicine, for example, there has been a shift over the past century from the use of the notion of 'dangerousness' used in relation to people with psychiatric disorders to that of 'risk'. In classical psychiatry, "'risk" meant essentially the danger embodied in the mentally ill person capable of violent and unpredictable action' (ibid.: 283). Dangerousness itself connoted an immanent quality of the subject, a potentiality that dwelt within and may or may not have been manifested. Therefore there could only ever be imputations of dangerousness, observation of a patient's present symptoms speculations about what these might mean for future behaviour. All insane people were deemed as carrying this potentiality for dangerousness within them, despite their otherwise benign exteriors, and were subsequently treated with such preventive strategies as confinement from the rest of the society (ibid. 1991).

In contrast, the notion of risk, although also acknowledging potentiality, is calculated through systematic statistical correlations and probabilities based on populations rather than the close observation of individuals. Risk is therefore more selective and precise, but at the same time applies to a larger group of people than the notion of dangerousness: 'A risk does not arise from the presence of particular precise danger embodied in

a concrete individual or group. It is the effect of a combination of abstract factors which render more or less probable the occurrence of undesirable modes behaviour' (ibid.: 287, original emphasis). To designated 'at risk' is to be located within a network of factors drawn from the observation of others, to be designated as part of a 'risk population'. A risk, therefore, is one step further from dangerousness in its potentiality.

Identifying and monitoring risks in populations constitutes 'a new mode of surveillance: that of systematic predetection' (ibid.: 288). This new form of surveillance may not necessarily require the actual presence of the 'risky' individual, but may be based on the monitoring of records.

To intervene no longer means, or at least not to begin with, taking as one's target a given individual, in order to correct, punish or care for him or her.... There is, in fact, no longer a relation of immediacy with a subject because there is no longer a subject. What the new preventive policies primarily address is no longer individuals but factors, statistical correlations of heterogeneous elements. They deconstruct the concrete subject of intervention, and reconstruct a combination of factors liable to produce risk. Their primary aim is not to confront a concrete dangerous situation, but to anticipate all the possible forms of irruption of danger.

(Castel 1991:288, original emphasis)

Under this new approach to surveillance, for a person to be identified as posing a risk no longer means that she or he has to be individually observed for signs of dangerousness. It is enough that she or he is identified as a member of a 'risky population', based on a 'risk profile' developed from calculations using demographic and other characteristics. Castel notes that this shift from dangerousness to risk results in the production of 'a potentially infinite multiplication of the possibilities for intervention. For what situation is there of which one can be certain that it harbours no risk, no uncontrollable or unpredictable chance factor?' (ibid.: 289).

Whereas in the early modern era members of the 'dangerous classes' were disciplined and managed via coercive, exclusionary and correctional strategies (such as incarceration), risk-based tactics and strategies may focus very little on the individual characteristics of those identified as being 'at risk', but rather direct attention at changing the environment in which such individuals operate. O'Malley (1992:262) gives as an example situational crime prevention. As a risk management strategy, situational crime prevention deals hardly at all with individual offenders. It is not interested in the causes of crime and does not support correctional strategies for managing crime. The philosophy of this approach, as articulated in an American National Crime Prevention Institute document, is that: 'Criminal behaviour can be controlled primarily through the direct alteration of the of potential victims.... As environment criminal opportunity is reduced, so too will be the number of criminals' (quoted in O'Malley 1992:262). The focus of this approach is upon prevention of crime rather than rehabilitation of offenders.

The shift in focus in expert knowledges and practices from the concept of 'dangerousness' to 'risk' has a number of consequences. The future behaviour of a marginalized individual deemed subject to external intervention and regulation is no longer linked exclusively to that individual's own behaviour, based on close observation of her or him. Because the concept of risk has emerged, which is predicated on techniques of the surveillance and measurement of populations and statistical calculations based on data derived from these techniques, marginalized individuals are now dealt with differently. Under the discourse of risk, these people are typically categorized as a member of a specific 'risk group' and their future

behaviour is gauged and the interventions that are judged to be required are based on the characteristics of this group. These calculations rest upon a notion of management that highlights the importance of rationalized and standardized assessment and prediction and a notion of the individual actor that represents her or him as behaving predictably, in alliance with patterns identified in wider populations.

CONTEMPORARY RISK STRATEGIES

The above discussion on the change from 'dangerousness' to 'risk' as a discourse and strategy of regulation and intervention tends to imply a homogeneous approach to risk in contemporary western societies. This is not quite the case. Dean (1997, 1999) has identified three types of risk rationality in neo-liberal societies. These include epidemiological risk insurantial risk. and casemanagement or clinical risk. There are certain differences between these risk rationalities, based on the types of risk calculations that are manifested and the specific risks to which they are directed.

Ewald (1991) has discussed three characteristics of insurantial risk, or that operating in the discourses and strategies of insurance. The first is that risk distinguished from a bet in that it is seen to be calculable, governed by identifiable laws. For an event to be a risk in insurance discourses, it must be possible to evaluate the probability of it happening. The second characteristic is that risk is collective, affecting a population rather than an individual: 'Strictly speaking there is no such thing as an individual risk; otherwise insurance would be no more than a wager' (Ewald 1991:203). Rather, risk is seen as something that only becomes calculable when it is spread over a population. Each individual in a specified population is understood to be a factor of risk or exposed to a risk, but not each individual is equally likely to fall prey to a risk or cause the same degree of risk (ibid.: 203).

The third characteristic of insurantial risk is that it is a capital. What is insured against is not the injury or loss but rather a capital against whose loss the insurer offers a guarantee. The injury or loss is not prevented or repaired, but is given a price for financial compensation.

Insurance, therefore, is a means for dealing with the vagaries of fate, a technology through which risk is constructed as a schema of rationality, of ordering elements of reality, allowing for a certain way of objectifying things, people and the relationships between them. Insurers 'produce risk' by rendering a range of phenomena into a risk—death, bankruptcy, litigation, an accident, a disease, a storm—through the specialized actuarial calculations available to them, and then offer guarantees against them. These phenomena would once have been accepted with fatalistic resignation: now they have become objects of risk, given value via the compensation that has been calculated for them (Ewald 1991). It is in this sense that anything can be a risk, if it is amenable to being turned into a risk through insurantial discourse, or any other kind of discourse directed at identifying and managing risk (ibid.: 200). Participation in insurance is about conducting one's life as an enterprise, to ensure that even when misfortune occurs, it has been planned for. It is, however, a socialized rather than wholly individualized approach, for insurance distributes the burden of risk among a large population and underpinned by a notion of social rights in which members of an association agree to accept responsibility for each other's burdens (Dean 1999).

The second type of risk rationality is epidemiological risk, in which the calculus of risk is undertaken by bringing together assessment of a range of abstract factors with the incidence of health outcomes in targeted populations. Epidemiological risk adopts a similar approach to insurantial risk, but has a different target—illness and disease rather than loss of capital. Epidemiological techniques involve the tracing of illness

and disease in specified populations using statistical and screening techniques, linking illness and disease with their causal variables in the attempt to predict health outcomes at the population level and thus to better control them and reduce health risks (Dean 1997:218).

In the past, epidemiological risk strategies tended to be directed not at individuals' behaviours, but rather at altering environmental conditions in the attempt to improve health at the population level. Thus, for example, nineteenth-century public health endeavours sought to tackle hygiene and sanitary conditions in the city, such as air and water quality and sewerage arrangements, to reduce the incidence of epidemic diseases. While this focus on environmental health continues to some extent at the end of the twentieth century, there has developed a far greater emphasis on individuals' 'lifestyle' choices in relation to health status (Lupton 1995). Epidemiological risk factors are now often used to exhort individuals to engage in self regulation. Thus, for example, if a certain population group is identified through statistical calculations as being at 'high risk' of developing heart disease, based on such attributes as gender, age and diet, then members of that group are then encouraged to deal with the risk factors themselves. This process does not necessarily involve consultations and examination of individuals by health professionals, but rather often takes place through mass-targeted media campaigns which rely on individuals identifying themselves as being 'at risk' and taking steps voluntarily to reduce their exposure to risk (Lupton 1995). This is an example of 'government at a distance', for it relies upon voluntary participation in technologies of self-surveillance and a sense of selfresponsibility rather than direct intervention.

The third type of risk rationality, case-management risk, is linked to clinical practice with individuals deemed to be threatening or disruptive in some way to the social order (the mad, the unemployed, the criminal, the dysfunctional, the poor, the long-term unemployed). Risk calculation in this type involves the qualitative assessment of risk for individuals and groups who are deemed to be 'at risk' (Dean 1997:17). In contrast to the other two types of risk rationalities, the case-management approach uses more individualistic sources of data derived from interaction with and observation of specific clients, such as interviews, case-notes and files. Once risk is assessed, techniques for managing it on the part of the relevant experts (for example, social workers, health workers, police officers) are brought into play. These include therapeutic practices directed at self-help through expert assistance, pedagogic practices designed to train dangerous others and more coercive measures such as detainment and imprisonment, removing the 'risky' individuals from society (Dean ibid.: 217–18).

Population-based risk strategies, however, also now enter the clinical arena. When patients visit their doctors, for example, their symptoms are not only treated as specific to them as individuals, but as manifestations of the patient's location in a wider sociodemographic context. In this way, epidemiological calculations about the likelihood of the occurrence of a condition in a given population are implemented upon the bodies of individuals, by applying risk categories derived from large-scale data sets. The case-management approach that is central to clinical strategies, with its emphasis on individual pathology and therapeutic intervention, is brought together with epidemiological risk, with its emphasis on indirect intervention via populations (Dean 1997).

Dean argues that the case-management type of risk rationality has proliferated in neo-liberal societies, moving from the spheres of social work and clinical medicine to address such problems as unemployment and 'welfare dependency'. The language of risk is taking over from that of need or welfare in the literature on personal social services, such as probation, mental health and childcare services, with risk assessment, risk management, the

monitoring of risk and risk-taking itself having become the raison d'être and organizing principle of agencies providing such services. Risk-related discourses and strategies have taken on a key role in decision-making about service delivery, including the rationing of services and decisions about need.

For all three types of risk rationality, individuals and groups are increasingly expected to engage in practices identified as ways of avoiding or minimizing the impact of risks to themselves. This approach has been called by critics the 'new prudentialism', a neo-conservative approach which progressively removes the responsibility for risk protection from state agencies—as embodied in social insurance for such misfortunes as unemployment and ill health—and places it in the hands of the individual or community-based groups (O'Malley 1992; Dean 1997, 1999). As a result, the concept of risk has become more privatized and linked ever more closely to the concept of the entrepreneurial subject, calling into question the very notion of social rights: 'Here, we witness the "multiple responsibilization" of individuals, families, households and communities, for their own risks-of physical and mental ill health, of unemployment, of poverty in old age, of poor educational performance, of becoming victims of crime' (Dean 1997:218).

For example, as Dean notes, in several neo-liberal societies there has been an increasing focus on the importance of individuals managing their own risks by taking out private insurance rather than participating in the social insurance schemes offered by the state. The latter served to offer security through the spreading of the costs of unfortunate events among the general population, so that risk was socialized. In doing so it deflected attention away from those designated as being 'at risk', providing them with a security net rather than exhorting them to change their behaviour. In contrast, private insurance arrangements place the onus on individuals to take responsibility for insuring themselves against

misfortune. It is accompanied by a range of other risk strategies that remove the responsibility for managing and dealing with risk from the state.

Under the 'new prudentialism', the acceptance of personal responsibility is presented as a practice of freedom, relief from state intervention, an opportunity for the entrepreneurial subject to make choices about the conduct of her or his life. More and more domains of life are identified as amenable to and requiring of these choices. In the context of neo-liberal democracies, which value self-autonomy over direct state intervention, these strategies are seen both to work to minimize risks and to protect individuals' rights. In this context, the role of government is to provide advice and assistance for the self-management of risks, encouraging the active, free citizen who voluntarily engages in risk avoidance, rather than providing large-scale financial support.

Contemporary technologies of risk calculation and control, therefore, comprise one aspect of a change in ways of viewing the role of society. What 'the social' is understood to be has changed from notions of a mass collectivity to dynamic smaller groupings. We are progressively understanding and acting upon ourselves not as members of a specific society or through the ethos of the welfare state, but as self-actualizing individuals who move between loose and fluid social aggregations, taking up different roles in each. Small community or affiliationbased groupings are set in place to deal with such phenomena as risk, which have limited and dynamic constituencies and interests (Dean, 1997, 1999). This is taking place in a sociocultural and historical context in which dominant notions of selfhood privilege the self who is able to exert strong control over her or his mind and body, constantly engages in self-examination, is able to engage in self-denial for the greater good and readily takes up the injunctions of experts in making 'lifestyle' choices. Those individuals who are deemed to be at 'high risk' either of being a victim of risk or of perpetrating risk are

expected to take control to prevent risk through their own actions rather than rely on social insurance apparatuses as a safety-net.

This representation of the individual is that of *homo economicus*, a subject who is invested with additional moral and political characteristics and conforms to the self-interested and responsible actor found in neoconservative discourses. Situational crime discourse, for example, represents the potential offender as a universal 'abiographical individual', a 'rational choice' actor who weighs up the pros and cons before committing an offence. So too, victims are understood as rational choice actors, with the responsibility to protect themselves, and it is therefore regarded as their fault if they become vulnerable to crime (O'Malley 1992: 264–66).

As the welfare state is wound back, there is less incentive for the state to provide social insurance schemes such as unemployment benefits or socialized health insurance: 'in the present era, the success of programs inspired by economic rationalists and neo-conservatism has been stripping away socialized risk management, and replacing it with a programmatic combination of privatized prudentialism and punitive sovereignty' (ibid.: 261). The lack of interest in the biography or motivation of the 'at risk' individual deflects attention away from the socioeconomic underpinnings of risk, and divorces misfortune from questions of social justice. This leads back to the early modern risk strategies of coercion and punishment and the construction of new 'dangerous classes' requiring active surveillance and disciplining. It appears that societies dominated by neo-liberal politics in the late mod-ern era are returning to previous forms of discipline in relation to individuals and social groups that are identified as being 'at risk' or imposing a risk upon others. In relation to crime prevention, for example, the broader social structural underpinnings of crime, such as socioeconomic disadvantage, is ignored in favour of

strategies that are punitive for those who are seen to lack self-control (ibid.: 265).

CONCLUDING COMMENTS

This chapter has highlighted the ways in which the concept of risk, employed to address governmental concerns, has contributed to the production of certain kinds of rationalities, strategies and subjectivities. It has been argued that according to the Foucauldian perspective, risk strategies and discourses are means of ordering the social and material worlds through methods of rationalization and calculation, attempts to render disorder and uncertainty more controllable. It is these strategies and discourses that bring risk into being, that select certain phenomena as being 'risky' and therefore requiring management, either by institutions or individuals.

The accounts provided by Foucauldian scholars of risk have shown that it is not simply a matter of risk becoming less calculable, or shifting from local to global contexts, as the 'risk society' thesis would have it (Dean 1999). Rather, changes in risk rationalities have occurred which have resulted in risk being conceptualized and dealt with in diverse ways that have strong links to ideas about how individuals should deport themselves in relation to the state. Some of those taking up a Foucauldian perspective have remarked upon recent changes in the governance of risk, in which there is far less reliance upon social insurance and far more upon individual self-management and self-protection from risk. This is an outcome of the political ethos of neo-liberalism, which emphasizes minimal intervention on the part of the state and emphasizes 'self-help' and individual autonomy for citizens.

Foucault himself and those taking up his perspectives on the regulation of subjects via the discourses of governmentality may be criticized for devoting too much attention to these discourses and strategies and not enough to how people actually respond to them as part of their everyday lives. The question of how riskrelated discourses and strategies operate, how they may be taken up, negotiated or resisted by those who are the subject of them, remains under-examined. Further, the Foucauldian view of the self tends to represent it as universal, without recognizing differences between the ways in which people of different gender, age, ethnicity and so on may be treated by and respond to these discourses and strategies differently. The next chapter takes up these questions, looking at how people respond to risk in the contexts of their everyday lives and located within particular social structures, power relations and subgroups.

6 RISK AND SUBJECTIVITY

The theoretical perspectives on risk that have been reviewed thus far provide various approaches to understanding how concepts of risk influence subjectivity. For the 'cultural/symbolic' approach, risk is used to reproduce and maintain concepts of selfhood and group membership, particularly in defining self from the polluting or 'risky' Other. For the 'risk society' perspective, reflexive awareness and concern about risk pervades modern sensibilities, creating new forms of relating to the self and others, including experts and institutions. For the 'governmentality' perspective, risk discourses contribute to the constitution of a particular type of subject: the autonomous, self-regulating moral agent who voluntarily takes up governmental imperatives. While these insights are important in the abstract theorizing of risk responses, the writers from whom these perspectives are drawn have tended not to explore in detail the diverse and dynamic ways in which lay people respond to risk.

This chapter looks at how lay people engage reflexively or otherwise with risk. In doing so, the discussion frequently draws on the empirical findings of studies exploring the ways in which people construct their risk knowledges in the context of their everyday lives. The chapter begins with a review of risk knowledges and reflexivity. It then goes on to explore the importance of social structures and power relations and finally the role

of aesthetic or hermeneutic judgements and habituation in people's responses to risk.

RISK KNOWLEDGES AND REFLEXIVITY

Late at night on 30 July 1997, eighteen people were killed when a landslide occurred at Thredbo village, a popular ski resort in the alpine area of Australia. A ski lodge collapsed and slipped down a steep slope, falling onto a second lodge and crushing the occupants of both. Only one occupant survived the incident, eventually being dug out by rescue workers after being entombed for days under a pile of concrete and rubble. This incident received a high level of national news media coverage, particularly in relation to the gradual recovering of the bodies of the dead from the wreckage of the lodges and the rescuing of the sole survivor.

Six months later, the findings of a geotechnical study into the stability of other buildings in the area were released, identifying a number of other lodges as being 'at risk' of being damaged or destroyed in a similar landslip. A Sydney newspaper reported the reaction of one inhabitant of a lodge that was labelled as being 'at high risk'. She described her shock at this finding, saying that the lodge was 'our home' and that her husband and two daughters had never felt unsafe living there: 'We love living here. It's just beautiful. It's been very special to our kids. I have never felt unsafe here, so having words like "risk" being thrown around now is just devastating because we don't feel that risk. Even when the [July] slide happened, I never felt unsafe.' Another woman interviewed, who was staying at a lodge designated by the engineers who wrote the report as being 'at very high risk', said that she was not concerned by the finding: 'You go when your time is up,' she said. 'You could get killed on the road.' (quoted in the Sydney Morning Herald, 31 December 1997).

As the psychometric literature has documented (Chapter 1), 'lay' people often hold perspectives on risk that differ from those put forward by 'experts'. But as Douglas has further contended (Chapter 3), this is not simply a matter of their 'ignorance' or inability to understand probabilities. The existence of varying perspectives on 'risk', among both experts and lay people, suggests that the phenomenon of risk is a production of competing knowledges about the world. The words of the first woman quoted above suggest, for example, that people may feel particularly safe in places in which they have chosen to live and bring up their families, with which they have an everyday familiarity, in sharp contrast to the assessments of experts. Alternatively, they may adopt a fatalistic approach, as articulated by the second woman quoted in the newspaper article. Both approaches have their own logic and rationale, and make eminent sense to those who adopt them as part of their views on life.

It was argued in earlier chapters that risk discourses position social actors in certain specific ways. These discourses tend either to identify subjects as responsive to risks that are identified as threatening them, actively making choices in relation to risk prevention, or as riskmakers, the causes of risk, and thus requiring observation, regulation and discipline. As Beck and Giddens have pointed out (Chapter 4), as a consequence modernization and individualization, increasingly more aspects of life are considered to be subject to human agency. The contemporary self, therefore, is placed in a position of making choices about a myriad of aspects of life, such that 'Choosing is the inescapable fate of our time' (Melucci 1996:44).

With an apparent openness of lifestyle comes additional burdens, including the assumption of all the risks that go with decision-making (ibid.: 44). The complexity of living in a late modern world, in which change is rapid and intense and the number of choices to make have proliferated, renders choice-making very difficult and fraught with uncertainty:

We live with an increasingly large quota of uncertainty and we are often overwhelmed. What are we to do in a *different* context? How can we tackle a *new* problem? Or, more simply and generally, *what* are we do to do, which choice should we make? Many of our tasks become exercises in problem-solving, compelling us to acquire information, study the instructions, and, in the end, make a choice.

(Melucci 1996:45, original emphases)

People are expected to take personal responsibility for these choices, to follow their own interests. Hence the proliferation of control over their environment than they were in previous eras: risk discourses, which represent humans as far more able to exert 'The idiom of risk presupposes ideas of choice, calculation and responsibility, so that whether the risk attitude prevails or even makes sense in a given area of life depends on the degree to which that area is regarded as fixed and inevitable, or as subject to human agency' (Szerszinsky *et al.* 1996:12).

It is clear that many aspects of people's lives are influenced by their awareness of risk and the responsibilities involved with avoiding risks. It is also evident that individualization, which emphasizes personal responsibility for life outcomes, is dominant in late modern societies. Many people appear to have accepted the notion that one should make oneself aware of risks and act in accordance with experts' risk advice so as to prevent or diminish the impact of risk. Indeed the notion that one is personally responsible for the control of risk appears to be acculturated very early in life. Research conducted by Green (1997) found that even among children as young as seven years, the notion that accidents can be prevented and must have a cause which can be used to blame

someone, was very dominant. The English children in her study, ranging in age from seven to eleven, argued that accidents were more likely to be the result of known or knowable sets of risk factors than blameless and unpredictable events, and therefore in theory could be prevented through responsible and rational behaviour. Almost all the children accepted the responsibility for keeping themselves safe from harm and saw themselves as culpable if an accident happened to them. Their stories about engaging in risks or avoiding risks also served as signifiers for social identities and relationships beyond the meaning of accidental injury: representing oneself as a 'careful' or 'daring' person to one's friends, for example, as part of conforming to friendship group norms.

Accepting personal responsibility for risks and taking up experts' risk advice, however, are not the only responses to risk, despite the claims of writers such as Beck and Giddens. The notion of reflexivity itself suggests a rational, calculating actor: people are portrayed as choosing rationally between various perspectives on risk provided by expert knowledge systems. It also appears to give credence to the role played by experts in constructing risk meanings over that of lay actors. The reflexivity thesis implies that individuals develop and exercise reflexivity in response to expert knowledge, rather than generating their own risk knowledges through their own experiences of the world. While they acknowledge that reflexivity is practised in the sphere of the intimate and the everyday, Giddens and Beck tend to suggest that this reflexivity is again primarily in response to expert knowledge systems. They give little recognition to the ways in which lay actors drawn upon their own situated knowledges of the world in constructing risk understandings and responding to experts' pronouncements on risk (Wynne 1996).

It has been argued by other commentators that while risks may be debated at the level of expertise and public accountability, they are dealt with by most individuals at the level of the local, the private, the everyday and the intimate. Lay knowledges tend to be far more contextual, localized and individualized, reflexively aware of diversity and change, than the universalizing tendencies of expert knowledges (Wynne 1996:70). Lash and Wynne (1992; see also Wynne 1996) highlight what they see as the multilayered response to risk on the part of lay people as a form of 'private reflexivity' which, they argue, must be the basis for the more public forms of reflexivity (that is, debate about risk at the political level).

Some sociologists have sought to demonstrate that lay perceptions of risk are founded on sources of knowledge that should be acknowledged as being equally as important and rational as scientific expert assessments, which themselves are often based on 'optimistic fantasies about behaviour in the real world' (Wynne 1989:39). Their argument tends to revolve around debates concerning the supposed 'irrationality' of lay people's responses to risk, contending that what may seem 'irrational' is in fact based on rational judgement. For example, Wynne (1989) studied the claims of British agricultural workers that the herbicide 2,4,5-T (Agent Orange) was harmful to their health, against the claims of the manufacturers and scientists that 'if properly used' the chemical was not hazardous. Wynne argues that the claims of the workers were founded on their everyday use of the herbicide and their empirical observations that in practice, because of the contingencies of their work, it was rarely prepared and used according to the manufacturers' safety instructions. The scientists judging the risks of the herbicide used implicit assumptions based on ideal-world use failed to recognize that most workers did not follow the prescribed instructions for preparation and use.

Members of the lay public also incorporate into their assessment of risk their pre-established knowledge of how the relevant industries and regulatory bodies have tended to deal with risks in the past (Wynne 1989, 1996). People judgements about the persuasiveness trustworthiness of experts, involving recognition of the

sociocultural frames and interests which shape experts' risk knowledges. Rather than this knowledge being a 'distorting' factor in lay assessment, it is an important element of their judgement of risk. It is not simply a matter, therefore, of individuals weighing up the relative magnitude of physical risks as different experts assess them. Rather, they go through a process of evaluating risk experts and institutions themselves. Such responses are not rationalist-calculative assessments of science, but are cultural, reflecting on different ways of understanding and representing such phenomena as agency, predictability, control and values. As members of social groups and networks, people's responses to risk are embedded within these relationships, and are therefore collective as well as developed through individual biographies (Wynne 1989, 1996).

Lay actors also often resist or directly challenge experts' judgements on risk. In constructing private reflexivity, people struggle with the reconciliation of different and often conflicting interests and identities: they 'informally but incessantly problematise their own relationships with expertise of all kinds, as part of their negotiation of their own identities' (Wynne 1996:50). Even where there seems to be no evidence of public dissent over risk, at the everyday, personal level such dissent constantly takes place. This suggests that the relationship that people have with 'expert' knowledge systems is highly complex and ambivalent. It is not simply a question of lay people deciding which of two or more bodies of dissenting expert knowledge to trust when they are making judgements about risk. Rather, they construct their own expert knowledges, with or without the use of risk professionals' knowledge (Wynne 1996).

Lay people are aware of their dependency on expert knowledges when it comes to disputes about risk. They are also aware of their lack of agency and opportunity, as 'non-experts', to challenge expert knowledges, even if the expert knowledges are uncertain or conflicting (Wynne 1996; Michael 1996). In the context of their everyday lives, aspects of expert knowledge may be considered irrelevant or distracting. What is considered to be 'ignorance' on the part of experts may be thought of by lay actors as a deliberate ignoring or avoidance of expert knowledge because it is regarded as essentially peripheral to the key issues at stake, or at worst, inaccurate and misleading. 'Ignorance' on the part of lay people becomes a positive and agential choice in these cases, not a passive deficit requiring greater access to expert knowledges (Michael 1996:119-20).

Sociological research investigating the ways in which logics of risk are established, maintained or revised as part of individuals' location with specific sociocultural settings points to the complexities and ambiguities of such knowledge formation. It shows that these alternative rationalities, often portrayed by experts as inaccurate or irrational, often make sense in the context of an individual's life situation, including the frameworks and meanings that shape subjectivity and social relations and the institutions and social structures within which individuals are placed. The reflexivity of lay people in relation to risk may develop from their observations of the ways in which everyday life operates and from conversations and interactions with other lay actors. For example, people often develop a fatalistic attitude towards risk because they have observed that life does not always 'play by the rules'. Someone who drinks heavily and smokes may live to a ripe old age, while an ascetic non-smoking, jogging vegetarian may die young. Most people can cite such examples among the people they know (Davison et al. 1992). Conforming to expert advice about how to avoid risk, therefore, is seen not necessarily to guarantee protection from harm. If fate steps in, preventive action is seen to be useless.

Macgill (1989) looked at the competing logics of risk emerging in relation to an alleged link between radioactive discharges from a nuclear processing installation at

Sellafield in the north of England and an increase in the incidence of leukaemia and other cancers among children in the surrounding area. She found that among the residents of Seascale, a village close to Sellafield and therefore most directly 'at risk', there was a wide diversity of opinion concerning the risk of radioactive discharges from the installation, ranging from those who were very concerned about the risk to those denying that there was a risk. Expert assumptions about 'the public view', therefore, served to homogenize what was a diverse range of attitudes and feelings about the alleged risk. Based on her findings, Macgill argues that people's perceptions understandings of risk are established over a life-time of personal experiences as well as their location within social milieux and networks of communication. These include their use of the mass media and conversations with others as well as expert knowledges:

Social interaction leads to positions being confirmed, adapted and, in some cases, discovered or expressed for the first time. Positions and arguments which are 'successful' get repeated and so both an individual's own rationalities and the cohesion of particular social groups is maintained and, likewise, the rejection or suspicious appraisal of opposing or threatening positions. Prejudices develop. These are not irrational whims that may be ignored by anyone with a serious interest in understanding perceptions or that can be suppressed by an individual. Rather, they are forces which act as filters to people's openness to the world and are a part of their very being. They are simply conditions whereby what people experience or encounter says something to them.

(Macgill 1989:57)

Risk positions not only emerge from people's locations within social milieux, but also serve to position them

within such milieux, as means of developing and supporting social cohesion or group membership (ibid.: 57). Risk positions, therefore, may also be important to people's sense of self-identity as part of a social group or sub-culture. People who live in areas designated as 'high risk', for example, may define themselves positively as 'survivors' or 'battlers', as part of a community that has chosen to ignore experts' warnings and continue to live in these areas (as exemplified by the women from Thredbo referred to at the beginning of this chapter). Concerns about risks often generate temporary political alliances between people, united by their anxieties and their desire to fight against the agencies they see as imposing the risk upon them. So too, those who voluntarily face risks, such as mountain climbers and parachuters (or even, these days, cigarette smokers), may see the shared challenge as a unifying force (see Chapter 8). It is important to note, however, that rather than remaining static, risk positions are often constantly shifting and changing in response to changes in personal experience, local knowledge networks and expert knowledges.

SOCIAL STRUCTURES AND POWER RELATIONS

The approach to risk behaviours which assumes rational calculation, the weighing up of costs and benefits, also tends to ignore the role played by power relations. Individuals are represented as agential, operating in fields of social relations in which they are able to move between different logics of risk at will. Some critics have responded to Beck and Giddens' writings on risk by asking whether reflexivity is an important aspect of everyday subjectivity for most people in late modern societies. They have drawn attention, in particular, to what they consider to be the continuing influence of social class, gender, ethnicity, position in the life course and so on in shaping subjectivity and individuals' life chances. While these broader

structuring factors may have weakened somewhat in their influence in the contemporary era, particularly in the move towards individualization, it is argued that they are still important.

The designation of the label 'at risk' often serves to reinforce the marginalized or powerless status of individuals. Certain social groups have tended to be singled out as 'at risk' of a constellation of harms—children and young people, members of the working class, pregnant women, people who use illicit drugs, the elderly, sex workers, the homeless, the mentally ill. The 'at risk' label tends either to position members of these social groups as particularly vulnerable, passive, powerless or weak, or as particularly dangerous to themselves or others. In both cases, special attention is directed at these social groups, positioning them in a network of surveillance, monitoring and intervention.

The self-reflexive individual, as presented by Beck and Giddens, is a socially and economically privileged person who has the cultural and material resources to engage in self-inspection. Many people, however, simply lack the resources and techniques with which to engage in the project of self-reflexivity. Lash (1994; Lash and Urry 1994) sees Beck's reflexive modernization thesis as having at its core the assumption that agency is progressively freeing from structure, that people are increasingly able to define their own lives. He questions this, however, claiming that rather than structuring factors disappearing entirely, new ones are generated in late modernity. The important social structures of early modernity—such as the family, the welfare state, trade unions, government bureaucracy and social class—have been largely replaced by information and communication structures.

Lash argues, therefore, like the 'governmentality' writers, that: 'The risk society is thus not so much about the distribution of "bads" or dangers as about a mode of conduct centred on risk' (1994:141). As such, there might be said to be a distinction between 'reflexivity winners'

and 'reflexivity losers'. The former are those who are equipped through access to social resources such as education to engage in self-reflexivity, while the latter are those people who are unemployed or otherwise socially under-privileged. As Lash asks; 'Just how "reflexive" is it possible for a single mother in an urban ghetto to be?... Just how much freedom from the "necessity" of "structure" and structural poverty does this ghetto mother have to self-construct her own "life narratives"?' (1994: 120). He argues that access to and place in the new modes of information and communication structures is now more important to people's life chances than is their access to productive capital. Those who are not equipped to acquire new forms of information and access to knowledge flows, or who are excluded from acquisition, are those who are 'reflexivity losers'.

In an analysis of the life chances of young people in contemporary Britain, Furlong and Cartmel (1997) point to what they describe as an over-emphasis in policy and mass media discourses on the extent to which individuals are free to determine their future. Over the past two decades, individualization has increased dramatically in relation to how young people's educational and work experiences are conceptualized. Young people now, compared with those twenty or thirty years ago, are faced with a greater range of uncertainties and choices to make about how to conduct their lives. This emphasis on individualization is an outcome of such broader social changes as a transformation in the economic order, the extension of education and the associated demand for credentials, a growing commodification of education, increasing youth unemployment and the emergence of 'flexible' employment practices. As in the past, however, class, gender and ethnicity still strongly influence the opportunities available to young people in relation to education and employment. Young people from workingclass backgrounds, for example, still have less access to elite tertiary education and professional careers than do

those from middle-class backgrounds. Furlong and Cartmel also, therefore, question the emphasis that Beck and Giddens place on individuals' capacity to shape their life trajectories.

The area of occupational risk has received attention from sociologists interested in the social context of risk. The conclusions drawn by researchers into workers' knowledge and perception of risk in the workplace, and their ability to act upon such knowledge, provide insights into both the structural and cultural nature of risk. One study carried out in the United States into the perceptions of risk held by individuals who work with dangerous chemicals focused upon the meaning and social context of risk perceptions, with particular emphasis on importance of subjective experience and the connections between such experience and behaviour (Nelkin and Brown 1984). The workers interviewed conveyed feelings of psychological and physical isolation from co-workers, management, government and even their families. Workers felt frustrated by company doctors who dismissed their health complaints, and with supervisors who were more interested in maintaining productivity than monitoring safety levels. The unions to which workers belonged had limited resources to deal with occupational hazards, and often concentrated on other priorities. Workers thus tended to individualize the health problems and risks they were routinely exposed to at work, and to blame themselves for becoming ill or injured.

In terms of individuals' perceptions of risk, Nelkin and Brown found that the attitudes workers held towards their jobs in general tended to colour their attitudes towards the risks involved. Those who enjoyed their work tended to minimize the significance of its risks. Other workers, particularly those engaged in low-paid, physically demanding and monotonous tasks, although well aware of the risks, indicated that they had no choice. Lack of opportunity in the job market and the existence of family responsibilities meant remaining in the job and ignoring

the risks as a matter of economic necessity. Few of these workers were willing to risk changing jobs or even risk speaking out about their fears relating to workplace conditions in case they would be fired. As this study demonstrates, such features as lack of autonomy in the workplace and anxiety about losing one's job have an important influence both on how risk is constructed and how it is dealt with.

Some of the most interesting work on the power relations and social contexts in which risk is understood and acted upon has been undertaken in relation to the activities of marginalized groups such as sex workers and gay men. Bloor (1995:93) gives the example of female and male prostitutes working on the streets, who often lack the power to enforce safer sex practices upon their clients, even though they are highly aware of the risk of contracting HIV/AIDS and feel susceptible to infection. Hart and Boulton (1995) have further pointed out that sex workers like 'rent boys', who lack permanent accommodation and are ill-paid for their work, may often choose temporary comforts such as a warm place to sleep and a hot meal that unprotected sex with a punter may afford them over the more distant risk of contracting HIV. Access to material resources, therefore, may be a central feature of risk behaviours. Whittaker and Hart (1996) note in a study of female sex workers who work from flats (rather than on the streets) the importance of social context and the organization and conditions of work for these individuals' risk practices. They emphasize how the women's social status, their position as a marginalized group given little protection by the police or other authorities, means that they must develop strategies amongst themselves to deal with the risks of physical violence, robbery and abuse from their clients. This includes working with other women who act gatekeepers, helping to maintain control over the interaction by monitoring the clients and the time they spend with the women.

Such studies reveal that people's social location and their access to material resources are integral to the ways in which they con-ceptualize and deal with risk. Rather than responding as autonomous agents to the risks they perceive, people act as members of social groups and social networks. Their membership of these groups and networks may well be more dynamic than was the case in earlier eras, but it is still influential in mediating the capacities of individuals to act as reflexive subjects in relation to risk.

AESTHETIC AND HABITUAL DIMENSIONS

In a further critique of Beck's and Giddens' model of reflexivity in modernity, it has been argued that their approach leaves aside the cultural and aesthetic aspects of judgement: Their idea of the subject is an entity that reflexively controls bodies rather than as something which itself is bodily' (Lash and Urry 1994:32). It has been contended that reflexivity should be understood not simply as a process of rationalist self-monitoring through cognitive or normative categories. Reflexivity may also incorporate self-interpretation and interpretation of social processes, conducted through aesthetic and hermeneutic understandings, those that seek to understand the deep meaning and significance of actions, words, deeds and institutions (Lash 1993).

Aesthetic or hermeneutic reflexivity is embodied in such aspects of life as taste and style, sense of time and space, consumption, leisure and popular culture and membership of subcultural groups. It is rooted in background assumptions and unarticulated practices and in intuition, feeling, emotion and the spiritual. This type of reflexivity involves the processing of signs and symbols rather than simply 'information'. It is a product of an individual's embodied 'being-in-the-world', in which knowledge about the world is developed through—and not just in relation to

—the body (Merleau-Ponty 1962). Aesthetic reflexivity relies upon an individual's membership of a community, the moral and culturally learned and shared assumptions, preferences and categories to which Douglas refers in her work on risk (Chapter 3). Aesthetic and hermeneutic reflexivity is not rooted in self-monitoring, but rather in self-interpretation, involving intuition and the imagination above moral and cognitive judgement. It pre-exists the development of moral and cognitive judgements, and is based in bodily predispositions that are acculturated from individuals' entry into society (Lash 1993:9-10).

An example of the hermeneutic dimension of risk responses is the ways in which many people define others as 'risky', as posing a threat to them in some way. Many studies have shown that people tend to deal with the risk of HIV/AIDS and other sexually transmissible diseases by constructing categories of riskiness based not necessarily or only on expert definitions but on their culturally acquired understandings of purity and danger. People who are thought to be 'clean'—based on judgements of such attributes as their social class, ethnicity, sexual preference, how long one has known them, their manner and physical appearance—are treated as less risky, while those who are symbolically 'dirty'—again based on consideration of the attributes listed above—are designated and treated as posing a threat to oneself, as contaminating. A study of young Canadians, for example, found that they tended to categorize potential sexual partners as 'risky' or otherwise based on consideration of how well known they were to the interviewee, where and how they first met them, whether they were judged to be 'sleazy' or not and their appearance. The primary mode of selection was choosing someone who was not the 'wrong kind of person'. If an individual was defined as 'not risky', based on these considerations, they were considered not to be likely to carry sexually transmissible disease and were therefore trusted as safe to have sex with (Maticka-Tyndale 1992). The assumptions on which these judgements of 'riskiness'

are based are founded in binary oppositions, stereotypes and other systems of meaning that individuals begin to learn from childhood as part of their acculturation into society. (See Chapter 7 for further discussion of risk and Otherness.)

Lash (1993) further contends that contradiction and contingency are more characteristic of contemporary reflexive subjectivity than is allowed by Beck and Giddens. While both acknowledge the contingency inherent in 'risk society', particularly in relation to the incalculability of risk consequences and the changing state of expert knowledges, their focus on reflexivity tends to suggest that certainty is possible. Compulsive self-monitoring is not consistent with uncertainty and ambivalence, for these allow no guidelines by which self-monitoring can take place. Indeed, 'the very notion of "risk" entails making calculable the incalculable or the monitoring of contingency' (Lash 1993:6). People often feel, however, that knowledges about risk, including their own, are so precarious and contingent that they simply do not know what course of action to take. As a result, they may move between different risk positions at different times, sometimes attempting to control risk, at other times preferring a fatalistic approach that simply accepts the possibility of risk without attempting to avoid it.

The model of the rational actor, constantly calculating her or his exposure to a myriad of risks and then taking steps to diminish these risks, also tends to suggest that risk avoidance invariably takes place on a conscious level. This model assumes processes of human reasoning and behaviour that are based on calculation and economic rationality, with individuals represented as carefully weighing up the costs and benefits of various actions and making decisions based on these assessments (Bloor 1995: 92). Risk responses, however, need not necessarily take place on a conscious or 'rational' level. A distinction may be made between reflexive actions versus conventional or habitual conduct, which does not require conscious

problematization. Such actions are products of acculturation which do not involve 'inner ethical problematization' but rather are acquired as 'a brute outcome of [individuals'] habitus or mode of life' (Hunter 1993: 165). Risk-related practices, therefore, may include both activities that may need high levels of problematization, the seeking out of advice or self-interrogation (should I do this or not?) but also those practices (perhaps better entitled 'habits') that do not involve such deliberation, but rather are experienced as 'second nature' to us.

Bourdieu's (1984) notion of the habitus is useful here to explore the habitual, acculturated nature of risk-related actions. The habitus is a set of dispositions and bodily techniques, modes of behaving and deporting oneself, that is passed on from generation to generation and is linked to membership of subcultural groups. These sets dispositions and techniques are organizing principles by which sociocultural practices are maintained and reproduced. Style of dress, accent, way of walking, type of food eaten, the way one decorates one's home, the ways one spends one's leisure time, are all part of an individual's habitus, which are strongly influenced through notions of taste, style, pleasure and aesthetics. Most of the dispositions and behaviours that comprise an individual's habitus operate as habits, at the sub-conscious level. Indeed, the very difficulty in changing one's habits is founded on their almost automatic nature, their taken-forgranted participation in our mode of being and embodiment each day.

This perspective sees some aspects of risk avoidance as a part of everyday life that is often habitual, barely thought about because it has been adopted as part of people's everyday routines. Thus, for example, buckling up one's seat-belt each time one enters a car may well be a matter of habit for many people rather than a response to a feeling of concern or anxiety that is consciously recognized and dealt with on each occasion. So too, what may be labelled

as 'risk-taking' activities in some discourses may instead be viewed by those who engage in them as part of everyday life and not as 'risks'. For instance, people who inject heroin are portrayed in public health discourses (and other sites, such as the mass media) as having lives characterized by the risk of injecting the drug, particularly in relation to contracting HIV and hepatitis. These 'risks' (as they were identified by others) may not be the only risk in these people's lives, or indeed, the most important risk from their perspective or even conceptualized as risk by them. In researching the everyday lived experiences of people using heroin in London, Rhodes observes that what is considered to be risk-taking behaviour from the perspective of public health tends to be merely routine to the users: 'the act of injection, for established injectors, is situated within the world of the everyday. It is mundane and unspectacular. Preparing for an injection may become something akin to making breakfast or having a drink' (Rhodes 1995:140). Here again, the disparities in how different groups of people conceptualize risk are evident.

CONCLUDING COMMENTS

This chapter has developed a viewpoint on lay actors' responses to risk that recognizes the importance of reflexivity in relation to risk in the context of late modernity. However, it goes beyond what is seen to be a limited notion of reflexivity as represented in the work of such writers as Gxiddens and Beck in acknowledging that reflexivity involves not simply rational, individualistic assessment of expert knowledges but also the development of knowledge based on everyday experience and relationships with other lay actors. Further, reflexivity is not simply based on cognitive judgements but is also founded in aesthetic or hermeneutic judgements that are developed through acculturation. People may respond habitually to risk, which means that they do not consciously weigh up risks and benefits but rather include

risk-avoidance as part of the habits of their everyday lives. Risk responses are also developed via people's membership of social groups and networks, their access to material resources and their location within the life course and relations of power. These are aspects which shape the capacity of individuals to deport themselves as the ideal autonomous citizen that is expected of them in risk-related discourses.

The importance of acknowledging the shared and symbolic meanings of risk has again been highlighted in this chapter. The next chapter addresses these aspects of risk in greater detail, by exploring the ways in which notions of Otherness constantly emerge in individuals' and social groups' responses to risk.

7 RISK AND OTHERNESS

As I observed in earlier chapters, many theoretical discussions of risk tend not to acknowledge the differentiation of the targeting and effects of risk discourses on specific social groups. They represent the risk actor, in many cases, as lacking a gender, age, ethnicity, social class or sexual identity. Statements about risk and subjectivity tend to elide differences, presenting the risk subject as universal. Beck claims that in risk society, the globalized and therefore 'democratic' nature of risks is such that the propensity to identify the Other as the source of danger recedes: 'The "end of the Other", the end of all our carefully cultivated opportunities for distancing ourselves, is what we have become able to experience with the advent of nuclear and chemical contamination' (Beck 1992a:109). A close examination of the ways in which risk discourses operate as strategies of normalization, of exclusion and inclusion, however, demonstrates that this is not the case. Notions of Otherness remain central to ways of thinking and acting about risk.

The 'cultural/symbolic' approach, in its focus on risk and Otherness, goes some way to emphasize that anxieties and fears about risk tend to be projected onto certain social groups: those that are defined as the marginalized and stigmatized 'risky' Other. As Mary Douglas' writings have shown, the Other—that which is conceptualized as different from self—is the subject of anxiety and concern,

particularly if it threatens to blur boundaries, to overtake the self. These anxieties and fears tend to emerge from and cohere around the body, which itself is a highly potent symbolic object. This chapter begins, therefore, with an analysis of the links between dominant notions of the body and concerns about risk and Otherness. Then follows accounts of the ambivalence produced by hybridity and liminality and the psychoanalytic dimension of responses to Otherness, particularly in relation to the abject. The chapter ends with a discussion of spatiality as it relates to questions of risk and the Other.

EMBODIMENT

Many of the 'risks' identified in past eras and in the present are seen to directly threaten the integrity or health of one's body. As Douglas' work has shown, notions of risk often use the fleshly body as a symbol for the body politic. Symbolic meanings, however, are also directly applied to material bodily practices. Across historical eras and sociocultural contexts, a range of bodily practices that have been established to deal with risk may be identified. Changing notions of the body over the centuries in western societies have been accompanied by changes in the ways in which risk has been conceptualized and dealt with. In particular, the notion of the 'open' body that dominated in antiquity and the medieval era in Europe has gradually been replaced by that of the ideal of the 'closed' body, emergent in early modernity and becoming dominant in the contemporary era. Each model of the body has implications for risk epistemologies.

In the pre-modern era, the body was thought of as a micro-cosm, an open world which embraced forces that gushed forth through the orifices of the lower body. The 'higher' regions of the body (the head/mind/soul) were privileged as more noble than the 'lower' regions, which were considered to be the sites of generation and decay. The medieval body, however, remained a largely

uncontrolled, sensuous and volatile body, open to the world. The body was seen as part of a communal whole, freely interacting with others' bodies. Since the body was thought of as open, it was conceptualized as constantly susceptible to invasion. While there were many fears and uncertainties relating to the body, these were linked largely with magic and the supernatural, those phenomena thought of as beyond human control such as demons and evil spirits (Elias 1939/1994; Bakhtin 1984; Muchembled 1985; Ferguson 1997). To deal with the threat of invasion of the body by such sinister forces, causing illness and death, a range of taboos and protective rites were established to police the body's openings and protect it (Muchembled 1985; Ferguson 1997). particularly potent, seen as bearing the essence of a person, and as a conduit between inside and outside the body. It was thought, for example, that urinating against the same wall upon which a leper had also urinated could result in contracting leprosy oneself (Muchembled 1985:72). The open body, however, was also celebrated, particularly in carnivalesque rituals, in which the 'grotesque' nature of the open body, its excessive physicality, was a source of revelry and pleasure (Bakhtin 1984; Ferguson 1997).

The emergence of the ideal of the 'closed' body in early modernity led to a change in ways of thinking about contamination and purity. A division between the fleshly body and the non-material spirit emerged, with the body more negatively identified with the 'flesh' 'as a helpless and inherently corruptible creature', thus requiring continual denial and discipline by the will (Ferguson 1997; Mellor and Shilling 1997). The inside of the body was conceptualized as disorderly, dangerous and ugly, compared with the outside of the body. The ideal body was represented as noble, the key to human progress through self contemplation and the insights of the senses. This body was autonomous, individuated and closed off from other bodies. In the early modern era, the body, which was previously subject to external regulation,

became viewed as principally self-regulated through rational thought and action. The ideal of self-discipline of the body became privileged in relation to what went in and out of the body and how the body was deported. This was accompanied by a network of regulations concerning the management and control of the body (Elias 1939/1994).

The progressive change from the open, 'grotesque' body to the closed or 'civilized' body (Elias 1939/1994) resulted in the intensification of anxieties about the orifices of the body and what flows in and out of it. When the body was conceptualized as open to the world, as inevitably porous and only weakly subject to the control of the individual, pleasure as well as fear accompanied the flow of forces in and out of the body. The increasing emphasis on selfregulation, the closing off of the body as much as possible, resulted in greater anxiety about the possibility of loss of self-control and the blurring of boundaries between inside and outside, and self and Other. By the nineteenth century, the 'grotesque' body—that which was seen to be unable to control its bodily boundaries—was greeted with disgust and horror, particularly on the part of the ruling and bourgeois classes (Stallybrass and White 1986).

In the late modern era, particularly in Anglophone and northern European societies, it is considered extremely important in most cases to maintain a distinction from others' bodies, to tightly shut one's body off. With an intense focus on the social importance of maintaining and presenting a 'civilized' body, there is evidence of a high level of concern and anxiety around manifold issues to do with body boundaries, including the fluids that flow in and out of the body, the ways in which others touch one's body and the deportment and appearance of the body. Apart from a very small number of intimates, those who are seen to get too close to one's body are perceived as dangerous, alarming and threatening (Miller 1997:50).

Parts of one's own body, such as the genitals and anus, also evoke fear, anxiety and disgust because they are culturally coded as contaminating and dirty, and therefore

as highly risky to touch, by oneself or others. As a result, 'We can become the other to ourselves and engage in various forms of disgusting behaviour that are understood as violations of the self' (Miller 1997:51). What we find disgusting is that which is seen to impose a threat of some sort to our bodily or self integrity. Central to our feeling of disgust is our sense that boundaries have been transgressed. Such substances as chewed food that has been spat out and vomit are found to be revolting and contaminating because they flout the rule that 'what goes in to the body should not come out again'. Substances that originate from within the body and emit from bodily orifices (the nose, the eyes, the genitals, the anus, the mouth) are risky and potentially defiling to oneself and others when they escape from the proper place—inside the body-to outside the body and we are made aware of their existence (Miller 1997:97).

Except in very special cases (such as consensual sexual intercourse, medical encounters and carers dealing with the bodily fluids of infants and young children), contact with bodily substances deemed to be contaminating or having one's bodily orifices penetrated by the bodily parts or bodily substances of others tends to invoke a host of strong emotions, including fear, anger, revulsion, horror, disgust and anxiety. These emotions may be produced not only in relation to one's own body but also to those bodies with whom one has an intimate relationship. This was graphically illustrated by a highly publicized case in New Zealand in 1996, in which a woman who breastfed another woman's eight-month old infant while she was being cared for in a child-care centre was accused by the mother of risking her child's health. The other woman argued that the child was inconsolable with hunger and the mother could not be found to breastfeed her. The ostensible cause of this risk, argued the mother, was that the other woman could have carried HIV or hepatitis in her breast milk which may have been passed on to the child. She demanded that the woman undergo blood tests

for these infections, considered laying an assault charge against her and took the matter to New Zealand's Commission for Children as well as the Human Rights Commission. She argued that the woman 'didn't care about my baby's health when she did it. How did she positively know that she didn't have any infections at the time?' (quoted in Crawshaw 1997:46).

At the bottom of the mother's outrage, however, was her sense that her intimate relationship with her child, developed partly through breastfeeding her, had somehow been destroyed by the other woman's act. The mother tellingly compared the incident with 'finding your partner in bed with someone else' and went on to assert that 'Breastfeeding isn't just about feeding your baby. It's an intimate, dynamic relationship I don't wish to share with a stranger'. She also noted that her main concern was that 'a stranger put her nipple in my child's mouth' (quoted in Crawshaw 1997:45-6). The idea that another woman's bodily part—the nipple—and bodily fluid—breast milk had entered her child's body via the process breastfeeding was distressing for this woman. transgressed ideas about the relationship between mothers and their infants and the role played by breast milk in this as well as norms about whose bodily parts and fluids should interact with one's infant. At risk was the integrity and purity of the child's body and her relationship with her mother.

Kroker and Kroker (1988) use the term 'panic bodies' to describe the emotions that people feel about their bodies and the threat of penetration by a range of phenomena that are currently deemed to be malign, such as viruses and bacteria, pollutants, food, drugs and other people's bodily fluids. They argue that diffuse anxieties individuals feel about the apparent breakdown in postmodern society are worked out on their bodies, in their intense concerns about protecting their body boundaries from invasion and dissipation. Kroker and Kroker describe this phenomenon 'Body McCarthyism', 'which insists

(unattainable) ideal of absolute purity of the body's circulatory exchanges as the new gold standard of an immunological politics' (1988:11). In this new politics, hygienic standards come to stand for ways of identifying self and Other, with the Other standing as the contaminated, polluting threat to the purity of self.

HYBRIDITY AND LIMINALITY

Otherness, as discussed in Chapter 3, is a product of observations of difference/strangeness. Otherness is dangerous because it confounds order and control: 'Encountering the other is to expose oneself to the abyss of difference...difference which attracts us precisely because of the richness it contains, but which is also fraught with risk and instant danger' (Melucci 1996:101). The Other represents the unknown and the threat of loss of one's own identity through contact with this unknown, the dissipation of boundaries and the realization of our own limits: 'encountering another always entails putting into question something of ourselves and of our uniqueness and venturing into an unknown land only to discover what we lack' (1996:101).

The use of dichotomies as a separation practice is central to ordering. The Other or opposite is the 'degraded, suppressed, exiled' side of the first member of the opposition (Bauman 1991: 14). Bauman gives the example of 'nature', positioned as Other to science and to humanity in post-Enlightenment thought. Modern science 'was born out of the overwhelming ambition to conquer Nature and subordinate it to human needs' (ibid.: 39). Nature was designated as anything that spoilt order, harmony or design and thus refused purpose or meaning: rivers that flow in the wrong direction, plants that seed themselves where they spoil the harmony of arrangements of flora, animals that are not useful to humans. Bauman identifies this thinking about science and nature in Nazi ideas about the importance of 'racial hygiene', underlying the impetus

of the Holocaust. The Jews were represented in Nazi writings as dirty, unhygienic, pathological, contaminating, literally as 'vermin' and therefore requiring 'extermination' in the interests of cleanliness and hygiene:

Defining the Other as vermin harnesses the deeply entrenched fears, revulsion and disgust in the service of extermination. But also, and more seminally, it places the Other at an enormous mental distance at which moral rights are no longer visible. Having been stripped of humanity and redefined as vermin, the Other is no more an object of moral evaluation.

(Bauman 1991:48)

As discussed above, the ideal notion of the human body in contemporary western societies is that which is tight, contained, exercizing full control over its boundaries and what comes inside and goes outside. At its most extreme, this ideal seeks to disallow the very existence of the material body, seeking the perfection and purity of rational thought over the impurities of fleshly desires and needs. The white, able-bodied, bourgeois, heterosexual masculine body is valued as most closely conforming to this idea of the contained, 'civilized' body, while the bodies of women, the working class, non-whites, the disabled and gay men are set apart as the Other, for they are represented as incapable of fully achieving the mastery of the body. Such bodies are culturally represented as subject to the will of the flesh rather than that of reason, prone to emotionality, excessive desire, violence or disarray.

Portraying certain social groups as more animal than human has served to represent them as the dangerous Other because of their supposed lack of humanity. Non-white peoples have frequently been portrayed as animalistic by bourgeois northern Europeans, as have such minority groups as the working class, the Gypsies, the Irish and the Jews (Sibley 1995:26–8). In the history of European writings on race, particularly in the colonial era,

black people have been constantly singled out as Other: as uncivilized, uncontrolled, irrational, dirty, diseased and therefore threatening to white people and their purity (ibid.: 19–24). Nineteenth-century British colonialist and medical discourses in South Africa, for example, portrayed black Africans as inherently dirty and diseased, as 'savage' rather than 'civilized', as both morally and physically degenerate. In describing black Africans as 'dirty' and 'greasy', the British portrayed the black body as porous, odorous and damp, and therefore as potentially contaminating to those who came into contact with it, in stark contrast to the ideal of the white European as clean, contained and controlled (Comaroff 1993).

Otherness does not only involve that which is placed directly in opposition to the self/us, as part of a binary opposition, but also that which is uncertain, confusing and blurs the ordering of binary oppositions—the hybrid and the liminal. As Douglas argued (Chapter 3), that which is seen to be anomalous, difficult to classify, creates feelings of unease and repulsion. Bauman (1991) has also written about the 'acute discomfort' and anxiety we feel about ambivalence, when we are able to assign an object or event to more than one category. As he notes: 'To classify... is to give the world a structure: to manipulate its probabilities; to make some events more likely than some others; to behave as if events were not random, or to limit or eliminate randomness of events' (ibid.: 1, original emphasis). In classifying, we perform acts of inclusion and exclusion. Ordering is the central task of modernity, an attempt to fend off chaos. The imperative of reason in modernity has led to intolerance of things that cannot readily be ordered and categorized: 'The other of order is the miasma of the indeterminate and unpredictable. The other is the uncertainty, that source and archetype of all fear' (ibid.: 7).

The hybrid is that which combines two types thought of as distinct from each other in such a way as to merge their characteristics into a new type, or the separation of a

single entity into two or more parts, rendering each different from the other: 'Hybridity thus makes difference into sameness, and sameness into difference, but in a way that makes the same no longer the same, the different no longer simply different' (Young 1995:26). It has been argued by some commentators that the globalizing processes of late modernity are producing new and transitional forms of hybrid identities. These identities confound, confuse and challenge established ideas about the distinctions between different types of cultural identity. Hybrid identities are constantly in flux, remaking boundaries rather than bolstering them. As a result, 'difference is neither One nor the Other, but something else besides, in-between' (Bhabha 1994:219, original emphasis). The fluid, contingent nature of identities, the requirement that they must be actively performed, means that they are never finished, complete. It is in the interstices of binary oppositions and definitions of identity that hybridity is generated and flourishes. Hybridity is always risky, however, because of its unbounded nature, its defiance of taken-for-granted categories.

Hybridity has long been the cause of great concern and anxiety among privileged groups who are threatened by the idea of their gene stock or 'blood', for example, being adulterated and contaminated by mixing with other races. Such a mixing was believed to produce a hybrid stock that was weak and degenerate, less civilized than the Aryan race. This anxiety was particularly dominant in the nineteenth century, a time in which colonialism involved the movement of Europeans into other continents peopled by races deemed by them to be inferior and animalistic. At the heart of racial theory espoused in the Victorian era in England, for example, was evidence of an obsession and fascination with issues of sexuality between members of different races, fertility, miscegenation and inter-racial transgression (Young 1995). Young refers to the 'voyeuristic tableau of frenzied, interminable copulation, of couplings, fusing, coalescence...this steamy model of

mixture' (ibid.: 181) between races that appeared again and again in the nineteenth-century literature of racial theories.

In a history of Vancouver's Chinatown area, Anderson (1996) documents how this space was set apart in mainstream discourses in the nineteenth and early twentieth centuries as a site of vice, disease, danger and corruption because of the presence of Chinese people. Like the Orient of western imagining described by Edward Said in his influential Orientalism (1978), Vancouver's Chinatown served as the place of the Other for the majority white population, helping it forge its own understanding of identity and cultural boundaries. There was a particular concern on the part of bourgeois whites about the sexual interaction of Chinese and white people, not only in relation to sexually transmitted diseases that the Chinese were believed to harbour but also to the threat of contamination of the white population of Canada through the production of mixed race 'stock' or the 'mongrelization' of white 'purity'. In visual and verbal representations, the Chinese were portrayed as flooding into Canada, as an 'influx' threatening to sweep away the dominance of whites and overcome them through sheer force of numbers. Canadian white family life was set up in opposition to and at risk from the corruption of the Chinese. White women in particular were positioned as being at risk from the sexual desires of the lascivious, exotic Oriental man.

The liminal is that which represents a transitional, middle stage between two distinctly different entities, identities or sites. It thus cannot be categorized into either: it is 'in-between'. In contemporary western societies, bodies that are seen to transgress or blur culturally important boundaries are the source of confusion, fear, anxiety and even hatred, revulsion and disgust. The blurring of the binary opposition between male and female, for example, inspires strong emotions. Such figures as gay men and lesbians, transsexuals and transvestites,

highly-muscled female bodybuilders, men in drag and those rare cases of individuals who are born with ambiguous genitalia are continually subjected to negative reactions because of the challenge they pose to the takenfor-granted major distinctions between male and female identity.

Liminal figures, such as the stranger who crosses borders between 'inside' and 'outside' a social group, are commonly treated as threats to social order. Bauman identifies the figure of 'the stranger' as being a central source of uncertainty in late modern societies. The stranger cannot yet be categorized as either friend or enemy and is therefore disorderly, blurring boundaries and division. The ambivalence created by strangers' liminal status, their monstrous status as neither-nor, creates uncertainty, which at best is discomforting and at worst carries a sense of danger. 'Undecidables' such as strangers paralyze knowledge and action: 'They bring the outside into the inside, and poison the comfort of order with suspicion of chaos' (Bauman 1991:56). Bauman takes up Douglas's discussion of the horror produced by the slimy (see Chapter 3) in using the term 'sliminess' to describe the indeterminate nature of neither-nor figures such as strangers, which blur 'a boundary line vital to the construction of a particular social order or a particular life-world' (ibid.: 61). He notes that the 'stranger' and other marginal figures are dealt with in two major ways by societies. One strategy is 'anthropophagic', involving 'annihilating strangers by devouring them and then metabolically transforming them into tissue indistinguishable from one's own'. The other strategy is 'anthropoemic', involving 'vomiting the strangers, banishing them from the limits of the orderly world and barring them from all communication with those inside' (Bauman 1995:2).

An analysis of American neighbourhoods and how certain individuals and social groups are defined as 'Other' points to the ways in which exclusionary categories are maintained and reproduced as part of everyday life (Perin 1988). The kinds of lines or boundaries that are routinely drawn in this sociocultural context include those

between business and pleasure, home and work, relatives and friends, between the things we do for love and those we do for money, between what is private and public, between good and evil...and, as well, those between family and community, between the genders, between the species, between adults and children, between the races, between body and spirit, even between city and suburb.

(Perin 1988:4)

It is when 'lines do not hold', when ambiguities and subsequent confusion appear to flourish, the 'most socially divisive, destructive, and estranging impulses can be revealed' (ibid.: 4).

In a similar analysis, Mossman (1997) has pointed out how encounters with homeless people, increasingly a common experience for Americans in the 1990s, confronts them with deeply felt anxieties about rational ordering of society and of individuals. Homeless people tend to be placed conceptually in that category usually reserved for animals. This is because the distinction between 'nature' (or the animal world) and 'culture' (the human world) rests upon such differences as animals have no permanent homes, live outdoors, roam about, void their wastes outside, do not wash, smell offensive and have little control over their actions, while humans dwell in fixed residences, void their body wastes privately and indoors, bathe and use deodorant to avoid smelling badly and exert control over their actions. In a liminal conceptual space between these categories are pet animals (who have become partly humanized through such actions as allowing them in the house and providing them with names) and homeless people. The latter are conceptualized as partly

animalistic because although they are biologically human, they do not fit the distinctions between 'animal' and 'human'. Homeless people may even be conceptualized as 'dirt', as matter out of place that requires removal so as to regain order and purity. Their liminal status means that they can be treated more harshly and as having fewer rights than other people, including confining them involuntarily to institutions.

People with a disability, because they fail to conform to standards of 'normality' as they are espoused in a culture, are also often treated as sub-human and subjected to fear, disgust, anger and abuse (Thomson 1997). The category of the 'physically disabled' is produced by legal, medical, political, cultural and literary narratives that comprise an exclusionary discourse, constructing the bodies of members of this category as insufficient and deviant. They thus become a repository for social anxieties about vulnerability, control and identity: 'the disabled figure operates as the vividly embodied, stigmatized other whose social role is to symbolically free the privileged, idealized figure of the American self from the vagaries and vulnerabilities of embodiment' (ibid.: 7). Here again Douglas' notion of dirt as anomalous and extra-ordinary may be used to argue that disabled bodies are conceptualized, like dirt, as lying outside the normative ordering system of western societies (ibid.: 33-4). Like dirt, people categorized as disabled have been dealt with through policies of policing, regulation, exclusion and avoidance in the effort to re-establish sociocultural order.

All these commentaries point to the importance of order and control in the late modern sensibility, and the emphasis that is placed on categorization and keeping things in their 'proper' conceptual place. Those things that are not easily categorized, that fail to stay in their categories, or that simply are too different from the self, tend to arouse anxieties and fears. They are culturally designated as potentially polluting and contaminating to self, and as a result, are typically dealt with using

exclusionary tactics that seek to locate them as far as possible, both symbolically and literally, from the self.

THE PSYCHODYNAMICS OF OTHERNESS

A further perspective on Otherness is articulated in psychoanalytic theory, which focuses on the projection of unconscious emotions and fantasies upon the Other as part of individuals' continuing attempts to maintain a coherent and untroubled subjectivity and to construct and maintain conceptual borders. Julia Kristeva (1982) has drawn upon both Douglas' theorizing of pollution beliefs and psychoanalytic theory, particularly that of object relations theory, to construct an argument looking at the ambivalent notion of the 'abject'. Object relations theory, as it has been presented in the work of writers such as Klein and Winnicott, examines the psychodynamics of the young child's individuation from the mother figure. It is argued that in infancy, the child experiences the body and self as joined with that of the mother—there is no distinct boundary between self and other, subject and object. As part of cognitive and emotional development, the child gradually comes to recognize that its body is separate from that of the mother. This realization is accompanied by feelings of terror, anger, insecurity, loss and grief, as well as desire to achieve oneness again with the mother's body.

Those who have adopted object-relations theory contend that the ontological state of selfhood, therefore, is a process that is constantly in tension with an individual's relationship with another: first the mother's body/self, but later other objects (people, things, emotions). Through gradual acculturation as infants into the social world, we define our selves, demarcate our boundaries of selfhood and embodiment, against that which is not us, which is Other:

The sense of border which emerges in infancy is not an innate sense but a consequence of relating to others and becoming a part of a culture. Thus, the boundary between the inner (pure) self and the outer (defiled) self, which is initially manifest in a distaste for bodily residues but then assumes a much wider cultural significance, derives from parents and other adults who are, by definition, socialized and acculturated.

(Sibley 1995:7)

Into adulthood, the feelings that accompanied our loss of the mother continue to be experienced as part of our relations with objects, which become invested with these feelings as they are split off from the self as a defence mechanism. Some objects become the repositories of very negative feelings: hate, anger, frustration, revulsion, disgust. Others become objects of our love and desire. Some are both simultaneously. That which we most fear, which we construct as the Other, is also often that which we most desire. For Kristeva (1982), this Other is the abject, the source of endless fascination as well as horror, which disturbs identity, boundaries and order, from which we continually seek to escape but yet are drawn to and inextricably linked to. Abjection, she argues, is a powerful feeling which is both symbolic and experienced as a bodily sensation. Above all, abjection is 'a revolt of the person against an external menace from which one wants to keep oneself at a distance' (ibid.: 135). Kristeva describes the emotional and bodily responses that the abject incurs loathing, spasms, vomiting, repugnance, retching—as a means of 'protection' from the Other.

The abject is viewed as dirty, filthy, contaminating, waste, that which confounds boundaries. As Kristeva notes: 'It is thus not lack of cleanliness or health that causes abjection but what disturbs identity, system, order. What does not respect borders, positions, rules. The inbetween, the ambiguous, the composite' (ibid.: 4). The abject, thus, is not fully Other because its boundaries continually threaten to merge with our own, and this is what renders it so threatening. The abject is always inherent within the self even though we seek constantly to expel it from our selves so as to render our selves pure and bounded: 'It is something rejected from which one does not part, from which one does not protect oneself as from an object. Imaginary uncanniness and real threat, it beckons to us and ends up engulfing us' (ibid.: 4). Rituals directed at maintaining boundaries between self and Other, the subject and the object, are thus attempts to ward off abjection by establishing separation, maintaining one's own body as 'clean and proper'. These goals are never fully achieved, for purity is only ever an optimum goal to aim for—it can never fully be accomplished.

The notion of the abject, as a psychodynamic phenomenon that is constructed through social and cultural processes but is experienced as a series of emotions arising from within, provides some explanation for the 'irrational' virulent feelings to which some individuals and social groups are exposed. Williamson (1989) has used a psychoanalytic approach to theorize the popular response to HIV/AIDS. She argues that the horror mode of narrative that has been routinely employed in popular and medical discourses on HIV/AIDS represents it as a 'primeval monster'. This monster is the result of the splitting off and projection of repressed anxieties and fascination about 'unacceptable' or 'perverted' sexualities. Perceptions of HIV/AIDS involve 'The oscillation between fascination with, and denial of, otherness and difference' (ibid.: 79). More than that, however, the monstrous features of HIV/AIDS lie in wait to break down of boundaries between self and Other, dissolving categories between the 'contaminated' and the 'pure'. As a result, people deemed to be 'at risk' of HIV infection, or who already are infected with the virus, have been routinely constructed as contaminated and polluting, the repository of fears and anxieties about the death, illness, chaos, lack

of control and the incipient permeability of the social and fleshly body (Crawford 1994). The notion that 'you can't tell by looking' whether a person is infected with HIV, used in several HIV/AIDS media education campaigns, represents the ultimate horror, for it implies that there is no way of distinguishing the contaminated Other from which one requires protection.

Kristeva identifies the corpse as a primary figure of abjection, because of its in-between role between humanity and organic matter: 'A decaying body, lifeless, completely turned into dejection, blurred between the inanimate and inorganic...the corpse represents fundamental pollution' (1982:4). Other figures who are routinely culturally defined as abject are those individuals who are hypocritical, who flout the moral order: the 'traitor, the liar, the criminal with a good conscience, the shameless rapist, the killer who claims he is a saviour' (ibid.: 4). For Kristeva, however, it is the figure of the woman's body that is most likely to be positioned as abject, because of the psychodynamic struggles that occur during individuation, when the child must separate its body and identity from that of the mother. This process of individuation, she argues, 'is a violent, clumsy breaking away, with the constant risk of falling back under the sway of a power as securing as it is stifling' (ibid.: 13). It can only take place by rendering the mother's body as abject, rejecting it and repelling it.

Because of this process of abjection via individuation, women in general tend to be positioned as Other, as transgressors of the proper. Women have traditionally been considered closer to 'nature' and the 'body', while men have been constantly associated with 'culture' and the 'mind'. Women have been constantly represented in western societies as less capable of physical and emotional control, supposedly ruled as they are by their hormones and reproductive organs. The pregnant woman, the postpartum woman, the menstruating woman and the sexually aroused or seductive woman are treated in a wide range of

cultures as transgressive, polluting, risky figures. Their bodies are represented as open to the world, as fluid and leaky, incapable of regulation, escaping and evading the boundaries of the clean and the proper.

Elizabeth Grosz has taken up Douglas' and Kristeva's discussion to analyze the symbolic nature of sexual bodily fluids, with a particular focus on gender implications. I have earlier referred to the symbolic contaminating nature of bodily fluids in general when they escape the body, going from 'inside' to 'outside'. The female body is conceptualized as more marginal, indeterminate, fluid, borderline and liminal than the male body, as seeping sexual fluids such as menstrual blood that are considered to be particularly 'dirty' because of their provenance (the mysterious uterus and hidden female genitals). The female body tends therefore to be viewed as more dangerous, defiling and diseased than the male body:

It is not the case that men's bodily fluids are regarded as polluting and contaminating for women in the same way or to the same extent as women's are for men. It is women and what men consider to be their inherent capacity for contagion, their draining, demanding bodily processes that have figured so strongly in cultural representations, and that have emerged so clearly as a problem for social control.

(Grosz 1994:197)

As a consequence of this cultural coding, in discourses on HIV/AIDS, for example, heterosexual women are represented as more potentially contaminating than are heterosexual men and the burden of responsibility is placed upon them for protecting both themselves and their male sexual partners from HIV infection. Women's fluid and open bodies, and particularly their sexual organs, are constructed as repositories of contamination. Because men's bodies are seen to be more contained, closed off

from the world, they are represented as less contaminating, more 'at risk' from women than as posing a risk to their partners (ibid.: 197).

To summarize, as part of the psychodynamics of constructing and maintaining boundaries of selfhood, other individuals and objects are invested with emotion, both positive and negative. Those individuals who are defined as 'dirty', as 'matter out of place' and as abject, are positioned in the unconscious as the 'bad object', marginalized beyond the boundaries of acceptable self. The emotions of fear, anxiety and loathing are projected onto them, as they become the repositories for that which members of the dominant group hate and fear within themselves.

SPATIALITY AND OTHERNESS

Knowledge and meaning, as cultural geographers emphasize, are inevitably spatially as well as socially, politically and historically situated. Spatial metaphors and binary oppositions are central in notions of self and Other. When we refer to the boundaries of the body/ society, to the distinction between inside and outside, to the marginalized or excluded, we are relying on spatial metaphors and binary oppositions. Notions of space themselves are cultural objects, constructed through social, political and historical processes. But the importance of space and place in relation to concepts of riskiness lies not simply in their value as metaphor, but in their materiality. The members of 'risky' marginalized groups are viewed by the dominant group as polluting public spaces, and they shrink from contact, physical or otherwise, with them. Strategies of exclusion directed at 'risky' individuals or subgroups are often explicitly concerned with maintaining bodies within certain geographical limits: 'The human landscape can be read as a landscape of exclusion' (Sibley 1995:ix).

In western societies there are many strategies directed at policing public spaces and attempting to remove members of threatening marginalized groups from areas designated as appropriate only for the privileged. Such spatial binary oppositions as high/low have often served to organize these strategies. Stallybrass and White (1986) have examined the high/low binary opposition as it appears and is acted upon in four symbolic domains—psychic forms, the human body, geographical space and the social order. As they note, the high/low opposition 'is a fundamental basis to mechanisms of ordering and sense-making in European cultures' (ibid.: 3). What is coded as the 'low' bears particular weight in terms of the contradictory and ambivalent emotions it evokes, including both revulsion and desire, repugnance and fascination. A recurrent pattern emerges in which the 'top' attempts to reject and eliminate the 'bottom' to preserve order and prestige. But through these processes of rejection and exclusion, the 'top' discovers a dependence on this Other and the low is symbolically reincorporated into its fantasy life:

The result is a mobile, conflictual fusion of power, fear and desire in the construction of subjectivity: a psychological dependence upon precisely those Others which are being rigorously opposed and excluded at the social level. It is for this reason that what is *socially* peripheral is so frequently *symbolically* central.

(Stallybrass and White 1986:5, original emphases)

It is not simply a matter, therefore, of the opposition between high and low being rigidly maintained as a practice of exclusion, for paradoxically exclusion breeds further incorporation: the repressed continually returns.

As was noted earlier in this chapter, for the bourgeoisie in early modern Europe the open or 'grotesque' body became disavowed, removed from the self and projected onto others. Indeed, according to Stallybrass and White

(ibid.: Chapter 3), the bourgeois public sphere became dependent on denial of the grotesque, the irrational, the bodily. In the nineteenth century the city became a focus of the bourgeoisie's anxiety about contamination and disorder, via a concern about the sewer and the slums and the figures of the prostitute, the servant, the labourer, the scavenger, the beggar and the criminal. 'As the bourgeoisie produced new forms of regulation and prohibition governing their own bodies, they wrote ever more loquaciously of the body of the Other—of the city's "scum" (ibid.: 126). The bourgeoisie's concerns about maintaining the propriety and cleanliness of the body, to disavow the presence of the 'lower' body and its functions, was mirrored in their concerns with the city and its lower strata. The working class and the poor, the people who lived and worked on the streets, were viewed as transgressing the boundary separating human from animal in their contact with dirt and disease, their degraded living conditions and their lack of bodily control. New forms of surveillance and discipline were developed in the attempt to control these contaminating bodies, including bodily regimes around cleanliness, dress, diet and deportment. Plans attempting to control the built environment of the city were developed and implemented so that working-class housing and slums would be kept apart from the eyes and noses of the bourgeoisie.

In contemporary western societies, the figure of the criminal is frequently positioned as risky and requiring exclusion from others. As part of the strategy of dealing with the risk and uncertainty of crime, people develop a 'mental map' of places, defining some as likely to be 'safe' and others as 'risky'. This 'mental map' does not simply rely on geographical aspects of a space or place, but also draws on ideas and assumptions about social relations and the kinds of people who inhabit or pass through these spaces and places at specific times of day and night. Fear of crime tends to be located within public rather than private space, as the criminal is considered to be an 'unpredictable

stranger' rather than someone known to oneself, and thus as inhabiting public space rather than being encountered in one's home. Members of such social groups as young working-class men, the unemployed and injecting drug users are typically nominated as potential criminals because of their assumed simmering resentments against society and lack of capacity for self-control. Those spaces in which they move about—the inner city, the shopping mall, the housing estate—are considered 'dangerous' in terms of the risk of crime and therefore as requiring increased surveillance, police presence and caution on the part of those who transverse them (Lupton in press).

Strategies of exclusion exerted on the part of the most power ful in a society in their attempts to avoid risk often serve to incite fear and anxiety in those they seek to exclude or intimidate. The bodies of white, heterosexual, bourgeois men tend to claim public space as a right, and frequently seek to dominate and exclude others through exerting an aggressive gaze or through violence. Other bodies must fight to establish their place in this space. Feminists have written about the ways in which women, as one of the Other categories of bodies within public spaces, are positioned as vulnerable to confrontation or attack and therefore tend to lack the self-possession of privileged men in the same space. Moving in public space, for women, is constantly problematic, making them feel uneasy or anxious, exposed to the gaze, evaluation and imminent threat of (masculine) others.

Gillian Rose, for example, has vividly described her own experiences moving about in public spaces as a woman and the heightened awareness that this gives her and other women of the relationship between space and power relations:

I have a strong sense of space as oppressive, for example, from being scared walking at night in the city in which I live. I have to tell my own fears of attack in terms of space: when I've felt threatened, space suffocatingly surrounds me with an opacity that robs me of my right to be there; I cannot look around, the details surrounding me swamp me, the innocent transparency of the empty street becomes like an aggressive plastic lens pushing on me. Space almost becomes like an enemy itself. This fear is partly about being defined as a woman.

(Rose 1993:143)

This experience of space renders entry and movement about space hazardous, full of potential dangers. If relatively privileged women such as the middle-class, white academic Rose feel vulnerable, alienated or fearful when entering public spaces, highly aware of being watched and of their movements in space, women who are even more marginalized are subject to an even greater intensification of surveillance and the implicit threat of those who watch them. Prostitutes, for example, disturb such boundaries as that distinguishing between private and public by inhabiting public spaces to invite strangers to perform acts with them that are deemed to be appropriate only in private relationships. Their presence on the streets is constantly challenged by police and other authorities, who attempt to 'clean up' the streets and make these spaces 'respectable' by removing them (Duncan 1996).

So too, gay men and lesbians are constantly challenged in relation to their right to display their affection for members of the same sex in public, including such acts as kissing or holding hands which are unproblematic for heterosexuals. Gay men and lesbians are subject to control by the legal and policing system as well as by heterosexuals, who may express their disapproval through stares, whispers or mutterings, or in more extreme ways with verbal abuse or physical violence. As a result, lesbians and gay men must constantly be on their guard in public places, policing their own behaviour and being wary of others' reactions and the imminent threat of open hostility and violence (Myslik 1996; Valentine 1996).

Strategies of spatial exclusion, therefore, are typically employed by members of dominant social groups to exert control over marginalized groups for which they hold hostility, contempt or fear of contamination. Such groups may be constructed as posing a risk to the dominant group through behaviour that is deemed to be too 'different' or potentially polluting and therefore confronting. The spaces these groups occupy are commonly singled out as dangerous and contaminating to the dominant groups. Alternatively, marginalized groups may be constructed as being vulnerable and 'at risk' from the greater power of the dominant group. For marginalized groups, constructed by dominant groups as the Other, requiring regulation or exclusion or both, this domination of space leads in turn to feelings of enhanced fear and anxiety, of being 'at risk' of intimidation, violence or coercion.

CONCLUDING COMMENTS

As this chapter has shown, ideas and strategies around risk often operate at the symbolic, conceptual level, organized around notions of self and Other. Risk beliefs and practices, as they are employed in the deportment and experience of the body, therefore go beyond the need to exert control against the threat of particular hazards (pollution, toxins, viruses or crime, for example). At the more symbolic level, the body is conceptualized as being 'at risk' when its autonomy and integrity appears to be threatened. Because the dominant ideal notion of the body is that of the body as controlled, its boundaries policed and regulated and kept separate from other bodies and the outside world, anything which appears to flout these boundaries, to break them down and allow intermingling of properly separate entities, is considered threatening, or 'risky'. Those bodies of others who are considered to lack the capacity for proper regulation of their bodily boundaries are routinely positioned as 'risky' to oneself. In western societies, it is typically members of stigmatized or

marginalized groups—women, the working class, the poor and unemployed, non-whites, injecting drug users, gays and lesbians—who are constructed as 'grotesque bodies' and therefore as 'risky' or 'at risk', needful of control, surveillance and discipline. It has also been argued, however, that constructions of Otherness also evoke feelings of fascination, excitement, desire, and that pleasure as well as anxiety is implicated in responses to the 'risky' Other. The next chapter takes up this question of risk and pleasure in greater detail.

8 RISK AND PLEASURE

Contemporary experts and popular cultures tend to represent risk as negative, something to be avoided. So too, much of the academic literature on risk represents individuals in late modernity as living in fear, constantly dogged by feelings of anxiety, vulnerability uncertainty in relation to the risks of which they are constantly made aware. As I observed in Chapter 1, risk is now often a synonym for danger or hazard, and the early modern concept of a 'good risk' appears largely to have been removed from the vernacular, appearing only in the parlance of economic speculation. So too, as was argued in previous chapters, the emphasis in contemporary western societies on the avoidance of risk is strongly associated with the ideal of the 'civilized' body, an increasing desire to take control over one's life, to rationalize and regulate the self and the body, to avoid the vicissitudes of fate. To take unnecessary risks is commonly seen as foolhardy, careless, irresponsible, and even 'deviant', evidence of an individual's ignorance or lack of ability to regulate the self.

Against these dominant discourses on risk, however, there also exists a counter discourse, in which risk-taking is represented far more positively. Against the ideal of the highly controlled 'civilized' body/self is the discourse which valorizes escape from the bonds of control and regulation, expressing a hankering after the pleasures of the 'grotesque' body, the body that is more permeable and

open to the world. This discourse rejects the ideal of the disembodied rational actor for an ideal of the self that emphasizes sensual embodiment and the visceral and emotional flights produced by encounters with danger, of 'walking on the wild side'. This chapter examines this counter discourse on risk, beginning with a discussion of escape attempts and edgework, moving on to an analysis of the gendered nature of risk-taking as performative practices, and finishing with a discussion of desire and transgression.

ESCAPE ATTEMPTS AND EDGEWORK

A recent newspaper article commented on the expansion of participation in 'extreme sports' among Australians. These sports include such activities as sky-diving, white-water rafting, swimming with sharks, abseiling, bungy-jumping, para-gliding, rock-climbing and ice-climbing. They are called 'extreme sports' because they are seen to test the limits of human endurance for fear. Their central attraction is the courting of danger, the active taking of risks for the excitement and sense of achievement that they bring with them. As the writer of the article put it, for those who decide to take part in these sports: 'The dare has been declared. Leave behind the speed limits, the smokefree zones, tight deadlines, the low-fat shopping list, the SPF-15 [sunscreen] routine and yell in the face of life' (Macken 1998). So too, adventure holidays involving feats of extreme physical endurance and bravery have become popular among some social groups. Expeditions to climb Mount Everest are now offered for people with minimal experience in mountain climbing. There is also a current trend for some travel companies to offer to the traveller wanting a 'different' and challenging holiday the opportunity to engage in 'real life' risky situations for a time, such as spending several days on the meanest streets of New York City, or accompanying an American bounty

hunter on his mission to find and capture criminals. Further, a plethora of advertisements—particularly those that are oriented towards a youth market, selling such products as sporting goods, four-wheel drive cars, alcohol or soft drinks—have used images of risk-taking, often in rugged outdoors settings, to market their products, appealing to the target audience's desire to be 'different' and out of the ordinary in their willingness to face danger and seek out thrills.

The increasing presence of these cultural products suggest a growing fascination with the pleasures and excitement of risk-taking. In their book Escape Attempts (1976/1992), Cohen and Taylor discuss a number of ways in which individuals seek to transcend the banal, routine nature of everyday life, to 'escape and resist reality'. They argue that 'escape attempts' all involve some risks, from the minor risk of offending one's partner or friends, to major threatening of life and limb. Activities such as going on holiday or taking drugs are 'interruptions in the flow of life, interludes, temporary breaks, skirmishes, glimpses of other realities' (ibid.: 45). They are means of breaking the routine, albeit temporarily, and avoiding a sense of boredom and predictability and the dissatisfaction that may accompany them: 'We talk of routines as dull and dreary; phrases like "breaking out of the routine" suggest their oppressiveness, whilst a reference to something as "just routine" decreases the significance of the activity so described' (ibid.: 48).

Giddens has similarly observed that most people have an ambivalent emotional response to their established routines and habits. Routines, rituals and habits are important to the establishment and maintenance of ontological security, fostering a sense of familiarity, permanence and certainty. When routines are shattered, anxiety, fear and hostility may be produced (Giddens 1990:98–9). However, people may also deliberately cultivate risks as a means of undermining ontological security: 'Such situations make possible the

display of daring, resourcefulness, skill and sustained endeavour, where people are only too aware of the risks involved in what they are doing, but use them to create an edge which routine circumstances lack' (Giddens 1991: 132).

Activities engaged in for the sake of novelty, however, often themselves become routine, and inevitably take place according to certain boundaries, norms and assumptions, or 'scripts' (Cohen and Taylor 1976/1992). It is perhaps only those activities that seem to counter taken-forgranted scripts, to transgress them in some way, that retain their excitement. Some risks, like the famous wild car trip taken by Hunter S. Thompson and his attorney while high on drugs between Los Angeles and Las Vegas, as recounted in his book Fear and Loathing in Las Vegas (1973), are part of adventures, requiring the active embracing of danger as part of the high: driving at high speeds while under the influence of various drugs, legal and illicit (Cohen and Taylor 1976/1992: 195-6). 'Excesses and outrages of all sorts must be built into the trip in order to transcend the limitations of ordinary landscaping, to construct a realm so far away from home that literally and metaphorically you are beyond reach.... For such trips, you must take risks' (ibid.: 196).

Lyng (1990) takes the term 'edgework' from Hunter S. Thompson's writings, and uses it to apply to participants' experiences in dangerous activities that are voluntarily undertaken as part of leisure. Edgework, he argues, takes place around cultural boundaries: such as those between life and death, consciousness and unconsciousness, sanity and insanity and an ordered sense of self and environment against a disordered sense of self and environment. Edgework is also characterized by an emphasis on skilled performance of the dangerous activity, involving the ability to maintain control over a situation that verges on complete chaos that requires, above toughness', the ability not to give in to fear (ibid.: 859). It does not involve the complete re-linquishment of control, therefore: to do so would be tantamount to attempting suicide or self-harming.

There is the belief among those who engage in edgework that 'mental toughness' is an innate ability, possessed by only a select and elite few. To take risks, therefore, is considered by such individuals not as foolhardy but rather as evidence of superior qualities that allow people to court danger without harming themselves (Lyng 1990). This attitude is often expressed by those well-known wealthy adventurers such as English entrepreneur Richard Branson, who attempt perilous and highly-publicized feats such as circling the globe in a hot-air balloon as an alternative route of excitement and achievement. It is also evident in the words of a sixteen-year-old Australian boy discussing the pleasure of 'train-surfing', or riding a moving train on its roof: 'It's so much fun. When you get off the train you feel like a superior being sometimes. You know some call it stupid but you just love it. It's stupid in other people's eyes but they just don't understand and you don't expect them to' (quoted in the Age newspaper, 6 October 1997).

The discourse of release from the overly regulated body/ self that appears in positive representations of risk-taking draws on a number of related discourses, including those that portray too tight a control over the self as a source of stress and illness and loss of self-authenticity. Participating in activities that are coded as dangerous or risky can bring an adrenalin rush that allows aficionados to escape the bounds of the rational mind and controlled body, to allow the body's sensations and emotions to overcome them for a time. There is a sense of heightened living, of being closer to nature than culture, of breaking the 'rules' that society is seen as imposing upon people. At such times, participants in such activities may attempt to experience the sublimity of losing their selves in the moment, of transcending the constraints of 'civilized' behaviour, as was championed by the Romantics in the eighteenth and early nineteenth centuries. For the Romantics, the turn towards emotion and feeling was important to avoid what they saw

as the emotional sterility of modern life in its obsession with rational order and self-mastery. Contemporary discourses of release, often draw on neo-Romantic ideals, concerning the return to the authenticity of nature, the central role of emotion in self-expression (Lupton 1998: 81–2). Ewald gives the example of the German Green movement, in which 'to live is to survive, an opportunity to rediscover in more elementary, precarious, almost savage lifestyles, the sensations, pleasures, pains, difficulties, intoxication, and roughness of a life they imagine they lost with the advent of civilization' (Ewald 1993:228).

The pleasures of risk-taking also inhere in the ways in which risk-takers may find a communal spirit with other like-minded souls. To engage in risky activities may bind people together closely in this common pursuit. particularly if they identify each other as being members of the elite group of skilled, tough-minded individuals who can cope successfully with edgework. This may lead to a state of 'collective effervescence', a Durkheimian term which refers to the intense emotionality produced through group activities. Durkheim used the term in relation to religious activities and spiritual elation. In the increasingly secular western societies, collective effervescence is also produced in other communal activities, often those that seek heightened sensual embodied experiences. Examples include taking part in a riot, mass fight or war, being a member of a violent gang, engaging in dangerous physical activities with others or drinking to excess or taking drugs with others. Through the activities evoking communal effervescence, participants may lose a sense of their autonomous selves, becoming, at least for a brief time, part of a mass of bodies/selves with a common, shared purpose. This sense of losing one's boundaries, merging with others, may be experienced as frightening, but also as intensely pleasurable in the partial relinquishing of selfcontrol and giving way to the collective will of the crowd (Maffesoli 1996; Mellor and Shilling 1997).

There is another meaning that coheres to risk-taking: that which represents it as part of the reflexive project of the self in terms of achieving personal 'growth'. This notion, which is commonly represented in self-help books, sees self-actualization in terms of a balance between opportunity and risk, choosing between an array of opportunities, some of which may be more 'risky' than others: 'The individual has to confront novel hazards as a necessary part of breaking away from established patterns of behaviour—including the risk that things could possibly get worse than they were before' (Giddens 1991:78). Risktaking, indeed, to some extent is considered vital to selfrealization and improvement. Lyng (1990:860) found that among the edgeworkers he spoke to, who engaged in parachute jumping, the notions of 'self-realization', 'selfactualization' and 'self-determination' were commonly claimed as goals of engaging in their dangerous activity: 'In the pure form of edgework, individuals experience themselves as instinctively acting entities, which leaves them with a purified and magnified sense of self' (1990: 860).

Cultivated risk-taking in this context is seen to provide an opportunity for individuals to display courage, to master fear, to prove something to themselves which allows them to live life with a sense of personal agency: 'Having survived the challenge, one feels capable of dealing with any threatening situation' (Lyng 1990:860). Giddens quotes a self-help book which argues that: 'If we reject deliberate risk-taking for self growth, we will inevitably remain trapped in our situation. Or we end up taking a risk unprepared. Either way, we have placed limits on our personal growth, have cut ourselves off from action in the service of high self-worth' (quoted in Giddens 1991:78). Some risks, therefore, are not to be avoided but rather embraced as part of the trajectory of self-actualization. To live a life that involves the avoidance of all risks, it is suggested in such literature, is to be stultified, moribund,

trapped in old habits and ways, to fail to develop as a person.

In discourses that privilege risk-taking, therefore, there is some reliance upon this quest for self-authenticity. Taking risks is sometimes seen as relinquishing the control of one's behaviour imposed by society, stepping outside and resisting expectations, being spontaneous, seizing the day. The risk-taker may be viewed as someone who possesses courage, not only in placing her or himself in danger but also in her or his deliberate contravening of societal norms. There is something extraordinary about the risk-taker. The person who attempts to sail or hot-air balloon around the world single-handed, the mountain climber scaling a dangerous peak, the daring entrepreneur who makes millions in business through taking risks, are all seen to stand out from the pack. Such risk-takers are often admired for their courage and strength of will and purpose, their willingness to face danger, to gamble with uncertainty.

A positive approach to risk-taking is also currently dominant in the sites of work and employment, particularly in relation to high-level managerial positions or those areas in which people are expected to demonstrate 'creativity', such as in marketing and advertising. Contests encouraging people to pit themselves against each other and to compete in terms of demonstrating daring are common risk environments, including institutionalized economic sector (Giddens 1991:132). The contemporary vogue for exhaustive testing of aspirants to positions using a battery of psychological and other tests often include probing them for their responses to potentially risky situations. Questionnaires ask people, for example, whether they would like to go scuba diving, or whether they ever fantasized about being a race car driver. Those who answer such questions in the affirmative are marked out as being willing to step out of conventional patterns of behaviour, to seek alternative ways of doing things—that

is, to take risks. This is valued in a professional workplace in which 'flexibility' is currently a buzz word.

In experiential training methods popular in the United States and elsewhere, workers are encouraged to engage in physical activities designed to test their will-power and fortitude and to develop a sense of 'survival' and responsiveness the rapidly changing to environment, including taking risks, tolerating fear and being innovative (Martin 1994). The experiences are designed to arouse fear and excitement to serve as models for workers' experiences in unpredictable work situations. They may include such activities as climbing and leaping off high towers and telegraph poles, walking across a high wire and abseiling down cliffs. The idea is that 'rigidity' is avoided for 'flexibility' and the ability to 'adapt' to changing conditions. Taking risks is viewed as one means of adapting to a changing environment. Risk-taking, therefore, in the work context may be considered part of an overall strategy for career advancement. The notion is that once a person has reached a certain level in their career, they may need to take a risk—to move to another position, perhaps involving a cut in salary—in order to achieve long-term goals. Risk-taking becomes the mark of someone who is ambitious and not content with remaining at a plateau in their career. The importance of personal flexibility is again implicated in this notion of risk-taking.

Risk-taking may be regarded as the flipside of modernity, a response to the ever-intensifying focus on control and predictability of modernity. On the other hand, however, the preparedness to take risks converges with some of the most basic orientations of modernity: 'The capability to disturb the fixity of things, open up new pathways, and thereby colonise a segment of a novel future, is integral to modernity's unsettling character' (Giddens 1991:133). The current insistent presence of risk may be associated with new modes of conduct and self-formation, the invention of new experiences of life and pleasures in response to the social changes emerging in late

modernity (Ewald 1993). The contemporary subject may be understood to require both routine and risk: to hanker after both predictability and unpredictability, constantly oscillating between the two. An excess of one state leads to a fervent desire for the other. Heightened awareness of risk may itself lead to a desire to take risk. Indeed, predictability itself may be viewed as a risk. Cohen and Taylor quote a slogan which argues 'We do not want a world in which the guarantee of no longer dying of hunger is exchanged for the risk of dying of boredom' (1976/1992:160).

RISK-TAKING AS GENDERED PERFORMANCES

Of all social groups, it is probably young people, and particularly young men, who most often take risks as part of their everyday lives. Drinking to excess and taking other drugs, speeding in cars, engaging in petty theft and train-surfing are ways of adding thrills to life, testing one's boundaries of fear and endurance, proving one's adulthood or masculinity. Such activities are attempts to engage in what Featherstone (1995:55) refers to as the 'heroic life', which involves deeds of virtuosity, courage, adventure, endurance and the capacity to attain distinction and a higher purpose through the risking of life itself. This higher purpose need not be altruistic, as in the archetype of the hero, but may also be a desire to achieve heights of sublimity. Transcending everyday life becomes an end and purpose in itself. Importantly, the discourse of the heroic life as it is counterposed against everyday life is profoundly gendered: 'A basic contrast, then, is that the heroic life is the sphere of danger, violence and the courting of risk whereas the everyday life is the sphere of women, reproduction and care' (ibid. 1995:59).

Interesting gender differences emerged in Green's (1997) study of children's discussions about accidents and risk-taking. She found that girls were more likely to stress the

importance of ensuring the safety of others in their care and taking the appropriate risk avoiding action in preventing accidents, while boys were more likely to deny that such care for others was their responsibility. Boys were also more likely to tell stories about accidents that emphasized the courageous and thrilling nature of risk-taking, stressing the danger involved and physical skills required. For the girls, peers who took what they considered to be careless risks were not celebrated, but considered to be silly, the object of contempt.

Similarly, in a study of adult Australians' childhood memories of transgression, fear and danger, Crawford et al. (1992) found that men tended to recount occasions in which, as children, they had tested adult authority. They remembered these incidents as being suffused with the emotions of glee and delight at breaking the rules, rather than the shame, regret and guilt that women tended to feel in relation to such experiences. Men's childhood memories of breaking adults' rules indicated the exhilaration of flirting with danger and taking risks. They knew that they were 'doing wrong' by adult standards, but their actions conformed to a different set of assumptions and logics emerging from their relationships with other children. Such experiences as being sent to the headmaster and enduring physical punishment as a child, for example, was remembered by the men as heroic, a means of demonstrating to other children one's disdain for adult rules and authority and bravery in seeking to break the rules and incur disapproval and punishment. Similar emotions were revealed in the men's memories of fear and danger in childhood, in which the extremity of the danger and the importance of the physical challenge they faced were emphasized as means of justifying their fear, as were the pride in dealing with danger and fear, and the shame in failing to cope adequately in front of others that the men felt as children.

Women's memories of such occasions, in contrast, were associated either with feelings of shame and guilt for

having behaved irresponsibly or incompetently, with having 'let down' adults, or else with anger at having been unjustly punished by adults. The men's memories revealed that as children they actively courted danger, testing limits and proving themselves in front of others, while women's memories of childhood fear tended to revolve around finding themselves in dangerous situations and then taking steps to avoid danger. There was no emphasis on the importance of 'keeping face' in front of others, as there was in the men's accounts (Crawford et al. 1992). These memories would suggest that girls are expected from an early age to be 'more adult' than boys—to act responsibly, to conform to adult rules and to be responsible for the wellbeing of others. Such behaviour is a taken-for-granted part of dominant femininity. In contrast, boys are almost expected to transgress, to misbehave and break rules and to endure punishment bravely for these transgressions: it is part of their construction and performance of masculinity.

Other analyses have identified young men as a subcultural group most likely to engage in activities that are deemed 'risky' as a means of performing dominant masculinities. Collison (1996) has discussed how young English working-class men in a young offenders prison described engaging in predatory street crime activities and taking drugs for the thrill of 'living on the edge'. He notes that for these young men, 'Edgework represents a sometimes spontaneous search for a dramatic self within a world of alienation and over-socialization...Being on the edge, or over it—beyond reason and in passion—is momentarily to grasp a spiritual and romantic utopia' (1996:435). As these comments suggest, in a sociocultural context in which dominant forms of masculinity privilege the ability to keep one's body/self separate from others, to be self-contained and autonomous, to be hard and dry, the opportunity to engage in risk-taking achieves several ends. Not only does it demonstrate courage and sometimes the enhanced capacity for self-control and containment, it may also allow men to relax the tight

control that is expected of them, if only for a short time and in limited ways. To engage in dangerous activities, for example, may demonstrate a man's control over the emotions of fear, vulnerability and anxiety, proving to others and himself the expanded limits of his control of self and the body. At the same time it affords him the opportunity to experience and enjoy heightened emotion and exhilaration. Some other occasions of risk-taking, however, may involve letting loose emotions in ways that are seen to bolster masculinity: as in fighting, for example, or drinking to excess. Such activities allow men to test and define their boundaries of selfhood and embodiment through occasional excess.

The popular media frequently draw upon and reproduce gendered notions of risk-taking in representing heroic figures. The figure of the round-the-world sailor, the mountain climber, is predominantly masculine, engaging in masculine-coded feats. The rare woman who engages in such activities is portrayed as though she were unique in her desire and ability to emulate these feats. Such a woman's achievement is seen as even more unusual because of her gender. The heroic male figure of action films—portrayed by actors such Schwarzenegger, Mel Gibson, Nicolas Cage and Bruce Willis-throws himself into danger without a moment's thought, actively seeking situations in which he risks his life and both receives and deals out brutal violence. While there are some equally courageous and fool-hardy female characters in mainstream film (such as the Ripley character played by Sigourney Weaver in the Alien series), they remain few in number compared with the male heroes. Female characters, in the main, continue to serve as the passive onlookers on masculine heroic risk-taking, often positioned as the reason why men must place themselves in danger, so as to rescue their women or return home to them.

While men may engage in risk-taking in the attempt to conform to dominant forms of masculinity, women's concepts of risk-taking are also highly related to assumptions about femininity.

As was discussed above, risk-taking is less valorized for the performance of femininity: indeed, dominant notions of femininity tend to represent the careful avoidance of danger and hazard as important. Women are acculturated from an early age to avoid situations of danger and are represented as particularly vulnerable to such risks as sexual assault and mugging because of their gender. They are more often portrayed as the passive victims of risk than as active risk-takers.

As I noted in Chapter 7, women's bodies are culturally represented as more prone to chaos and disorder compared with men's bodies. Control over the self may therefore be even more sought after by women than by men. In this context, women who find the constraints imposed by cultural notions of femininity may seek to counter these by deliberating engaging in masculine-coded risk-taking activities or other activities that allow them to 'let go' to some extent of the control that is expected of them. While the dominant masculine notion of risk-taking may revolve around placing oneself in situations courting injury or death, many women see risk-taking as related to expressing their sexuality.

For example, in her essay on the pleasures of dancing in night-clubs as a woman, Gotfrit (1991) describes the physical delight, sensualities and eroticism of dancing to loud dance music in the public realm of the night-club dance floor. She notes that the dance floor is 'a rare public place where letting go of the tight rein women often keep on their sexuality is possible, where the pleasures of the body are embraced and privileged' (ibid.: 178–9). Accompanied by the sensual pleasures of dancing is the contextualization of the night club as a sexually-charged place of 'unknown possibilities', where dancing might be seen as a preamble to flirting, romance and sex:

In the anticipation of one, or all, of those unrealized possibilities is a palatable pleasure. It is the potential offer to be taken up with, or refused to, some perhaps exotic other. It is the danger of being swept up, and away, in desire, and in the pleasures of the body. It is also the danger of letting go of control that may result in being swept away by that unknown other. This active mobilizing of desire becomes linked with notions of potential intimacy as well as risk, and pleasure becomes rooted in its own further possibilities, in the realization of fantasy.

(Gotfrit 1991:179)

In this account, therefore, central to the diffuse pleasures and sensualities of dancing are the notions of risk and danger implicated with losing control, being swept away, taking a leap into the unknown by engaging erotically with a stranger. Gotfrit goes on to describe the feeling of 'naughtiness' or a 'benign sense of mischief involved in the dancing experience, of doing things that she otherwise would avoid, such as daring to be potentially 'bad' by staying up late, participating in a sleazy nightclub, wearing short skirts and allowing herself to feel desire and be desired, of 'stepping out of "good girl" territory' (ibid.: 180). She conceptualizes her pleasure as being associated with resistance, particularly to dominant notions of appropriate femininity as asexual and contained in space. In dancing in a nightclub, women are able to privilege their bodies, to experience desire and sensual pleasure and express their sexuality, to boldly take up space and to allow their pleasure in dancing to overcome their selfconsciousness about the need to exert rigid control over their feminine embodiment.

Other women conceptualize risk-taking as challenging archetypes of feminine passivity by engaging in activities that are strongly coded as masculine. In an analysis of women who engage in competitive boxing (a much smaller group than those who enjoy dancing in nightclubs),

Hargreaves (1997) shows how the perceived risks of this enterprise are seen as part of its pleasure for the women:

They claim that facing danger and overcoming fear gives them an unbelievable buzz—they enjoy the physicality of fighting, the excitement, the roughness and the risk. For them, it is a uniquely sensuous bodily experience which, when mixed with the mental challenge, is addictive.

(ibid.: 42)

The women interviewed by Hargreaves said that they felt 'empowered' by facing the physical challenges and dangers of boxing. For them, engaging in a sport that carries with it an excess of masculine meanings was an additional challenge and source of excitement and pleasure, particularly in confronting the expectations that women should not be physical or aggressive, and are liable to be overcome with fear in situations involving fighting. They said that they enjoyed the feelings of fear and vulnerability that were part of entering the ring and facing an opponent in such a dangerous sport.

As this suggests, the dynamic and variegated nature of femininities and masculinities have implications for the gendered meanings of risk-taking. While risk-taking has been most closely linked to the performance of dominant masculinities, and risk-avoidance is associated with dominant femininities, there is evidence of some shifts in these meanings. Dominant notions linking certain risk-taking activities with masculinity have begun to be challenged by some women, who have sought to perform alternative femininities through engaging in such activities.

DESIRE AND TRANSGRESSION

I referred in Chapter 7 to the ambivalence provoked by Otherness, the mixture of fear, disgust and anxiety and fascination and desire that Otherness tends to inspire. At

the symbolic or psychodynamic level of meaning, it may be argued that the pleasures associated with risk-taking may emerge from the transgression of such conceptual boundaries: 'Crossing boundaries, from a familiar space to an alien one which is under the control of someone else, can provide anxious moments; in some circumstances it could be fatal, or it might be an exhilarating experience—the thrill of transgression' (Sibley 1995:32). While we may constantly seek to maintain conceptual boundaries, our very efforts impart a special power and significance to that which is defined as transgressive.

Sexual intercourse, for example, derives much of its excitement and pleasure for many because of its status as 'dirty' and 'forbidden', the transgressions of bodily boundaries it involves, the intermingling of usually reviled bodily fluids: 'Sex is perceived as dirty, bestial, smelly, messy, sticky, slimy, oozy, and that is precisely, for many, its attraction' (Miller 1997:127). Cohen and Taylor argue that some forms of sexuality—particularly those that are culturally represented as 'deviant', such as homosexuality or sadomasochism, or socially prohibited, such as adultery —are particularly exciting because of their association with guilt, fear or anxiety: 'the "specialness" of the experience, and therefore its potential escape status is oddly enough related to the degree to which it is accompanied by fear, guilt and anxiety' (ibid.: 129). They contend that the more such sexual activity is 'liberated' and thereby rendered more mainstream and less 'deviant', the more banal it will be become, losing its status as an extreme escape attempt.

In the final chapter of *Purity and Danger*, Douglas comments on the paradoxical nature of our yearning for the pure and our repudiation of the unclean. Purity is 'the enemy of change, of ambiguity and compromise' (ibid.: 162). When accomplished, purity is 'poor' and 'barren', it is 'hard and dead as a stone when we get it' (ibid.: 161). Notions of purity are also beset with contradictions, or enforce hypocrisy. For example, ideas of sexual purity, if

taken to their extreme, deny all sexual contact between the sexes, leading to the extinction of a culture (ibid.: 162). Cultures deal with this by rendering that which is defiling to be most sacred, allowing contact with it at certain prescribed times and places. By its very nature of lying outside, of being forbidden, that which is categorized as 'dirt', as 'polluting' is powerful: 'The danger which is risked by boundary transgression is power. Those vulnerable margins and those attacking forces which threaten to destroy good order represent the powers inhering in the cosmos. Ritual which can harness these for good is harnessing power indeed' (ibid.: 161). Indeed, religions often render sacred those things that are defined as impure. By this process the impure retains its special power, but becomes not destructive but creative. Some unclean things, but not all, are used constructively in ritual: blood in Jewish and Christian traditions, for example, is sacralized (ibid.: 159).

More secular rituals may also seek to overturn the negative meanings associated with the 'impure' by revelling in its very prohibited nature at certain prescribed times and places. Analyses of the medieval and early modern carnival as a popular cultural form in Europe demonstrates the use of the carnivalesque as a means of transgressing binary oppositions, particularly that between high and low (Bakhtin 1984; Stallybrass and White 1986; Burke 1994). The carnivalesque body provided a way of disrupting boundaries, of calling into question accepted norms of bodily deportment and thereby disturbing the social order. Carnivalesque activities included ritual spectacles such as fairs and shows, games, popular feasts and wakes, processions and competitions, comic shows, circuses, mummery and dancing, open-air entertainment, the use of costumes and masks, parodies, farce, jokes, puns and tricks, curses and slang and other forms of folk humour (Stallybrass and White 1986:8).

In the carnival season in early modern Europe, which began in late December or January and reached its

apotheosis after Lent, feasting and excessive drinking were central to the activities. Carnival, the festivities of 'the world turned upside down', had three major themes, both real and symbolic: food, sex and violence (Burke 1994: 186). Meat consumption had particular significance as a liberation from the austerities of Lent (indeed the name 'carnival' springs from the Latin word for meat). Rituals celebrated sexuality, and weddings often took place during carnival festivities. Aggressive and violent acts were depicted symbolically in rituals, but real acts of violence also took place, such as fighting among revellers, the torturing or killing of domestic animals or the stoning of Jews, and carnival was accepted as the time in which one could insult one's fellows freely. Inversions and transgressions were pivotal activities of carnival. Dressing in costume was also central to carnival festivities, including men dressing as women, women as men and lay people in the costumes of clerics. It was common for people to costume themselves as devils, fools, wild men and wild animals. Popular prints of carnival depicted reversals of the relationship between humans and animals, such as the ox turned butcher, carving up a man, or of hierarchical relationships between humans, such as the son beating his father, servants giving orders to their masters, the poor giving alms to the rich, the husband holding the baby and spinning while the wife smokes and holds a gun (Burke 1994: Chapter 7).

The rituals and celebrations of carnival, therefore, served to challenge traditional hierarchies, at least for a time, allowing temporary liberation from routine constraints, controls and the established order. The carnivalesque was about ritualistic inversion, excess and hybridity: it was fundamentally about corporeality rather than disembodied rationality. Unlike 'high' cultural forms, the carnival celebrated the lower strata, including that of the human body. The vulgar, 'grotesque' body was privileged over the bourgeois, civilized body (Bakhtin 1984; Stallybrass and White 1986: Introduction). The

openings and orifices of the body, and the fluids they ingested or emitted, were emphasized, and the lower regions of the body given priority over its upper regions (Stallybrass and White 1986:9). The 'grotesque' carnival body either served to directly oppose the 'low' with the 'high' or else was formed through a process of hybridization or intermixing of binary opposites and elements usually thought of as incompatible (ibid.: 44). These transgressions inspired not fear, anxiety or repulsion, but rather pleasure, excitement, exhilaration, desire.

However, there were limits to behaviours in carnival. It was not simply a case of allowing anarchic behaviour to flourish. The rituals and traditions of the carnivalesque served to maintain certain boundaries and abuse weaker as well as stronger social groups such as women and ethnic and religious minorities. It therefore did not do away with the dominant culture and the official hierarchy, challenging them only temporarily. Some carnivalesque activities, for example, served to scapegoat and pillory outsiders and punish unruly women by humiliating them (Stallybrass and White 1986:24). Stallybrass and White use the term 'displaced abjection' to refer to the process by which "'low" social groups turn their figurative and actual power, not against those in authority, but against those who are even "lower" (women, Jews, animals, particularly cats and pigs)' (ibid.: 53, original emphasis). Nonetheless, carnival activities have also been historically associated with political resistance and were not only about hedonistic revelry—revolts and rebellions often took place at times of major festivals (Burke 1994:203–4).

While many activities associated with the pre-modern carnivalesque may have disappeared by the eighteenth century, there remain elements of the carnivalesque in modern and late-modern culture. The disavowal of carnival by the eighteenth-century bourgeois meant that it became the festival of the Other: 'It encoded all that which the proper bourgeois must strive *not to be* in order to

preserve a stable and "correct" sense of self' (Stallybrass and White 1986:178, original emphasis). While the bourgeoisie into the nineteenth and twentieth centuries denied the transgressive elements, the 'grotesque' body of the carnival, they also defined themselves through this suppression. Their participation in carnival became voyeuristic, partial, guilty, ambivalent. The fair or carnival, for example, was gradually marginalized and confined to certain areas, banished from wealthy areas. In England, the outdoor carnivalesque site of pleasure moved to the seaside resort (Stallybrass and White 1986; Burke 1994: Chapter 9).

In the late twentieth century, such sites as the resort, the fun fair and theme and amusement parks provide the opportunity to engage in modern versions of the carnivalesque. Such activities as mardi gras parades (such as those that occur annually in Rio and the Gav and Lesbian Mardi Gras parade that takes place in Sydney each summer) and agricultural shows or fairs are some of the few examples of pre-modern-style mass carnivalesque events that still occur in western societies. These activities tend to be under the tight control, however, of dominant social groups. Carnivalesque activities in contemporary Britain involving minority groups such as blacks and Gypsies are heavily policed and contained (Sibley 1995:44– 5). The subordination of marginalized groups, therefore, is often confirmed through their participation in and association with the carnivalesque.

The marginalized and despised, those social groups which are considered threatening and polluting, are often the subject of fascination and desire because of their very difference, their otherness. Rejection and repudiation of the Other, therefore, is not simply about exclusion, but also involves the inclusion of the Other into self-identity. The attempt to expel and contain the Other is inevitably implicated with the production of desire: 'disgust always bears the imprint of desire. These low domains, apparently expelled as "Other", return as the object of nostalgia,

longing and fascination' (Stallybrass and White 1986: 191). As a consequence, the bourgeoisie is 'perpetually rediscovering the carnivalesque as a radical source of transcendence. Indeed that act of rediscovery itself, in which the middle classes excitedly discover their own pleasures and desires under the sign of the Other, in the realm of the Other, is constitutive of the very formation of middle-class identity' (ibid.: 201).

As noted in Chapter 7, the black body, for example, has been dominantly represented in white cultures as both sexually revolting and intensely erotically attractive. A significant feature of both responses to black bodies is their cultural coding as different from white bodies, particularly as more animalistic, more sexual, more primitive and closer to nature. Sexual encounters of white bodies with non-white bodies have typically attracted the meanings of forbidden desire and transgressive pleasures. Non-white bodies, both male and female, have been represented in white discourses as simultaneously inferior because of their imputed 'uncivilized' animality, and superior for the same reason: their 'animal' potency and allure. They have been portrayed as both potentially defiling and as intensely erotically attractive in their very exotic nature, their cultural position as Other. This would suggest that the boundary between disgust and desire is very tenuous. Nineteenth-century writings by the British on other races, particularly Africans, while replete with expressions of repulsion, were also often characterized by references to their beauty, attractiveness or desirability (Young 1995). One writer, Thomas Hope, described in an essay published in 1831 some of the black 'varieties of human races' as 'disgusting', 'repulsive' and 'hideously ugly'. However, he also noted of members of 'certain Nubian nations' that: 'Their complexion indeed still is dark, but it is the glossy black of marble or of jet, conveying to the touch sensations more voluptuous even than those of the most resplendent white' (quoted in Young 1995:97).

In an interesting analysis of contemporary luxury icecream advertisements using naked black models to sell the product, Nayak (1997) shows how the forbidden and transgressive eroticism of the black body is used to convey these meanings to the ice-cream, and vice versa. Thus, for example, an advertisement for 'Deep Chocolate' flavour showed an image of a white man diving from a springboard into chocolate flavoured ice-cream, shaped in curves to look like black breasts or buttocks. The copy suggested that eating this ice-cream was akin to the overwhelming and wicked pleasures of engaging in sexual activity with black bodies: 'If intensity scares you, great pleasure upsets you or love makes you flee, please don't try our new Deep Chocolate Ice Creams. The shock of real Callebaut Belgium chocolate might be a tad overwhelming Surrender or stay away' (quoted in Nayak 1997:66-7). This imagery and text implies titillation that is similar to the ways in which pornography is advertised. It supports the notion that interracial sexual relations is a specialist, transgressive pleasure for whites that is suffused with particular intensity and thrills.

Another example of the pleasure and desire produced by transgression is the contemporary horror film, which, like its predecessor the Gothic genre of literature, is preoccupied with transgression of cultural boundaries, order and over fantasy. with excess sadomasochistic eroticism and strongly-felt emotion over everyday realities. The Gothic novel was preoccupied with supernatural horror, with scenes of the torturing of a young female victim in the claustrophobic space of deserted and decayed castles, typically combined with romance and eroticism (Schubart 1995:225). Both the horror film and the Gothic novel reveal traces of the repressed, those thoughts and feelings that are pushed into the unconscious because of their transgressive nature. In their focus on the figure of the monster, which commonly blurs boundaries between the living and the dead, the human and the supernatural, the human and animal, the

human and machine, male and female, solid and liquid, the horror film and Gothic novel allow the projection of repressed and disturbing emotions such as desire, hatred, fear and disgust upon this Other (Schubart 1995; Tudor 1995). Ambivalent emotions are therefore central to the horror film experience, incorporating both fascination and pleasure with fear and dread: 'Ambivalence fundamen tal in horror where the audience voluntarily seeks out the dreadful, where we desire the repulsive, where we wish to break the limits of safety, sanity and normality by moving into a world of chaos dominated by unconscious desires' (Schubart 1995:227).

In summary, then, transgression is a potent source of pleasure as well as fear and anxiety. It is a risky activity because it calls into question accepted conceptual boundaries, threatening self-integrity by allowing the Other into the self. Yet this very act is also the source of the ambivalent pleasure that may be experienced in allowing these boundaries, at least for a time, to be disrupted, blurred or crossed. The power of the culturally forbidden and the contaminated provides the opportunity for frissons of exhilaration and heightened sensibility that go beyond the excitement afforded by merely engaging in 'dangerous' activities.

CONCLUDING COMMENTS

In a world in which self-containment and self-regulation are highly valued and encouraged, participation in activities that are culturally coded as 'risky' allows the contemporary body/self to revel, at least for a time, in the pleasures of the 'grotesque' or 'uncivilized' body. In some social contexts, risk-taking is actively encouraged as a means of escaping from the bounds of everyday life, achieving self-actualization, demonstrating the ability to go beyond expectations or performing gender. The thrills and excitement of the carnivalesque have shifted in location from the religious festival to sites such as the

seaside resort, the fun fair and theme park, the adventure holiday and the 'extreme sport'. Activities such as dancing, drug taking and sexual intercourse provide routes by which the culturally forbidden may be indulged in, at least for a time. The popular media also afford the opportunity to engage vicariously in transgression, to enjoy the sights and sounds of 'grotesque' bodies and thereby experience heightened emotions. Engagement with the marginalized Other is also a potent source of fascination and desire, often at the unconscious level. The courting of symbolic risk implicated by the crossing or blurring of boundaries is a central aspect of the pleasure and excitement associated with transgression and contact with Otherness.

BIBLIOGRAPHY

- Adams, J. (1995) Risk. London: UCL Press.
- Alexander, J. (1996) Critical reflections on 'reflexive modernization'. *Theory, Culture & Society*, 13, 133–8.
- Anderson, K. (1996) Engendering race research: unsettling the self-Other dichotomy. In Duncan, N. (ed), *BodySpace*. London: Routledge, pp. 197–211.
- Bakhtin, M. (1984) Rabelais and His World. Cambridge, MA: MIT Press.
- Bauman, Z. (1991) Modernity and Ambivalence. Cambridge: Polity Press.
- ——(1995) Making and unmaking of strangers. *Thesis Eleven*, 43, 1–16.
- Beck, U. (1992a) From industrial society to the risk society: questions of survival, social structure and ecological environment. *Theory, Culture & Society*, 9, 97–123.
- ——(1992b) Risk Society: Towards a New Modernity. London: Sage.
- ——(1994) The reinvention of politics: towards a theory of reflexive modernization. In Beck, U., Giddens, A. and Lash, S., Reflexive Modernization: Politics, Tradition and Aesthetics in the Modern Social Order. Cambridge: Polity Press, pp. 1–55.
- ——(1995) Ecological Politics in the Age of Risk. Cambridge: Polity Press.
- ——(1996a) World risk society as cosmopolitan society? Ecological questions in a framework of manufactured uncertainties. *Theory*, *Culture & Society*, 13(4), 1–32.
- ——(1996b) Risk society and the provident state. In Lash, S., Szerszinski, B. and Wynne, B. (eds), Risk, Environment and Modernity: Towards a New Ecology. London: Sage, pp. 27–43.
- Beck, U. and Beck-Gernsheim, E. (1995) *The Normal Chaos of Love.* Cambridge: Polity Press.

- Bhabha, H. (1994) *The Location of Culture*. London: Routledge. Bloor, M. (1995) *The Sociology of HIV Transmission*. London: Sage.
- Bourdieu, P. (1984) Distinction: A Social Critique of the Judgement of Taste. London: Routledge & Kegan Paul.
- Bradbury, J. (1989) The policy implications of differing concepts of risk. *Science*, *Technology & Human Values*, 14(4), 380–99.
- Brown, J. (1989) Introduction: approaches, tools and methods. In Brown, J. (ed.), *Environmental Threats: Perception, Analysis and Management*. London: Belhaven Press, pp. 1–18.
- Burke, P. (1994) *Popular Culture in Early Modern Europe* (second revised edition). Aldershot: Scolar Press.
- Castel, R. (1991) From dangerousness to risk. In Burchell, G., Gordon, C. and Miller, P. (eds), *The Foucault Effect: Studies in Govern-mentality*. London: Harvester/ Wheatsheaf, pp. 281–98.
- Cohen, S. and Taylor, L. (1976/1992) Escape Attempts: the Theory and Practice of Resistance to Everyday Life (second revised edition). London: Routledge.
- Collison, M. (1996) In search of the high life: drugs, crime, masculinities and consumption. *British Journal of Criminology*, 36(3), 428–44.
- Comaroff, J. (1993) The diseased heart of Africa: medicine, colonialism, and the black body. In Lindenbaum, S. and Lock, M. (eds), *Knowledge, Power and Practice: the Anthropology of Medicine and Everyday Life.* Berkeley, CA: University of California Press, pp. 305–29.
- Crawford, J., Kippax, S., Onyx, J., Gault, U. and Benton, P. (1992) Emotion and Gender: Constructing Meaning from Memory. London: Sage.
- Crawford, R. (1994) The boundaries of the self and the unhealthy other: reflections on health, culture and AIDS. *Social Science and Medicine*, 38(10), 1347–65.
- Crawshaw, M. (1997) Milk of human kindness? HQ, May/June, 45–7.
- Davison, C., Frankel, S. and Davey Smith, G. (1992) The limits of lifestyle: re-assessing 'fatalism' in the popular culture of illness prevention. *Social Science and Medicine*, 34(6), 675–85.

- Dean, M. (1997) Sociology after society. In Owen, D. (ed.), Sociology after Postmodernism. London: Sage, pp. 205–28.
- ——(1999) Risk, calculable and incalculable. In Lupton, D. (ed.), *Risk and Sociocultural Theory: New Directions and Perspectives*. Cambridge: Cambridge University Press.
- Douglas, M. (1966/1969) Purity and Danger: An Analysis of Concepts of Pollution and Taboo. London: Routledge & Kegan Paul.
- ——(1985) Risk Acceptability According to the Social Sciences. New York: Russell Sage Foundation.
- ——(1992) *Risk and Blame: Essays in Cultural Theory.* London: Routledge.
- Douglas, M. and Wildavsky, A. (1982) Risk and Culture: An Essay on the Selection of Technological and Environmental Dangers. Berkeley, California: University of California Press.
- Duncan, N. (1996) Renegotiating gender and sexuality in public and private spaces. In Duncan, N. (ed.), *BodySpace*. London: Routledge, pp.127–45.
- Elias, N. (1939/1994) *The Civilizing Process*. Oxford: Blackwell. Ewald, F. (1991) Insurance and risks. In Burchell, G., Gordon, C. and Miller, P. (eds), *The Foucault Effect: Studies in*
 - Governmentality. London: Harvester/Wheatsheaf, pp. 197–210,
- ——(1993) Two infinities of risk. In Massumi, B. (ed.), *The Politics of Everyday Fear*. Minneapolis, Minnesota: University of Minnesota Press, pp. 221–8.
- Featherstone, M. (1995) Undoing Culture: Globalization, Postmodernism and Identity. London: Sage.
- Ferguson, H. (1997) Me and my shadows: on the accumulation of body-images in western society. Part one—the image and the image of the body in pre-modern society. *Body & Society*, 3(3), 1–32.
- Flynn, J., Slovic, P. and Mertz, C. (1994) Gender, race, and perception of environmental health risks. *Risk Analysis*, 14 (6), 1101–8.
- Foucault, M. (1984) The politics of health in the eighteenth century. In Rabinow, P. (ed.), *The Foucault Reader*. New York, Pantheon Books, pp. 273–89.
- ——(1988) Technologies of the self. In Martin, L., Gutman, H. and Hutton, P. (eds), *Technologies of the Self: A Seminar with Michel Foucault*. London: Tavistock, pp. 16–49.

- ——(1991) Governmentality. In Burchell, G., Gordon, C. and Miller, P. (eds), *The Foucault Effect: Studies in Governmentality*. Hemel Hempstead: Harvester Wheatsheaf, pp. 87–104.
- Fox, N. (1999) Postmodern reflections on 'risks', 'hazards' and life choices. In Lupton, D. (ed.), *Risk and Sociocultural Theory: New Directions and Perspectives*. Cambridge: Cambridge University Press.
- Furlong, A. and Cartmel, F. (1997) Young People and Social Change: Individualization and Risk in Late Modernity. Buckingham: Open University Press.
- Giddens, A. (1990) The Consequences of Modernity. Cambridge: Polity Press.
- ——(1991) Modernity and Self-Identity. Cambridge: Polity Press.
- ——(1992) The Transformation of intimacy: Sexuality, Love and Eroticism in Modern Societies. Cambridge: Polity Press.
- ——(1994) Living in a post-traditional society. In Beck, U., Giddens, A. and Lash, S., *Reflexive Modernization: Politics, Tradition and Aesthetics in the Modern Social Order*. Cambridge: Polity Press, pp. 56–109.
- ——(1998) Risk society: the context of British politics. In Franklin, J. (ed.), *The Politics of Risk Society*. Cambridge: Polity Press, pp. 23–34.
- Gordon, C. (1991) Governmental rationality: an introduction. In Burchell, G., Gordon, C. and Miller, P. (eds), *The Foucault Effect: Studies in Governmentality*. Hemel Hempstead: Harvester Wheatsheaf, pp. 1–52.
- Gotfrit, L. (1991) Women dancing back: disruption and the politics of pleasure. In Giroux, H. (ed.), *Postmodernism*, *Feminism*, *and Cultural Politics: Redrawing Educational Boundaries*. New York: State University of New York Press, pp. 174–195.
- Graham, J. and Clemente, K. (1996) Hazards in the news: who believes what? *Risk in Perspective* 4(4), 1–4.
- Greco, M. (1993) Psychosomatic subjects and the 'duty to be well': personal agency within medical rationality. *Economy and Society*, 22(3), 357–72.
- Green, J. (1997) Risk and the construction of social identity: children's talk about accidents. *Sociology of Health & Illness*, 19(4), 457–79.

- Grosz, E. (1994) Volatile Bodies: Toward a Corporeal Feminism. Sydney: Allen & Unwin.
- Hacking, I. (1990) *The Taming of Chance*. Cambridge: Cambridge University Press.
- Hall, S. (1997) The work of representation. In Hall, S. (ed.), *Representation: Cultural Representations and Signifying Practices*. London: Sage, pp. 13–74.
- Handwerker, L. (1994) Medical risk: implicating poor pregnant women. *Social Science and Medicine*, 38(5), 665–75.
- Hansson, S. (1989) Dimensions of risk. *Risk Analysis*, 9(1), 107–12.
- Hargreaves, J. (1997) Women's boxing and related activities: introducing images and meanings. *Body & Society*, 3(4), 33–50.
- Hart, G. and Boulton, M. (1995) Sexual behaviour in gay men: towards a sociology of risk. In Aggleton, P., Davies, P. and Hart, G. (eds), *AIDS: Safety, Sexuality and Risk*. London: Taylor & Francis, pp. 55–67.
- Heimer, C. (1988) Social structure, psychology, and the estimation of risk. *Annual Review of Sociology*, 14, 491–519.
- Hilgartner, S. (1992) The social construction of risk objects: or, how to pry open networks of risk. In Short, J. and Clarke, L. (eds), *Organizations, Uncertainties, and Risk*. Boulder, Colorado: Westview Press, pp. 39–53.
- Hunter, I. (1993) Subjectivity and government. *Economy and Society*, 22(1), 123–34.
- Kaprow, M. (1985) Manufacturing danger: fear and pollution in industrial society. *American Anthropology*, 87, 342–56.
- Kemshall, H. (1997) Sleep safely: crime risks may be smaller than you think. *Social Policy & Administration*, 31(3), 247–59.
- Kendall, G. and Wickham, G. (1992) Health and the social body. In Scott, S., Williams, G., Platt, S. and Thomas, H. (eds), *Private Risks and Public Dangers*. Aldershot: Avebury, pp. 8–18.
- Kristeva, J. (1982) Powers of Horror: an Essay on Abjection. New York: Columbia University Press.
- Kroker, A. and Kroker, M. (1988) Panic sex in America. In Kroker, A. and Kroker, M. (eds), *Body Invaders: Sexuality and the Postmodern Condition*. Toronto: Macmillan.

- Lash, S. (1993) Reflexive modernization: the aesthetic dimension. *Theory, Culture & Society*, 10, 1–23.
- ——(1994) Reflexivity and its doubles: structure, aesthetics, community. In Beck, U., Giddens, A. and Lash, S., Reflexive Modernization: Politics, Tradition and Aesthetics in the Modern Social Order. Cambridge: Polity Press, pp. 110–73.
- Lash, S. and Urry, J. (1994) Economies of Signs and Space. London: Sage.
- Lash, S. and Wynne, B. (1992) Introduction. In Beck. U., *Risk Society: Towards a New Modernity*. London: Sage, pp. 1–8.
- Luhmann, N. (1993) *Risk: A Sociological Theory*. New York: Aldine de Gruyter.
- Lupton, D. (1995) The Imperative of Health: Public Health and the Regulated Body. London: Sage.
- ——(1998) The Emotional Self: A Sociocultural Exploration. London: Sage.
- ——(in press) Dangerous places and the unpredictable stranger: constructions of fear of crime. *Australian and New Zealand Journal of Criminology*.
- Lyng, S. (1990) Edgework: a social psychological analysis of voluntary risk taking. *American journal of Sociology*, 95(4), 851–86.
- Macgill, S. (1989) Risk perception and the public: insights from research around Sellafield. In Brown, J. (ed.), Environmental Threats: Perception, Analysis and Management. London: Belhaven Press, pp. 48–66.
- Macken, D. (1998) Life on the edge. *Sydney Morning Herald*, 17 January.
- Maffesoli, M. (1996) The Time of the Tribes: the Decline of Individualism in Mass Society. London: Sage.
- Marris, C. and Langford, I. (1996) No cause for alarm. *New Scientist*, 28 September, 36–9.
- Martin, E. (1994) Flexible Bodies: Tracking Immunity in American Culture From the Days of Polio to the Age of AIDS. Boston: Beacon Press.
- Massumi, B. (1993) Everywhere you want to be: introduction to fear. In Massumi, B. (ed.), *The Politics of Everyday Fear*. Minneapolis, Minnesota: University of Minnesota Press, pp. 3–38.

- Maticka-Tyndale, E. (1992) Social construction of HIV transmission and prevention among heterosexual young adults. *Social Problems*, 39(3), 238–52.
- Mellor, P. and Shilling, C. (1997) *Re-forming the Body: Religion, Community and Modernity*. London: Sage.
- Melucci, A. (1996) The Playing Self: Person and Meaning in the Planetary Society. Cambridge: Cambridge University Press.
- Merleau-Ponty, M. (1962) Phenomenology of Perception. London: Routledge & Kegan Paul.
- Michael, M. (1996) Constructing Identities: the Social, the Nonhuman and Change. London: Sage.
- Miller, W. (1997) *The Anatomy of Disgust*. Cambridge, Massachusetts: Harvard University Press.
- Mossman, D. (1997) Deinstitutionalization, homelessness, and the myth of psychiatric abandonment: a structural anthropology perspective. *Social Science and Medicine*, 44 (1), 71–84.
- Muchembled, R. (1985) Popular Culture and Elite Culture in France, 1400–1750. Baton Rouge: Louisiana State University Press.
- Myslik, W. (1996) Renegotiating the social/sexual identities of places: gay communities as safe havens or sites of resistance? In Duncan, N. (ed.), *BodySpace*. London: Routledge, pp. 156–69.
- Nayak, A. (1997) Frozen bodies: disclosing whiteness in Haagen-Dazs advertising. *Body & Society*, 3(3), 51–72.
- Nelkin, D. and Brown, M. (1984) Workers at Risk: Voices from the Workplace. Chicago: University of Chicago Press.
- O'Malley, P. (1992) Risk, power and crime prevention. *Economy and Society*, 21(3), 252–75.
- Perin, C. (1988) Belonging in America: Reading Between the Lines. Wisconsin: the University of Wisconsin Press.
- Reddy, S. (1996) Claims to expert knowledge and the subversion of democracy: the triumph of risk over uncertainty. *Economy and Society*, 25(2), 222–54.
- Rhodes, T. (1995) Theorizing and researching 'risk': notes on the social relations of risk in heroin users' lifestyles. In Aggleton, P., Davies, P. and Hart, G. (eds), *AIDS: Safety, Sexuality and Risk*. London: Taylor & Francis, pp. 125–43.

- Rose, G. (1993) Feminism and Geography: the Limits of Geographical Knowledge. Minneapolis, Minnesota: University of Minnesota Press.
- Royal Society (1992) Risk: Analysis, Perception and Management. London: Royal Society.
- Said, E. (1978) Orientalism. London: Penguin.
- Schubart, R. (1995) From desire to deconstruction: horror films and audience reactions. In Kidd-Hewitt, D. and Osborne, R. (eds), *Crime and the Media: The Postmodern Spectacle*. London: Pluto, pp. 219–42.
- Short, J. (1984) The social fabric at risk: toward the social transformation of risk analysis. *American Sociological Review*, 49, 711–25.
- Sibley, D. (1995) Geographies of Exclusion: Society and Difference in the West. London: Routledge.
- Skolbekken, J.-A. (1995) The risk epidemic in medical journals. *Social Science & Medicine*, 40(3), 291–305.
- Slovic, P. (1987) Perception of risk. Science, 236, 280-5.
- Smart, B. (1993) Postmodernity. London: Routledge.
- Stallybrass, P. and White, A. (1986) *The Politics and Poetics of Transgression*. Ithaca, New York: Cornell University Press.
- Szerszinski, B., Lash, S. and Wynne, B. (1996) Introduction: ecology, realism and the social sciences. In Lash, S., Szerszinski, B. and Wynne, B. (eds), *Risk, Environment and Modernity: Towards a New Ecology*. London: Sage, pp. 1–26.
- Thomson, R. (1997) Extraordinary Bodies: Figuring Physical Disability in American Culture and Literature. New York: Columbia University Press.
- Tudor, A. (1995) Unruly bodies, unquiet minds. *Body & Society*, 1(1), 25–42.
- Valentine, G. (1996) (Re)negotiating the 'heterosexual street': lesbian productions of space. In Duncan, N. (ed.), *BodySpace*. London: Routledge, pp. 146–55.
- Whittaker, D. and Hart, G. (1996) Research note: Managing risks: the social organization of indoor sex work. *Sociology of Health & Illness*, 18(3), 399–414.
- Williamson, j. (1989) Every virus tells a story: the meanings of HIV and AIDS. In Carter, E. and Watney, S. (eds), *Taking Liberties: AIDS and Cultural Politics*. London: Serpent's Tail, pp. 69–80.

- Wynne, B. (1989) Frameworks of rationality in risk management: towards the testing of naive sociology. In Brown, J. (ed.), *Environmental Threats: Perception, Analysis and Management*. London: Belhaven Press, pp. 33–47.
- ——(1996) May the sheep safely graze? A reflexive view of the expert-lay knowledge divide. In Lash, S., Szerszinski, B. and Wynne, B. (eds), *Risk, Environment and Modernity: Towards a New Ecology.* London: Sage, pp. 44–83.
- Young, R. (1995) Colonial Desire: Hybridity in Theory, Culture and Race. London: Routledge.

INDEX

abject, the 29, 141–42, 171 accidents 8, 108–8, 160–8 Adams, J. 58 Agent Orange 111 AIDS see HIV/AIDS Alexander, J. 84 Anderson, K. 135 ambiguities 45, 133	carnivalesque 169–9; concepts of, 126–9, 132, 151–9; female 143–1, 164–3; male 144, 163–60; monstrous 174–1 Bourdieu, M. 122–1 Bradbury, J. 18, 22
anomalies 45, 133	breastfeeding 129–8
	Brown, J. 18, 20
Bakhtin, M. 127, 169, 170	Bryant, M. 3–4
Bauman, Z. 131–30, 133, 137	Burke, P. 169 passim
Beck, U. 5, 10, 16, 25, 27, 59,	
125;	carnival 169–8
individualization 71–74, 107;	case-management risk 97, 99–
reflexive modernization	Castel, R. 6, 92-4
66–9;	class 70-9, 73-2, 92, 115-15,
reflexivity 68-9, 72-2, 83-	146–4, 171–9
3, 110, 116, 120, 121;	collective effervescence 156–4
'risk society' 60–66	cognitive science perspective
Beck, U. and Beck-Gernsheim,	17, 19–25, 37
E. 60	Cohen, S. and Taylor, L. 153,
Bhaba, H. 134	154, 160, 168
Bloor, M. 21, 119, 122	Collison, M. 163
body, the 26, 29, 79, 89, 155;	Comaroff, J. 133
black 133, 173–70;	contamination see pollution
boundaries, symbolic 40–2, 53–5;	Coorgs 42

Crawford, J., Kippax, S., Onyx, J., Gault, U. and	extreme sports 152
Benton, P. 162–9	fate, 6, 78, 81, 113
Crawford, R. 142	Featherstone, M. 11, 160
Crawshaw, M. 130	Ferguson, H. 127
crime 96, 103, 147, 163	Flynn, J., Slovic, P. and Mertz,
criminal, the 147	C. 24–4
critical structuralism 27, 83	forensic theory of danger 47, 49
dangerousness 92-5	Foucault, M. 16, 25, 27, 85
Davison, C., Frankel, S. and	passim
Davey-Smith, G. 113	Fox, N. 31
Dean, M. 97, 98–7, 99–102, 104	functional structuralism 27, 35, 57
dirt 41, 46, 121, 129, 138-6,	Furlong, A. and Cartmel, F.
141–9, 144–2, 168–5;	117
see also pollution	
disability 139	gay men 54, 55, 56, 149
discourse 15	gender 160-63;
disembedding mechanisms 75,	see also women, mother,
79	the
Douglas, M. 8, 10, 16, 19	Giddens, A., 4, 6, 11, 16, 25,
passim, 35–7, 56–7, 107,	27, 59;
120, 133:	globalization 48-8, 64-4,
blame 46–9;	68, 75;
culture 37–9;	modernity 74–7;
grid-group model 50–56;	reflexivity 15, 83–3, 107,
Otherness 126;	110, 116, 120, 121;
purity 39–46, 168–5	risk-taking 153–1, 157,
Douglas, M. and Wildavsky,	158, 160;
A. 46, 51–1, 58	trust 79-82
Duncan, N. 149	Gordon, C. 89
	Gothic literature 174
edgework 154-4, 163-60	Gotfrit, L. 165–2
Elias, N. 127, 128	governmentality 86–92, 104–3
embodiment see body	Graham, J. and Clemente, K.
employment see work	24–4
epidemiological risk 97, 98–7	Greco, M. 92
escape attempts 153	Green, J. 108–8, 160–8
Ewald, F. 5, 8, 29, 88, 98,	grid-group model 50–56
156, 160	Grosz, E. 28–8

habit 122-2 habitus 122-2 Hacking, I. 5 Hall, S. 34 Handwerker, L. 91 Hansson, S. 20 Hargreaves, J. 166–3 Hart, G. and Boulton, M. 119 health belief model 21 Heimer, C. 20 heroin users 123-2 high/low opposition 145–44 Hilgartner, S. 31–1 HIV/AIDS 53-5, 119, 121, 142, 144 homeless people 138-6 horror film 174–1 Hunter, I. 122 hybridity 131, 134–3, 170–7

individualization 26, 71–74, 79, 107, 108–8, 117 infancy 140–8 insurance, 4, 6–8, 101; see also insurantial risk insurantial risk 97–6 intimacy 73–2, 81–82

Kaprow, M. 58 Kemshall, H. 57 Kendall, G. and Wickham, G. 92 Kristeva, J. 28–8, 140–9 Kroker, A. and Kroker, M. 12, 130–9

Lash, S. 84, 116–15, 120, 121, 122 Lash, S. and Urry, J. 11, 84, 116, 120 Lash, S. and Wynne, B. 111 lesbians 149 liminality 135–6 love *see* intimacy Luhmann, N. 4, 12 Lupton, D. 99, 147, 156 Lyng, S. 154–2, 157

Macgill, S. 113–12 Macken, D. 152 Maffesoli, M. 157 Marris, C. and Langford, I. 20–21 Martin, E. 160 Massumi, B. 11, 12–13 Maticka-Tyndale, E. 121 Mellor, P. and Shilling, C. 127, 157 Melucci, A. 107–7, 131 men 160-8, 163-60 Merleau-Ponty, M. 120 Michael, M. 112–11 Miller, W. 129, 168 Mossman, D. 138–6 mother, the 140-8, 143 Muchembled, R. x-1, 2, 127 Myslik, W. 149

Nayak, A. 173–70 Nelkin, D. and Brown, M. 118 neo-liberalism 87, 89, 100 passim new prudentialism 101–102 normalization 88

object relations theory 140–8 O'Malley, P. 96, 101, 103 ontological security 80–9, 82 organizations 50–51 Otherness 16, 26, 29, 49, 105, 125–4, 131, 150;

abject, the 141–42; carnival 171–6; desire and, 172–70; embodiment 126–9, 132; hybridity 131, 134–3; liminality 135–6; psychodynamics of 140– 42; spatiality 145–6; transgression 167–71, 170– 8, 173–72	reflexive modernization 26, 66–9, 84 reflexivity 15, 68, 72–1, 75 passim, 83–2, 105 passim; aesthetic/hermeneutic 120–19; risk-taking and 157–7; see also reflexive modernization relativism 29–9, 34–4 Rhodes, T. 123–2 risk objects 31–1
panic bodies 130–9	'risk society' 60–66
Perin, C. 137–5	Rose, G. 148
phenomenology 28	Royal Society 19
pollution 39–46, 50, 139, 142–	
40, 149, 169;	Said, E. 135
see also dirt	Schubert, R. 174–1
Port Arthur see Bryant, M.	Seascale 114
poststructuralism 27–7, 85–5	Sellafield 113–12
power relations 27–7, 115–17	sex workers 119, 149
pregnancy 89–91	sexuality 168, 172–70
prostitutes see sex workers	Short, J. 8, 9, 10
psychiatry 93	Sibley, D. 133, 145, 168, 172
psychoanalytic perspective 28–	Skolbekken, JA. 9, 10
8	Slovic, P. 19, 20
psychological research see	Smart, B. 11
cognitive science	social constructionism 16, 29–
perspective, psychometric research	35, 39, 61–1, 83, 85–5
	spatiality 145–6
psychometric research 19–25, 37, 38, 57–7	Stallybrass, P. and White, A. 128, 145–4, 169 passim
pure relationship 81–82	stranger, the 137
purity 39–5, 121, 142, 168–5	
purity 39–3, 121, 142, 166–3	subjectivity 13, 14, 16, 89, 102–1, 105–5
rationality 21–2, 24, 110	Szerszinski, B., Lash, S. and
realist perspective 17, 34, 60–62	Wynne, B. 108
Reddy, S. 6	taboo 1, 46, 47, 48
reflexive biography 72	technico-scientific approaches 16–25

190 INDEX

Thompson, Hunter S. 154 Thomson, R. 139 Thredbo 106–6, 115 transgression 167–71, 170, 173–72 trust 79–82 Tudor, A. 174

Valentine, G. 149

water contamination 32–2 Whittaker, D. and Hart, G. 119 Williamson, J. 142 women 143–1, 148–5, 162–9, 164–3 171; *see also* mother, the, pregnancy work 73, 117–17, 158–6 Wynne, B. 34, 110, 111–10

Young, R. 134–3, 173 young people 117